Credit Constraints During the Financial Crisis: A Firm-Level Analysis

Inaugural-Dissertation zur Erlangung des Grades Doctor oeconomiae publicae (Dr. oec. publ.) an der Ludwig-Maximilians-Universitität München

vorgelegt von

Karl Manuel Wiegand

2014

Referentin:Prof. Dr. Monika SchnitzerKorreferent:Prof. Panu Poutvaara, Ph.D.Promotionsabschlussberatung:5. November 2014

Acknowledgements

I want to thank Christa Hainz for giving me the opportunity to write this Ph.D. thesis at the Ifo Institute in Munich and for all her guidance over the last years. I thank my supervisors Monika Schnitzer, Panu Poutvaara, and Christoph Trebesch for their inspiring comments and for reviewing this thesis.

Sincere thanks are given to my friends and colleagues at the Ifo Institute and the University of Munich for all their advice. Special thanks are also due to Merten Plihal for the time and effort he has spent to distract me from the day-to-day research work.

Finally, I express my deepest gratitude to Judith Saurer and my family for all the support during my work on this thesis.

Munich, June 2014

Manuel Wiegand

Contents

Introduction

1
T

1	Rela	ationsh	nip Banking During the Financial Crisis: Through Which	
	Cha	nnels	Did It Work?	9
	1.1	Introd	uction	9
	1.2	Hypot	heses	13
	1.3	Data		16
		1.3.1	The Data Set	16
		1.3.2	Impairments of Credit Finance	18
		1.3.3	Measuring Relationship Banking	19
		1.3.4	Control Variables for Firm Characteristics	21
		1.3.5	Control Variables for Bank Characteristics	23
	1.4	Metho	dology and Results	26
		1.4.1	Methodology	26
		1.4.2	Relationship Banking and Impairments of Credit Finance	26
		1.4.3	Other Determinants of Impairments of Credit Finance	29
	1.5	Robus	tness Checks	29
		1.5.1	Estimation of Nonlinear Models	29
		1.5.2	Addressing Potential Reverse Causality	30
		1.5.3	Controlling for Bank Balance Sheet Data	31
	1.6	Conclu	ision	33

TABLE OF CONTENTS

2	Friendship and Money, Oil and Water? Credit Constraints and "Family				
	and	Frien	ds" Finance	35	
	2.1	Introd	luction	35	
	2.2	Hypot	heses	39	
	2.3	Data		40	
		2.3.1	The Data Set	40	
		2.3.2	The Use of F&F Finance in Germany	40	
		2.3.3	F&F Finance and Unsuccessful Bank Credit Negotiations	42	
		2.3.4	F&F Finance and Firm Characteristics	43	
	2.4	Metho	odology and Results	46	
		2.4.1	Methodology	46	
		2.4.2	Results: Ordinary Least Squares Estimations	49	
		2.4.3	Results: Instrumental Variables Estimations	50	
		2.4.4	Signalling Effect of F&F Finance	51	
	2.5	Discus	ssion of the Instrumental Variable	52	
	2.6	Robus	stness Checks	56	
		2.6.1	Controlling for Sample Selection Bias	56	
		2.6.2	The Role of Discouraged Borrowers	58	
	2.7	Concl	usion	58	
3	Are	Real	Effects of Credit Supply Overestimated? Bias from Firms	,	
	Cur	rent S	ituation and Future Expectations	61	
	3.1	Introd	luction	61	
	3.2	Hypot	besis	65	
	3.3	Data		67	
		3.3.1	The Data Set	67	
		3.3.2	The Treatment of Constrained Credit Supply	69	
		3.3.3	Measuring Firm-Level Real Economic Activity	70	
		3.3.4	Heterogeneity in Firms' Balance Sheets	72	
		3.3.5	Heterogeneity in Firms' Current Business Situation and Future Ex-		
			pectations	73	

TABLE OF CONTENTS

3.4	4 Methodology and Results				
	3.4.1	Methodology	75		
	3.4.2	Ordinary Least Squares Estimations	76		
	3.4.3	Propensity Score Matching Estimations	78		
3.5	Robus	tness Checks	80		
	3.5.1	Potential Caveats of the Survey Data	80		
	3.5.2	An Alternative Measure for Firm-Level Production	82		
	3.5.3	Employment Effects	83		
	3.5.4	Sample Selection in PSM Estimations	84		
	3.5.5	The Role of the Financial Crisis of 2007-09	85		
3.6	Discus	sion: No Real Effects of Credit Supply?	85		
3.7	Conclu	nsion	86		
Appen	dices		88		
List of	Figure	es	117		
List of Tables		3	117		
Bibliog	raphy		121		

Introduction

Access to finance is a key driver of economic growth.¹ It allows firms to fund both longterm investments and working capital, thereby spurring economic activity. In addition, well-functioning financial markets ensure that funding is allocated to the most profitable investment opportunities in an economy, which leads to a maximisation of social welfare. History, however, has shown that failures in financial markets come at enormous costs. During the financial crisis of 2007-09, for example, many banks ceased to lend and researchers, politicians, and practitioners bemoaned that a credit crunch was hampering economic growth.² Governments responded immediately with different policy measures to secure banks' liquidity and ensure that credit continued to flow to the real sector; for instance, bail-out funds for banks were set up and the levels of deposit insurance coverage were increased. Drawing these measures effectively requires a deep understanding of a credit crunch and its impact on the economy.

Empirical research on the causes and consequences of credit constraints provides a solid foundation for designing the right policy measures in response to a credit crunch. In particular, firm-level data allows in-depth analyses along several lines. Which kinds of impairments of credit finance do firms face due to a credit crunch? Even if credit is granted to firms and credit volumes remain stable, do firms suffer from a deterioration of terms and conditions under which they can borrow (e.g., higher interest rates or collateral requirements)? Does firm behaviour alleviate impairments of credit finance? Do firms

¹See King and Levine (1993), Jayaratne and Strahan (1996), Rajan and Zingales (1998), Beck, Levine, and Loayza (2000), Demirgüç-Kunt and Maksimovic (1998), and Beck, Demirgüç-Kunt, and Maksimovic (2005).

²In April 2008, the International Monetary Fund (IMF) estimated that a credit crunch in subsequent quarters could slow down year-on-year GDP growth in the U.S. by 1.4 percentage points (International Monetary Fund (2008), p. 35).

react to credit constraints by using other sources of finance? What are the consequences of credit constraints with respect to real economic activity? Are these real effects caused by credit supply-side factors (e.g., banks ceasing to lend because of liquidity constraints) or by credit demand-side factors (e.g., firms' deteriorating creditworthiness)?

This dissertation addresses these questions by providing three empirical analyses of credit constraints at the firm-level. Chapter 1 shows which kinds of impairments of credit finance firms experienced due to the financial crisis of 2007-09 and whether the risk of facing them was lowered by relationship banking (i.e. the concentration of bank business on a small number of closely connected banks). Chapter 2 assesses firms' behaviour under credit constraints by testing whether they use "Family & Friends" (F&F) finance in response to unsuccessful bank credit negotiations. Finally, Chapter 3 analyses whether the link between credit constraints and real economic activity is caused by credit supply-side factors, or if constrained credit supply is just a mirror picture of firm-side factors that actually hamper real economic activity.

When analysing a credit crunch, it is crucial to understand which symptoms firms could face and how firms' behaviour can help alleviate these. For example, firms can resolve asymmetric information by establishing a close bank relationship to secure access to bank credit (Sharpe, 1990; von Thadden, 2004). This could be of particular importance during times of financial crisis, when high levels of uncertainty keep banks from lending at all or make them do so according to terms and conditions that are less favourable to the firm.

Chapter 1 provides evidence of different kinds of impairments of firms' credit finance due to the financial crisis of 2007-09. The data stems from a survey that was conducted among German firms in September 2011. In the survey, firms indicate which impairments of credit finance they have faced and they report their number of main bank relationships as a measure of relationship banking. Estimations show that relationship banking lowers a firm's risk of facing higher information requirements by banks. It also lowers the risk of facing worse non-price terms of credit (i.e. shorter maturities and higher collateral requirements). However, relationship banking does not affect the probability of experiencing constrained credit availability. Finally, the results are mixed with respect to the impact of relationship banking on the risk of facing higher interest rates due to the finan-

cial crisis. The diverging effects of relationship banking on different impairments suggest that soft information, which banks gather over the course of a customer relationship, influences the negotiation of price-terms and non-price terms of credit, but not a bank's credit granting decision.

The key contribution of Chapter 1 stems from the fact that the survey data captures all kinds of impairments within one analysis. In contrast, estimating the impact of relationship banking on loan contract data restricts analyses to firms to which credit was granted and neglects situations in which no credit contract is completed. The analysis of data on rejected credit applications as a measure of credit availability solves this problem, but is limited to firms that have applied for credit. This approach does not account for firms that do not apply for credit because they expect negotiations to be unsuccessful. The analysis in Chapter 1 does not face any of these limitations. Asking firms in a perception-based approach which kinds of impairments of credit finance they have faced allows a broad analysis of the effects of relationship banking.

Besides analysing the symptoms of a credit crunch, the question of how firms' behaviour changes in response to credit constraints must be raised. For example, firms could have a tendency to move into other sources of finance. One of these could be informal finance when sources of formal funding dry up. This is particularly worrying because it keeps potential welfare gains of financial intermediation from being realised and increases firms' borrowing costs (Djankov, Lieberman, Mukherjee, and Nenova, 2003). The use of informal finance has been analysed in previous empirical studies for developing economies and for start-up businesses in developed countries. However, the question of whether established firms in highly developed economies also turn to informal finance when facing credit constraints has been widely neglected in empirical research.

Chapter 2 fills this gap by using a sample of mostly established firms from Germany, a highly developed economy, to analyse whether firms use "Family & Friends" (F&F) finance as a substitute for unsuccessfully negotiated bank credit. In the survey that is also used for the analysis in Chapter 1, firms report whether they receive funding from F&F related to the business ("F&F Business") and from F&F privately connected to the entrepreneur ("F&F Private"). Firms also indicate if their most recent negotiations with

banks on a line of credit and/or a loan led to credit being granted. Estimations show that firms use both kinds of F&F finance in response to unsuccessful bank loan negotiations. They do not, however, use it as a substitute for unsuccessfully negotiated lines of credit. Therefore, F&F finance is more important in financing particular investment projects than in working capital finance.

Apart from containing seminal evidence on the use of F&F finance among established firms in Germany, Chapter 2 provides key contributions in the empirical analysis of firms' behaviour under credit constraints. Endogeneity issues are resolved using an instrumental variables estimation approach and sample selection issues are addressed. This stands in contrast to previous literature, which primarily provides evidence on the correlations between the use of informal finance and different firm characteristics.

Knowing that firms' credit finance is impaired, the question arises as to how this affects real economic activity. It is commonly acknowledged that real economic activity and credit market conditions are intertwined (e.g., Bernanke, Gertler, and Gilchrist (1996)). At the firm-level, a correlation between business activity and the experience of credit constraints has two potential explanations. First, credit supply shocks (e.g., a dry-up in the interbank market) could make potential lenders unable to meet firms' credit demand. Without access to credit, firms must postpone investments or reduce their business activity, which leads to lower real economic activity. Second, firm-side factors (e.g., failed business models) could contribute to the slowdown of real economic activity, and banks' reluctance to lend to firms could be a mirror picture of the firms' lack of creditworthiness. Therefore, economic activity could in fact not be hampered exclusively by credit supply-side factors, but also by firm-side factors.

Chapter 3 analyses the question of how to rule out bias from firm-side factors in the estimation of the real effects of credit supply using firm-level data for Germany between 2003 and 2011. The results show that a failure to control for firms' current business situation and future expectations leads to a significant overestimation of credit supply-side effects on real economic activity. In the analysis, a treatment variable indicating constrained credit supply at the firm-level is derived from monthly survey data measuring a firm's perception of bank lending supply. To estimate the treatment effect of constrained credit

supply, firms' monthly changes in production and annual employment growth are used as outcome variables. When controlling for balance sheet variables only, estimations suggest that constrained credit supply reduces real economic activity. This effect, however, is significantly reduced when adding survey-based control variables that explicitly capture firms' current business situation and future expectations. In several specifications, the effect even turns insignificant, which suggests that there is no evidence for credit supply-side effects on real economic activity in the data analysed here.

Chapter 3 therefore provides insights into how survey data contributes to the estimation of real effects of credit supply. The results underline that the sole reliance on balance sheet data could lead to incorrect conclusions about the causes and consequences of credit constraints. The shown importance of holding constant contemporaneous and forwardlooking credit demand-side factors should be considered in empirical studies estimating the real effects of credit supply.

In summary, this dissertation provides valuable insights into how firm-level data allows the application of advanced econometric methods to draw the correct conclusions with respect to policies in response to credit constraints during times of financial crisis. For example, firm-level data can be used to assess how the resolution of asymmetric information between banks and firms helps alleviate credit financing impairments. This could affect the decision on regulatory measures affecting the flow of information between firms and banks. For instance, requiring banks to base their credit granting decisions on hard facts about the firm could crowd out positive effects of soft information gathered from relationship banking. Furthermore, firm-level data offers unique opportunities to analyse how firms behave under credit constraints. This allows policy makers to react to behaviour that could lead to welfare losses such as firms widely borrowing from unregulated financial institutions or using informal finance as a substitute for bank credit. Finally, the decision whether to take measures addressing banks or firms depends on whether credit constraints are driven by credit supply-side or firm-side factors. For this purpose, firmlevel data provides crucial control variables in the analysis of credit constraints and their impact on real economic activity. This dissertation contributes to research along these lines by applying different micro-econometric approaches to seminal firm-level data.

Chapter 1

Relationship Banking During the Financial Crisis: Through Which Channels Did It Work?*

1.1 Introduction

Firms' access to credit finance is impeded by asymmetric information. Banks face problems of adverse selection and moral hazard when they lend to firms. These problems were aggravated by the recent financial crisis when firms operated under increasing uncertainty. In some countries, this arguably contributed to a severe credit crunch.¹ Especially during times of tight credit supply, it is crucial to understand whether firms have less problems with credit finance when they concentrate their bank business on a small number of closely connected banks ("relationship banking") instead of spreading it over a larger number of banks without establishing close ties ("transactional banking"). In particular, we raise the question which kinds of impairments of credit finance due to the financial crisis can be alleviated by relationship banking.

^{*}This chapter is based on joint work with Christa Hainz.

¹For empirical analyses of banks' credit supply during the financial crisis, see, for example, Ivashina and Scharfstein (2010), Puri, Rocholl, and Steffen (2011a), Popov and Udell (2012), deYoung, Gron, Torna, and Winton (2014), and Jimènez, Ongena, Peydro, and Saurina (2012).

So far, empirical studies have analysed different channels of relationship banking in isolation. Its impact on interest rates and non-price terms of bank credit can be analysed using data from credit contracts. Such data, however, inhibits the analysis of credit availability because it only considers situations in which credit is granted. Estimating the effects of relationship banking on the probability of credit approvals addresses this issue, but it does not account for firms that did not apply for credit in the first place because they expect the application to be rejected, for example.

To capture all potential channels of relationship banking, we use survey data to estimate the effect of relationship banking on four different kinds of impairments of firms' credit finance: higher information requirements by banks, constrained credit availability, increased interest rates, and deteriorated non-price terms of credit. To this end, we shed light on the channels through which relationship banking works. Our analysis is based on data from the Ifo "Financing of the German Economy" survey, which encompasses 1,139 firms from the German manufacturing sector. Using German data is beneficial for this analysis because relationship banking is widespread among German firms and the German manufacturing sector relies heavily on funding from banks. In the survey, each firm reports which kinds of impairments of credit finance it faced due to the financial crisis between 2007 and 2009. A firm's number of main banks serves as a measure of relationship banking, with firms that have only one main bank being considered as following the concept of relationship banking in the narrowest sense.

Our analysis first shows that implementing relationship banking by having only one main bank significantly lowers the probability of higher information requirements by banks and the deterioration of non-price terms (i.e. shorter maturities and higher collateral requirements) due to the financial crisis. Second, the effect of relationship banking on the likelihood of increased interest rates is ambiguous as the risk of this impairment is even lower for firms with two main banks than for those with only one. Finally, relationship banking does not reduce the probability of a firm facing constrained credit availability. To estimate these effects without bias from unobserved heterogeneity, we use a large set of variables to control for other firm-side and bank-side factors.

Although contradictory at first sight, these results can be explained by the role of hard and soft information at different stages of bank credit negotiations. Over the last decades, technological progress and bank regulation induced banks to base their lending decisions primarily on credit scores, which are based on hard information.² Therefore, the provision of soft information through relationship banking is no longer affecting a firm's credit availability. If credit is granted, however, the negotiation of the terms and conditions of credit is still affected by the soft information from relationship banking.

Our analysis is most closely related to studies using the number of bank relationships as a measure of relationship banking. In line with our results, Petersen and Rajan (1994) find that a large number of bank relationships increases quoted interest rates. Their analysis is based on U.S. data from the 1987 National Survey of Small Business Finances (NSSBF). Harhoff and Körting (1998) ran a survey similar to the NSSBF among German firms in 1997. They do not find that the number of bank relationships has an impact on interest rates, but provide evidence that a larger number of bank relationships increases the probability that a firm has to pledge collateral, which is in line with our findings. Petersen and Rajan (1994) and Harhoff and Körting (1998) both find that the availability of credit is improved if a firm maintains a smaller number of bank relationships. They measure credit availability indirectly using late payments of trade credit (Petersen and Rajan, 1994) and fast payment discounts not taken (Harhoff and Körting, 1998), arguing that trade credit, the most expensive source of finance for a firm, will only be used if no other source (i.e. bank credit) is accessible. Cole (1998) and Cole, Goldberg, and White (2004) use the 1993 NSSBF to show that a large number of bank relationships lowers the probability that credit is granted by a bank, which constitutes a more direct measure of credit availability.

Other studies use different measures of relationship banking to analyse its impact on credit availability, interest rates, and collateral. Several studies use the length of a rela-

²For loans exceeding \in 750,000, banks in Germany are obliged to make a firm disclose its balance sheets. Furthermore, Basel II bank regulation provided incentives for banks to use ratings to assess the borrower's creditworthiness in order to reduce capital charges (Behn, Haselmann, and Vig, 2014).

tionship,³ others use the scope of products that the firm receives from a bank⁴ to measure relationship banking. In addition, Cole (1998) and Bharath, Dahiya, Saunders, and Srinivasan (2011) use a dummy variable indicating a pre-existing business relationship and Lehmann and Neuberger (2001) use bank-level data to identify relationship banking by asking the bank whether it considers itself to be the main bank of a firm. Overall, these studies do not show a clear effect of relationship banking on credit finance. All these measures of relationship banking focus on features of an individual bank relationship. In contrast, the number of main bank relationships captures a firm's whole portfolio of bank relationships and therefore provides a broader measure of a firm's attitude towards relationship banking.

Furthermore, our analysis relates to studies that analyse the impact of relationship banking on firms' credit finance during times of financial crisis. For the financial crisis of 2007-09, Jimènez, Ongena, Peydro, and Saurina (2012) show that a longer bank relationship increased a firm's probability of getting a loan during the crisis in Spain. Bolton, Freixas, Gambacorta, and Mistrulli (2013) confirm this for Italy and show that relationship banks charge lower interest rates in crisis times. Using data for Korea during the period of 1997-98, Bae, Kang, and Lim (2002) show that a firm's market value is affected by the deteriorating financial health of their main banks. Similarly, Yamori and Murakami (1999) examine the failure of a Japanese bank and its effect on the stock prices of the bank's client firms, thereby evaluating the economic value of the main bank relationship.⁵

We contribute to the literature in important ways to estimate the effects of relationship banking on credit finance. First, we use a perception-based approach to measure impairments of firms' credit finance, an approach that is also used by Campello, Graham, and Harvey (2010), for example. This allows us to analyse the effects of relationship banking

³See Petersen and Rajan (1994), Berger and Udell (1995), Harhoff and Körting (1998), Cole (1998), Degryse and van Cayseele (2000), Lehmann and Neuberger (2001), Bharath, Dahiya, Saunders, and Srinivasan (2011), and Santikian (2014).

⁴See Petersen and Rajan (1994), Harhoff and Körting (1998), Degryse and van Cayseele (2000), Cole, Goldberg, and White (2004), and Santikian (2014).

⁵Several other studies estimate the effects of bank health on client firms during financial crises, but do not look at the particular role of relationship banking in this concern (see, for example, Brewer, Genay, Hunter, and Kaufman (2003), Kang and Stulz (2000), and Ongena, Smith, and Michalsen (2003)).

on different aspects of credit finance within one study. In particular, the impact on the probability of higher information requirements by banks has not been analysed in previous studies, although this shows most directly whether relationship banking influences the flow of information between a firm and its banks. Furthermore, firms' perceptions provide comprehensive measures of impairments of credit finance because firms' decisions in response to these impairments (e.g., the cancellation of investment projects) may depend on their perceptions, and not just on factual outcomes of credit negotiations. Second, we amend existing literature by using a firm's number of main banks as a measure of relationship banking, because this is a more precise measure than the number of business relationships to banks in general.

This chapter is organised as follows. In Section 1.2, we derive four testable predictions about how relationship banking could affect a firm's credit finance. Section 1.3 provides information about the data set, our measures of relationship banking, and impairments of credit finance. We show estimation results in Section 1.4 and robustness checks in Section 1.5 before presenting our conclusions in Section 1.6.

1.2 Hypotheses

Based on theoretical models, we build hypotheses about the impact of relationship banking on four kinds of impairments of a firm's credit finance: higher information requirements by banks, constrained credit availability, increased interest rates, and deteriorated non-price terms. For the design of policy measures, it is important to figure out which of these impairments relationship banking alleviates.

First, we expect that banks are less likely to require further information when they are closely tied to the firm through relationship banking. During ex ante screening and interim monitoring, banks often require firms to provide information, which induces costs. For example, the provision of interim financial statements and business plans to a bank is highly burdensome, especially for small firms. This may be particularly prevalent in times of a financial crisis, when banks require more information as uncertainty increases considerably. The theoretical literature argues that a flow of information about the firm is adjunct to loan contracts (Fama, 1985; Sharpe, 1990), and more generally to all kinds of interactions. If this was the case and if such information helped a bank monitor the firm, relationship banking should imply the provision of information, thereby lowering the probability of additional information requirements by banks. Furthermore, as banks also gain from relationship banking, they may put more effort into processing existing information instead of requiring additional ones from their customers.

Hypothesis 1: Relationship banking lowers the probability of higher information requirements by banks due to the financial crisis.

Second, we expect relationship banking to improve a firm's credit availability. Stiglitz and Weiss (1981) present a theoretical model in which asymmetric information leads to credit rationing. Consequently, the resolution of asymmetric information through relationship banking should improve a firm's access to bank credit. Thakor (1996), however, argues that approaching a large number of banks for credit leads to ambiguous effects on overall credit availability. Although the likelihood that at least one bank will grant credit is higher when many banks are approached, each bank's expected profit from screening declines. Therefore, each bank, knowing a firm's optimal application strategy, is more likely to save the screening effort and ration credit when a firm approaches a large number of banks. During the financial crisis, we expect another aspect to affect a firm's credit availability: banks with constrained lending capacities might first lend to firms about which they are best informed. The risk of ending the relationship and destroying "informational capital" accumulated over the course of the relationship may keep the bank from constraining credit to a relationship borrower.

Hypothesis 2: Relationship banking lowers the probability that the availability of bank credit is constrained due to the financial crisis.

Third, with respect to interest rates, the predictions for the impact of relationship banking are somewhat ambiguous. Relationship banks clearly have an informational advantage compared to outside lenders, which gives them an incentive to extract an informational rent from proprietary information ("hold-up problem"). According to Sharpe (1990) and Petersen and Rajan (1995), banks grant lower rates to young firms, and increase them

later in the course of the relationship. Thereby, loan contracts do not necessarily break even in every period because both parties agree on (implicit) long-term contracts. Boot and Thakor (1994), however, argue that interest rates decrease over time as banks learn about the quality of a firm and can commit to granting lower rates. Assuming limited monopoly power of the relationship lender, the model by Bolton, Freixas, Gambacorta, and Mistrulli (2013) predicts that relationship banks charge higher interest rates than transactional banks in good times and lower interest rates in bad times.

Hypothesis 3: Relationship banking lowers the probability that interest rates increase due to the financial crisis.

Finally, relationship banking should render the non-price terms of credit less stringent for the firm, although non-price terms differ in nature from the interest rate.⁶ With respect to collateral, Inderst and Müller (2007) obtain this result in a model in which a local relationship lender has an informational advantage vis-a-vis transactional lenders. They show that this advantage lowers the lender's collateral requirements. Another aspect of collateral is that it can only fulfill its role of mitigating moral hazard and adverse selection problems if the value of the collateral can be observed by the bank (Rajan and Winton, 1995). Relationship banking leads to proximity between a bank and the firm, which improves the bank's ability to assess the value of the collateral (Boot, 2000). Less uncertainty about the collateral value could lower the overall requirement.

Hypothesis 4: Relationship banking lowers the probability that non-price terms of credit contracts are impaired due to the financial crisis.

⁶While the interest rate induces a pecuniary transfer from the borrower to the bank, non-price terms (such as maturity and collateral requirements) are an outcome of the risk-sharing between the contract parties.

1.3 Data

1.3.1 The Data Set

The following analysis tests the four hypotheses concerning the channels of relationship banking based on data from the Ifo "Financing of the German Economy" survey, the Bureau van Dyk (BvD) Amadeus, and the BvD Bankscope database. The Ifo "Financing of the German Economy" survey was based on a written questionnaire, which was sent to a sample of CFOs of German manufacturing firms in September 2011.⁷ All firms are part of the address database of the Ifo Investment Survey for which the Ifo Institute continuously ensures representativeness of the sample of addressees for the German manufacturing sector. In total, 1,139 firms participated in the survey. The response rate was close to 25 percent, leading to a sample in which small firms (less than 50 employees), mediumsized firms (50-249 employees), and large firms (more than 249 employees) are evenly represented.

German manufacturing firms represent an ideal environment to test our hypotheses for two reasons. First, they are highly dependent on bank credit. According to Hainz and Wiegand (2013), 72.9 percent of the firms in the survey reported that they use bank credit. Second, the concept of having a main bank is deeply rooted in the bank-based German economy (see Section 1.3.3). The number of main banks therefore provides an accurate measure of a firm's degree of following the concept of relationship banking.

The Ifo "Financing of the German Economy" survey data is particularly suited to analyse the channels of relationship banking because it contains both a firm's number of main bank relationships and perception-based information on which kinds of impairments of credit finance that firms have experienced due to the financial crisis. In addition, firm and bank balance sheet data from the BvD Amadeus and the BvD Bankscope database provide a large set of control variables. All variables used in the following analysis are described in Table 1.1.

⁷During the six months prior to the survey, personal meetings with executives were scheduled to conduct pre-tests to rule out that the survey design was subject to problems arising from response behaviour. The pre-test talks were also used to ensure that the possible answers listed in the questionnaire captured the firms' reality.

Variable	Description
Impaired	Firm answered the following question with yes: "Was the credit finance of your firm impaired by the financial crisis between 2007 and 2009?"
Information	Higher information requirements from banks
Poduation	Production of existing lines of gradit
Availability	Constrained availability of new loans or lines of credit
Interest	Increase in interest rates for existing leans or lines of
mierest	credit
Maturities	Bank credit was offered only at shorter maturities
Collateral	Banks required more collateral
	· · · · · · · · · · · · · · · · · · ·
Main bank	Number of main banks
Fmplouces	Number of employees (in heads) at the end of 2010
Accets	Total assots
Fauitu	$F_{\text{outty}} / \Lambda_{\text{scots}}$
Lyuny Long term debt	Long torm dobt / Assots
Cash	Cash / Assots
Raturn	EBIT / Turnovor
Farlunau	Cash discounts drawn / Cash discounts offered to firm
Lariypay	Firm ago in years, based on year of foundation
Aye Incornorated	Firm is a corporation by its logal status
Frt rating	Firm has external rating basides banks' internal ones
Crown	Firm is part of a group company
Group Familu	I argest shareholder is a single person or family
Funnit	Firm generates turneyer abread
Concentration	Share of business that is conducted with the three most
Concentration	important sustamors
	Important customers

Table 1.1: Variable descriptions

Sources: Ifo "Financing of the German Economy" survey, BvD Amadeus database.

1.3.2 Impairments of Credit Finance

The survey question about impairments of credit finance due to the financial crisis started by asking firms: "Was the credit finance of your firm impaired by the financial crisis between 2007 and 2009?". The descriptive statistics for the dummy variable Impaired in Table 1.2 indicate that about 22 percent of the firms in the sample answered this question with "Yes".

	Ν	Freq.	Perc.	Min	Max
Impaired	1,062	235	22.13%	0	1
Information	1,046	153	14.63%	0	1
Reduction	1,046	75	7.17%	0	1
Availability	1,046	114	10.90%	0	1
Interest	1,046	112	10.71%	0	1
Maturities	1,046	27	2.58%	0	1
Collateral	1,046	103	9.85%	0	1
Other	1,046	50	4.78%	0	1
	· ·				

Table 1.2: Impairment of credit finance due to the financial crisis

To shed light on the channels through which relationship banking works, firms with impaired credit finance were asked what kinds of impairments they experienced due to the financial crisis. The possible answers are listed in the lower part of Table 1.2.

First, the dummy variable *Information* indicates that a firm faced higher information requirements by banks (Hypothesis 1). This impairment was reported by over 14 percent of the firms, which makes it the most frequent of all. Second, variables indicating a reduction of existing lines of credit (*Reduction*) and constrained availability of new loans or lines of credit (*Availability*) are used as measures of constrained credit availability (Hypothesis 2). These impairments were reported by 7 percent and 11 percent of the firms, respectively. Third, *Interest* indicates an increase in the interest rate for an existing loan or line of credit (Hypothesis 3). Almost 11 percent of the firms faced this impair-

Note: The table shows descriptive statistics for all kinds of impairments of credit finance due to the financial crisis; *Impaired* equals one if a firm answered the following question with yes: "*Was the credit finance of your firm impaired by the financial crisis between 2007 and 2009?*", and zero if a firm answered the question with no; all other variables equal one if a firm has reported the respective impairment and zero if not; firms could report more than one impairment.

ment. Finally, the dummy variables *Maturities* and *Collateral* are used to measure the probability of non-price terms and conditions of bank credit being impaired (Hypothesis 4). The former variable indicates that banks offered credit only for shorter maturities (reported by less than 3 percent of the firms), the latter indicates that banks requested more collateral (reported by almost 10 percent of the firms).

1.3.3 Measuring Relationship Banking

Based on the following theoretical foundation, we use a firm's number of main bank relationships to measure the extent to which it follows the concept of relationship banking. Boot (2000) defines two conditions for relationship banking to be present:⁸

- 1. Multiple interactions between a bank and its customer over time or across products, through which the bank gathers soft information about the customer.
- 2. The information gathered is non-public and remains proprietary to the bank.

We argue that these conditions are fulfilled for firms with one main bank, whereas having a larger number of main banks violates the conditions in the following two ways. First, a larger number of main banks indicates that a firm spreads its business among different banks. Consequently, each single bank runs less business with the firm and therefore learns less about it, which violates Condition 1. Second, if there is a larger number of main banks gathering information from interactions with the firm, this information can no longer be considered to be proprietary to each bank, which contradicts Condition 2. Therefore, firms with a larger number of main banks do not apply relationship banking as defined by Boot (2000).

Our focus on main bank relationships amends the approach of using the number of banks from which a firm receives financial services as a measure of relationship banking⁹. The number of main banks is a much clearer signal of relationship banking than business

 $^{^{8}{\}rm The}$ "learning theory" of relationship banking contains both elements and is modelled by Bolton, Freixas, Gambacorta, and Mistrulli (2013).

⁹For studies using the number of business relationships to banks as a measure of relationship banking, see, for example, Petersen and Rajan (1994), Harhoff and Körting (1998), Cole (1998), and Cole, Goldberg, and White (2004).

relationships to banks in general and therefore gives us a more stringent view of a firm's attitude towards relationship banking. In addition, our study is based on information provided by the firm. This provides a better insight than data from banks' perceptions on whether they consider themselves to be the main bank of a firm (Elsas, 2005; Lehmann and Neuberger, 2001).¹⁰

Table 1.3 shows the distribution of the number of main banks of the firms in the sample. The largest share of firms (40.4 percent) follows the principle of relationship banking by having only one main bank, which underlines the importance of the main bank concept for German firms. 37 percent of the firms have two main banks. About 22 percent, however, report that they do not have a main bank at all or that they have three and more main banks. For the following empirical analysis, these two groups of firms are defined as not following the concept of relationship banking.

Furthermore, impairments of firms' credit finance due to the financial crisis are linked to the number of main bank relationships in Table 1.3. The fraction of firms whose credit finance was impaired by the financial crisis is lowest among firms with one main bank (15.7 percent). The fraction is slightly higher among firms with a second main bank (21 percent) and by far highest among firms with zero or three and more (30.6 percent). The probability of higher information requirements by banks is smallest for firms that apply relationship banking. For the reduction of existing lines of credit, however, there are hardly any differences between firms with different numbers of main banks. Increased interest rates are the least likely for firms with two main banks, while the likelihood of deteriorated non-price terms is lowest for firms with one main bank.

These descriptive statistics provide initial evidence showing that the gains from relationship banking seem to outweigh the benefits from creating a competitive situation between a large number of banks. The probabilities of different kinds of impairments of credit finance, however, appear to be not equally affected by relationship banking.

¹⁰To gain an impression of what exactly a main bank is for a firm, the data set provides information on whether certain features characterise the two most important bank relationships and if the respective bank is a main bank. This leads to a sample of over 1,600 main bank relationships for which features are listed in Table A.2 in the Appendix. The most important criteria for main banks are the long duration of a relationship, personal support by the bank, and the short distance between the headquarter of the firm and the bank. Only 32 percent of the firms report that the main bank is the most important creditor.

	Main bank (1)	$Main \ bank \ (2)$	Main bank $(0/3+)$
Ν	1,042	1,042	1,042
Freq.	421	386	235
Perc.	40.40%	37.04%	22.55%
Impaired	15.68%	20.98%	30.64%
Information	9.03%	15.28%	23.83%
Reduction	5.94%	8.03%	8.09%
Availability	8.08%	11.92%	14.47%
Interest	8.55%	8.29%	18.72%
Maturities	0.95%	2.85%	5.11%
Collateral	6.65%	11.40%	13.19%
Others	3.33%	4.66%	7.66%

Table 1.3: Relationship banking and impaired credit finance

Note: The table shows descriptive statistics for firms' number of main banks and all kinds of impairments of credit finance due to the financial crisis; *Impaired* equals one if a firm answered the following question with yes: "*Was the credit finance of your firm impaired by the financial crisis between 2007 and 2009?*", and zero if a firm answered the question with no; all other variables equal one if a firm has reported the respective impairment and zero if not; firms could report more than one impairment.

1.3.4 Control Variables for Firm Characteristics

To rule out that the correlation between relationship banking and impairments of firms' credit finance is driven by hitherto unobserved firm characteristics, the following estimations incorporate a set of control variables from survey data and firms' balance sheets. We expect these to determine both a firm's creditworthiness and its attitude towards relationship banking. All variables are listed and summarised in Table 1.4.

First, the variables log(Employees) and log(Assets) in Table 1.4 show that firms that apply relationship banking are smaller than firms that do not. Firm size is also widely acknowledged to affect a firm's access to credit. Large firms often have demand for large scale funding. Access to large amounts of credit may be difficult when banks are short of lending capacities and want to diversify their risk exposure across several firms, rather than clustering risks by granting large loans to few firms. On the other hand, asymmetric information might be lower for large firms as they are typically older and more transparent, which could lower the risk of impaired credit finance. Furthermore, a broad set of variables to control for a firm's financial situation constitute the most obvious predictors of impaired credit finance. Table 1.4, however, does not provide a clear picture of the correlation between a firm's financial situation and its attitude towards relationship banking.

	Main bank (1)	Main bank (2)	Main bank $(0/3+)$
	(N=271)	(N=254)	(N=134)
	(/	()	· /
log(Employees)	4.2	4.4	5.9
log(Assets)	8.5	8.8	10.4
Equity	31.48%	34.39%	34.69%
Long-term debt	28.00%	25.80%	21.40%
Cash	10.22%	12.39%	8.85%
Return $(<3\%)$	42.80%	37.80%	47.01%
Return $(3\% \text{ to } < 7\%)$	33.21%	38.19%	35.07%
Return (7% to $< 10\%$)	16.24%	14.57%	9.70%
Return $(10\% +)$	7.75%	9.45%	8.21%
Earlypay (0%)	3.32%	3.15%	5.22%
Earlypay $(0\% to < 25\%)$	18.08%	11.81%	12.6'9%
Earlypay $(25\% to < 50\%)$	11.07%	13.39%	5.97%
Earlypay $(50\% \text{ to } < 75\%)$	12.92%	9.45%	14.93%
Earlypay $(75\% +)$	54.61%	62.20%	61.19%
log(Age)	3.9	4.1	4.2
Incorporated	69.74%	64.17%	70.15%
Ext. rating	20.66%	20.47%	21.64%
Group	25.83%	26.38%	42.54%
Family	79.70%	84.65%	76.87%
Export	85.24%	87.40%	90.30%
Concentration $(<10\%)$	15.50%	13.78%	26.87%
Concentration $(10\% to < 30\%)$	49.08%	46.46%	42.54%
Concentration $(30\% \text{ to } < 50\%)$	18.45%	25.20%	16.42%
Concentration $(50\% +)$	16.97%	14.57%	14.18%

Table 1.4: Firm characteristics and relationship banking

Note: The table shows descriptive statistics for all firm characteristics that are used as control variables in the estimations in this study; see Table 1.1 for a description of all variables.

For the balance sheet variables *Equity*, *Long-term debt*, and *Cash*, we calculate the average value of the balance sheets in 2007, 2008, and 2009 in order to capture a firm's financial situation during the entire financial crisis. Data on return on sales (*Return*) and taken early payment discounts (*Earlypay*) are only available in categorical variables so that we

have to use the respective variables in 2008 as control variables, instead of calculating average values of the three crisis years.

In addition to firm size and balance sheet variables, survey data provides additional control variables for firm age (log(Age)), transparency (*Incorporated, Ext. rating*), ownership (*Group, Family*), and export status. All these variables are commonly used as predictors of firms' access to credit finance. Table 1.4 shows that firms that implement relationship banking do not differ much in age and transparency from those that do not, but the latter are more likely to belong to a group company, less likely to be family-owned, and less likely to export.

Finally, the concentration of bank business on one main bank could be driven by a firm's risk preferences. We approximate this with the share of a firm's business that is concentrated on the three most important customers (*Concentration*). Table 1.4 shows that firms that concentrate business on one main bank also have a high customer concentration, whereas firms that do not do so also spread their own business across customers.

1.3.5 Control Variables for Bank Characteristics

Estimating the effects of relationship banking on impairments of credit finance raises the need to control for bank-side effects. Empirical research has shown that bank balance sheet channels are an important determinant of firms' access to credit¹¹ and that banks differ in their business model with respect to being a relationship lender (e.g., Bolton, Freixas, Gambacorta, and Mistrulli (2013)).

To rule out estimation bias from bank-side factors, we use the fact the German banking system consists of different classifications of banks that differ in their financing structure, their lending behaviour during the financial crisis, and their business model with respect to being a relationship bank.¹² First, commercial banks are universal banks engaged in both corporate banking and investment banking. They are privately owned and only to a small extent deposit-financed. Second, savings banks are publicly owned and serve

¹¹See, for example, Ivashina and Scharfstein (2010), Popov and Udell (2012), Jimènez, Ongena, Peydro, and Saurina (2012), and deYoung, Gron, Torna, and Winton (2014).

 $^{^{12}\}mathrm{See}$ Hackethal (2004) for a detailed description of the German banking system.

public interest by taking in deposits and providing credit to borrowers within their region. Third, Landesbanken, which are also publicly owned, provide services to federal states and offer large scale funding to companies, which cannot be offered by small savings banks. The fourth classification contains cooperative banks. Like savings banks, they only operate within their region and are largely deposit-financed. Cooperative banks are owned by their members, whose interest they serve.

	Commercial	Savings bank	Landesbank	Cooperative	Other
Freq.	401	373	63	214	93
Perc.	61.41%	57.12%	9.65%	32.77%	14.24%
Impaired	21.45%	21.18%	23.81%	17.76%	23.66%
Information	15.71%	13.94%	17.46%	12.15%	16.13%
Reduction	6.98%	6.43%	3.17%	5.61%	5.38%
Availability	10.22%	11.26%	7.94%	8.88%	8.60%
Interest	11.22%	10.19%	15.87%	9.81%	13.98%
Maturities	2.74%	2.41%	4.76%	0.93%	3.23%
Collateral	10.47%	9.92%	4.76%	11.68%	12.90%
Others	3.49%	3.22%	4.76%	2.34%	9.68%
Main bank (1)	37.66%	41.82%	30.16%	45.33%	46.24%
Main bank (2)	38.15%	42.36%	38.10%	41.59%	30.11%
Main bank $(0/3+)$	24.19%	15.82%	31.75%	13.08%	23.66%

Table 1.5: Bank classifications, impaired credit finance, and relationship banking

Note: The table shows descriptive statistics for bank-side variables; *Commercial, Savings bank, Landes-bank, Cooperative*, and *Other* equal one if a firm reports that at least one of its two most important banks belongs to the respective classification of banks.

The data provides the classification of the two most important banks of each firm. We expect this to be a sufficient control variable for bank-side factors for two reasons. First, banks' attitude towards being a relationship lender is unlikely to differ between individual banks within each classification so that no further bias reduction stems from controlling for bank-level data. Second, the majority of firms in our sample have one or two main banks so that it is reasonable to assume that impairments of credit finance are primarily

determined by the two most important banks. Nevertheless, we use control variables from bank balance sheets as a robustness check in Section 1.5.3.

Table 1.5 shows that impaired credit finance (*Impaired*) is most likely for firms that have a Landesbank and the least likely for firms that have a cooperative bank among the two most important banks. This was to be expected because Landesbanken were seriously affected by the financial crisis and therefore most likely to cause impairments of firms' credit finance. Savings banks and cooperative banks are expected to be the least likely to do so because they are largely deposit-financed and Ivashina and Scharfstein (2010) show, albeit for U.S. data, that deposit-financed banks reduced lending to firms less drastically in response to the financial crisis than other banks. Furthermore, Table 1.5 provides clear indications that different kinds of impairments are associated with having different classifications of banks among the two most important bank relationships.

Table 1.5 also shows that the fraction of firms that do not implement relationship banking is highest among firms with commercial banks and Landesbanken among the two most important banks. Relationship banking is more widespread among firms for which savings banks and cooperative banks are most important. This could be explained differences in business models between banks. Savings banks and cooperative banks serve customers only within their region, which makes the establishment of close long-term bank-firm relationships likely. In addition, firms that rely on Landesbanken and commercial banks could maintain more bank relationships because they anticipate difficulties in getting credit from these banks. According to the theoretical model of Detragiache, Garella, and Guiso (2000), firms with only one bank relationship face the risk of adverse selection if the relationship bank cannot roll over loans due to liquidity constraints and a firm has to address banks that have not learned about a firm's credit quality before. Establishing multiple bank relationships reduces this risk.

In sum, Table 1.5 shows that bank-side factors affect both a firm's risk of experiencing different kinds of impairments of credit finance and its probability of following the concept of relationship banking. Therefore, controlling for these bank-side factors is crucial to rule out that they induce bias in the estimated effects of relationship banking on the probability of impairments of credit finance.

1.4 Methodology and Results

1.4.1 Methodology

To investigate the channels of relationship banking, we run separate OLS estimations in which the dependent variables indicate the different kinds of impairments as listed in Table 1.2 in Section 1.3.2. For example, to test whether relationship banking affects the probability of facing higher information requirements by banks due to the financial crisis, we estimate the linear probability model

$$Information_{i} = \alpha_{0} + \alpha_{1}MB_{i} + \alpha_{2}Firm_{i} + \alpha_{3}Bank_{i} + \alpha_{4}Industry_{i} + \epsilon_{i}$$
(1.1)

where MB_i is a set of the two dummy variables *Main bank (1)* and *Main bank (2)* so that firms whicht do not use relationship banking by having zero or three and more main banks constitute the baseline category. $Firm_i$ is the set of firm characteristics listed in Table 1.4 in Section 1.3.4, $Bank_i$ is the set of dummy variables for having the respective classification of banks among the two most important banks, and $Industry_i$ is a set of industry dummy variables based on the two-digit WZ 2008 industry classification.

1.4.2 Relationship Banking and Impairments of Credit Finance

Confirming Hypothesis 1, the results in Estimation (1) in Table 1.6 show that the probability of facing higher information requirements by banks due to the financial crisis is about 15 percentage points lower for firms that concentrate on one main bank relationship relative to firms that do not follow the concept of relationship banking. The effect of concentrating business on two main banks is also statistically significant, but substantially smaller. The difference between the two coefficients is not statistically significant.¹³ This result supports the view that information provision through relationship banking

¹³Throughout the estimations in this study, the difference between the estimated coefficients of *Main* bank (1) and *Main* bank (2) turn out to be statistically insignificant, which is why test statistics are not reported for the sake of conciseness.

limits asymmetric information, which facilitates the bank's screening process and interim monitoring. This reduces the necessity to require more information from the firm.

Not supporting Hypothesis 2, Estimations (2) and (3) show that relationship banking does not affect a firm's credit availability. The number of main banks does not have a significant impact on a firm's probability of experiencing the reduction of existing lines of credit or constrained availability of new loans or lines of credit. This could be explained by the banks' credit granting decisions being primarily based on credit risk models. Since these models only process hard information and do not account for soft information, it is reasonable that we do not find a statistically significant effect of relationship banking on a firm's credit availability.

Furthermore, Estimation (4) indicates that relationship banking lowers the likelihood of an increased interest rate for existing loans or lines of credit by almost 10 percentage points. The effect of having two main banks, however, is even larger, which is not entirely in line with Hypothesis 3. A potential explanation for this result stems from the fact that interest rates are subject to a bargaining process between a bank and a firm. Having a second main bank may improve a firm's bargaining position and prevent hold-up problems faced by firms with only one main bank.

Finally, we assess how impairments of non-price terms of credit are affected by relationship banking (Hypothesis 4). According to Estimation (5), having only one main bank lowers the probability that banks offer credit only at shorter maturities significantly by almost 5.5 percentage points. When looking at higher collateral requirements in Estimation (6), the effect is even larger at 7.9 percentage points. There is no such effect for firms with two main banks, which indicates that relationship banking affects non-price terms of credit differently than interest rates. In contrast to interest rate payments, these non-price terms are not a pecuniary transfer and do not immediately affect the profit of the bank. Instead, they are part of the risk-sharing process between the bank and a firm, which does not seem to be affected by hold-up problems and their resolution through competition between banks. One could also argue that soft information from relationship banking plays a larger role in the risk-sharing process than in bargaining about interest rates.

	(1)	(2)	(3)	
	Information	Reduction	Availability	
Main bank (1)	-0.1543***	0.0017	-0.0371	
	(0.05)	(0.03)	(0.04)	
Main bank (2)	-0.1044**	0.0125	-0.0298	
	(0.05)	(0.03)	(0.04)	
			()	
Firm char.	Yes	Yes	Yes	
Industry	Yes	Yes	Yes	
Bank class.	Yes	Yes	Yes	
Adj. R^2	0.0522	0.0164	0.0441	
N	652	652	652	
				-
	(4)	(5)	(6)	_
	(=) Interest	(0) Maturities	(0) Collateral	
	11001030	1/10/07/0005	Contaicrat	
Main bank (1)	-0 0990**	-0.0547**	-0 0789**	
	(0.0330)	(0.02)	(0.04)	
	(0.01)	(0.02)	(0.01)	
Main bank (2)	-0 1122***	-0.0455**	-0.0472	
	(0.04)	(0.02)	(0.04)	
	(0.01)	(0.02)	(0.01)	
Firm char.	Yes	Yes	Yes	
Industry	Yes	Yes	Yes	
Bank class.	Yes	Yes	Yes	
Adj. R^2	0.1059	0.0183	0.0275	
N	652	652	652	

Table 1.6: Baseline OLS estimations

Note: The table shows OLS estimation results for regressions of dummy variables for different kinds of impairments of credit finance due to the financial crisis on the number of main bank relationships, a set of control variables for firm characteristics, bank classification dummy variables, and industry dummy variables; the baseline category for the main bank relationship dummy variables is *Main bank* (0/3+); robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

In sum, firms that implement relationship banking have a lower probability of facing impairments of information requirements and non-price terms of credit due to the financial crisis. Credit availability, however, is not affected by relationship banking. For interest rates, we find a positive effect of concentrating business on one main bank, but evidence suggests that having a second main bank is even more advantageous.

1.4.3 Other Determinants of Impairments of Credit Finance

Table A.3 in the Appendix shows the coefficients of all control variables. As expected, the probabilities of all kinds of impairments of credit finance are lower for firms in a better financial situation. Furthermore, export activity and incorporation increase the probability of facing higher interest rates.

In addition, Table A.3 provides the estimated effects of having different classifications of banks among the two most important banks. Most of the effects are statistically insignificant, so that the results provide only weak evidence of bank-side effects on the firms' risk of facing different kinds of impairments of credit finance.

1.5 Robustness Checks

1.5.1 Estimation of Nonlinear Models

Instead of estimating a linear probability model, the effects of relationship banking could be estimated using nonlinear estimation procedures based on the following model:

$$Pr(Information_i = 1) = \Phi(\alpha_0 + \alpha_1 M B_i + \alpha_2 Firm_i + \alpha_3 Bank_i + \alpha_4 Industry_i) \quad (1.2)$$

where $\Phi(\cdot)$ is the cumulative distribution function of a standard normal distribution. Table A.4 in the Appendix shows that the binary probit marginal effects differ somewhat in size from the OLS estimators in Table 1.6. However, they confirm our findings that relationship banking works through information requirements and non-price terms of credit, but does not affect credit availability. The results for interest rates remain ambiguous. Binary probit estimations, however, induce the problem that many observations are dropped because some control variables (mainly the industry dummy variables) perfectly predict the value of the outcome variable. Table A.4 in the Appendix shows that this is particularly problematic in Estimation (4) when using *Maturities* as the dependent variable. Therefore, linear probability models are preferable as baseline estimations.

1.5.2 Addressing Potential Reverse Causality

Challenging our interpretation of the results in Section 1.4, one could argue that firms have reacted to impaired credit finance by setting up further main bank relationships, which would raise endogeneity concerns. From this perspective, our estimations would suggest that firms with a high probability of impaired credit finance maintain more main bank relationships, instead of vice versa.

First, an argument against reverse causality can be derived from the characteristics of main bank relationships. As Table A.2 in the Appendix shows, a long duration of the bank relationship is a key criterion for a firm to refer to the respective bank as a main bank. Even if firms had added business relationships to banks when experiencing impaired credit finance, these would not constitute long-term relationships. Therefore, it is unlikely that this increased the number of main banks reported in the survey.

Second, we test for the impact of reverse causality by re-running all estimations after excluding firms with potential dynamics in their main bank relationships. The survey data contains the durations of the two most important bank relationships of every firm. We therefore know whether these were established during the financial crisis, or before. This allows us to identify firms with dynamics in the structure of their bank relationships, which may have been caused by the financial crisis and its impact on a firm's credit finance. In total, 153 firms are dropped from the sample because their first or second most important bank relationship was established in 2007 or later.¹⁴

¹⁴The reduction in the number of observations used in the estimations is smaller as some of the dropped firms were not considered in the first place because of missing values.
Relationship Banking during the Financial Crisis

If reverse causality would drive our results, dropping these firms should affect the estimated effects of relationship banking. The comparison of the results in Table A.5 in the Appendix to Table 1.6 shows that this is not the case. For the firms left in the sample, we find that relationship banking affects credit finance by lowering the probability of higher information requirements and impaired non-price terms of credit due to the financial crisis. We do not find any effects of relationship banking on credit availability. The ambiguous results for interest rates do not change either. Therefore, estimations without firms with young important bank relationships allows us to rule out that our findings are driven by reverse causality.

1.5.3 Controlling for Bank Balance Sheet Data

Another problem of the estimations in Section 1.4 could stem from the fact that the bank classification dummy variables do not rule out estimation bias from bank-side factors entirely. If banks within the different classifications differ with respect to their lending during the financial crisis (as shown for German savings banks by Puri, Rocholl, and Steffen (2011a)) and in their attitude towards being a relationship bank, our results could be driven by remaining heterogeneity in bank characteristics.

To test the robustness of our results with respect to bank heterogeneity within bank classifications, we re-run our estimations using bank balance sheet variables as control variables instead of bank classification dummy variables. Unfortunately, data from the Ifo "Financing of the German Economy" survey does not allow the identification of the banks to which firms maintain business relationships. Therefore, we draw this information from the BvD Amadeus database. We then link the balance sheet of each of these banks to the firm-level data set.

We control for banks' total assets, their equity ratio, their liquidity ratio, and their deposit-financing ratio. Each ratio is calculated relative to total assets. We measure banks' financial situation during the entire financial crisis by calculating the average of these balance sheet items over each bank's balance sheets from the years 2007-2009.

	Main bank (1)	Main bank (2)	Main bank $(0/3+)$
Ν	649	649	649
Freq.	266	251	132
Perc.	40.99%	38.67%	20.34%
$log(Bank \ assets)_{max}$	11.8	12.4	13.5
$log(Bank \ assets)_{min}$	8.2	8.1	8.7
Bank $equity_{max}$	6.03	6.06	5.83
$Bank \ equity_{min}$	3.28	2.98	2.34
Bank liquidity _{max}	27.44%	28.32%	31.91%
$Bank \ liquidity_{min}$	16.99%	14.67%	13.50%
$Bank \ deposits_{max}$	65.53%	67.31%	63.29%
$Bank \ deposits_{min}$	41.58%	38.29%	28.25%

Table 1.7: Bank balance sheets and relationship banking

Note: The table shows descriptive statistics for bank balance sheet variables; for each variable, the maximum and minimum value across all banks to which a firm maintains a business relationship according to the BvD Amadeus database are presented.

As different firms work with different numbers of banks, we have to control for the financial situation of the whole portfolio of banks. We do so by using the maximum and minimum values for every variable across the balance sheets of each firm's banks.¹⁵ Table 1.7 provides descriptive statistics for the bank balance sheet variables depending on a firm's number of main bank relationships. We find that firms that follow the concept of relationship banking work with somewhat smaller banks. Their banks also have higher equity ratios and rely more on deposit financing. The picture for banks' liquidity is mixed as minimum and maximum liquidity across a firm's banks show opposing correlations with relationship banking.

When controlling for bank balance sheet variables instead of using bank classification dummy variables, estimation results in Table A.6 in the Appendix confirm most of our previous findings. Concentrating on one main bank relationship still reduces the risk of higher information requirements significantly. The effect of having two main banks is also statistically significant, but substantially smaller. There is, as before, no effect of relationship banking on credit availability and the results with respect to interest rates

¹⁵Without reporting the results in this chapter, we also run estimations using only maximum values, only minimum values, or mean values across the balance sheets of each firm's banks as control variables. The results for the effects of relationship banking on the different kinds of impairments of credit finance due to the financial crisis are not altered by using different approaches.

remain ambiguous. There is still a significantly lower risk of the impairment of maturities when following relationship banking. However, the risk of higher collateral requirements is no longer significantly affected by relationship banking. Table A.6 in the Appendix also provides the estimated coefficients of the different bank balance sheet variables. However, it is hard to interpret these because using minimum and maximum values leads to problems with multicollinearity between these variables. This, however, does not bias the estimated effects of relationship banking.

1.6 Conclusion

This chapter addresses the question through which channels relationship banking affects firms' credit finance during times of financial crisis. We test whether applying relationship banking by focusing business on one main bank relationship lowered a firm's risk of facing different kinds of impairments of credit finance due to the financial crisis of 2007-09. Our results show that relationship banking lowers the probability that a firm reports higher information requirements by banks due to the financial crisis. However, relationship banking does not significantly affect the impairment of firms' credit availability. For the interest rate, our results are ambiguous. Here, it seems that a firm benefits from maintaining only one main bank relationship, but having a second main bank is even more advantageous. For non-price terms (i.e. maturity and collateral), relationship banking lowers the probability of impairments significantly.

The effect of relationship banking on information requirements is the most direct evidence that banks gather information about firms from a bank-firm relationship. The divergent results for credit availability and the other channels of relationship banking might be due to the fact that banks use hard rather than soft information in their credit risk models on which they base their credit granting decision. If credit is granted, however, soft information from relationship banking affects the negotiation of the credit contract. This holds even more for non-price terms and conditions that determine the risk-sharing between banks and firms than for the pecuniary transfer constituted by interest payments.

Relationship Banking during the Financial Crisis

These findings underline that relationship banking was beneficial for firms (and thereby for the whole economy) during the financial crisis, although it was perceived as somewhat superseded before the crisis. Therefore, in times of uncertainty, regulators should facilitate the flow and use of information from bank-firm relationships. Many magazines and guidebooks for practitioners also picked up on this topic. For example, a report by Deloitte & Touche GmbH (2012) concludes that: "the quality of the relationship of medium-sized firms and their banks is often underrated and thereby neglected. In particular during a crisis, a long-term and positive relationship is of utmost importance." Our analysis provides strong support for this view.

Chapter 2

Friendship and Money, Oil and Water? Credit Constraints and "Family and Friends" Finance

2.1 Introduction

Facing credit constraints induces firms to seek other sources of funding. Recent empirical evidence shows that many firms used more public market finance¹ and trade credit² in response to the financial crisis of 2007-09. Alternatively, firms may seek capital from informal sources like "Family and Friends" (F&F) in response to bank credit constraints.³ When going informal, welfare gains from financial intermediation through banks no longer materialise. Djankov, Lieberman, Mukherjee, and Nenova (2003) argue that firms may then face higher borrowing costs and that "finance from friends and family is unreliable, untimely and can bear significant non-financial costs".

 $^{^1 \}mathrm{See},$ for example, Adrian, Colla, and Shin (2012), Barraza, Lee, and Yeager (2014), and Becker and Ivashina (2014)

 $^{^2 \}mathrm{See},$ for example, Carbó-Valverde, Rodríguez-Fernández, and Udell (2013) and Coulibaly, Sapriza, and Zlate (2013)

 $^{^{3}}$ Other informal sources such as moneylenders are widespread in developing countries, but are not common in developed economies.

Empirical evidence of the use of informal finance has been primarily provided in the context of the capital structure of start-up businesses in developed economies and financial systems in developing countries.⁴ However, not much attention has been paid to the question whether informal finance is also used by established firms in highly developed countries. Furthermore, previous studies analyse correlations, but do not estimate the causal effect of bank credit constraints on the use of informal finance.

This chapter analyses the use of F&F finance by non-start-up firms in a highly developed economy and tests whether it is driven by unsuccessful bank credit negotiations. The analysis is based on data from the Ifo "Financing of the German Economy" survey, which was conducted among German manufacturing firms in September 2011. The data distinguishes between F&F finance from sources connected to the business ("F&F Business") and sources privately connected to the entrepreneur ("F&F Private"). Descriptive statistics show that 15.46 percent of the firms in the sample use at least one of the two kinds of F&F finance, which is surprisingly high. The data also contains detailed information on each firm's last bank credit negotiations between 2008 and 2011. The causal effect of unsuccessful bank credit negotiations on the use of F&F finance is estimated using instrumental variables (IV) estimations to deal with endogeneity from a potential signalling effect of F&F finance on a firm's likelihood of receiving bank credit.

IV estimations show that firms use both kinds of F&F finance in response to unsuccessful negotiations of bank loans. This effect, however, is not found for unsuccessful negotiations of lines of credit. Therefore, F&F finance seems to be especially relevant in financing particular investments (substituting loans), but not as much in financing working capital (substituting lines of credit). The comparison of OLS and IV estimators indicates that "F&F Business" serves as a positive signal of a firm's creditworthiness in bank credit negotiations. This is in line with Berger and Udell (1998), who argue that

⁴Allen and Qian (2010), Allen, Qian, and Zang (2011), and Allen, Carletti, Qian, and Valenzuela (2013) provide theories on the finance-and-growth-nexus arguing that informal finance may have advantages over formal finance in supporting economic growth in developing countries. This is supported by the analysis of data from India (Allen, Chakrabarti, De, Qian, and Qian, 2012) and China (Allen, Qian, and Qian, 2005). Also using Chinese data, Ayyagari, Demirgüc-Kunt, and Maksimovic (2010), however, do not find a relationship between the use of informal finance and economic growth. Degryse, Lu, and Ongena (2013) find a positive complementary effect of formal and informal finance on sales growth for small firms in China, but not for large firms.

different financial instruments are interconnected and initial insider finance often serves as a "predicament" for receiving external finance. For "F&F Private", however, evidence of its signalling effect is ambiguous.

This study is most closely related to a strand of research on the use of F&F finance and other kinds of informal finance. The theoretical model by Myers and Majluf (1984) predicts that firms' financing decisions follow a pecking order. Because of asymmetric information, firms preferably rely on internal funding (i.e. retained earnings). If they need external finance, they prefer debt over equity instruments. However, Myers and Majluf (1984) do not explicitly state the role of F&F finance in the pecking order. Filling this gap, Berger and Udell (1998) find that "insider finance" in the form of debt or equity from the start-up team, family, and friends plays a role in the early stages of development, but is phased out as the firm matures. This has been supported by several empirical studies on the determinants of the use of different kinds of informal finance in start-up businesses and small firms.⁵ Altogether, these studies find that younger, smaller, less transparent, and less financially sound firms are more likely to use informal finance. However, such studies do not explicitly observe the presence of credit constraints.

In this regard, more direct evidence of the correlation between credit constraints and the use of informal finance is provided by Allen, Chakrabarti, De, Qian, and Qian (2012), who show that Indian firms use informal finance in response to "limited access to institutional finance". Using data on a broad set of firms from 48 countries in the World Business Environment Survey (WBES), Beck, Demirgüç-Kunt, and Maksimovic (2008) further find that there is a correlation between financing obstacles and the use of finance from informal lenders.⁶ Such correlations, however, do not allow a causal interpretation.

The following empirical analysis amends this literature by using an IV approach to estimate the causal effect of unsuccessful bank credit negotiations on the use of F&F finance. The comparison of OLS and IV estimators provides evidence of whether F&F finance is

⁵For analyses of start-up businesses, see, for example, Chavis, Klapper, and Love (2011), Sanyal and Mann (2010), Astebro and Bernhardt (2003), Basu and Parker (2001), Romano, Tanewski, and Smyrnios (2001), and Fluck, Holtz-Eakin, and Rosen (1998). For analyses of small firms, see, for example, Denis and Mihov (2003), Berger and Udell (2002), and Bitler, Robb, and Wolken (2001).

⁶Beck, Demirgüç-Kunt, and Maksimovic (2008) define informal finance as financing coming from "informal moneylenders and other traditional sources". They categorise F&F finance as internal finance and do not provide further evidence of its use.

a positive or a negative signal of a firm's creditworthiness in credit negotiations with banks. A particular contribution stems from the use of data on firms of all size and age groups in a highly developed economy without a focus on start-up businesses.

Since the use of F&F finance requires both demand and supply of this kind of funding, the following analysis can also be seen as a test of the theoretical predictions by Giannetti and Yu (2014). They develop a model to investigate lending and borrowing between connected sources, such as F&F finance. Their model predicts that "financiers allocate capital on the basis of prior connections, instead of collecting information on the [...] entrepreneur", depending on the initial capital, transparency, and the quality of investment opportunities in an economy. They show that, even in advanced economies (i.e. those economies with a high level of initial capital), high costs of information acquisition and low average quality of potential borrowers can prompt financiers to "forfeit information acquisition" and lend to "connected entrepreneurs" only.

This prediction is applicable to Germany during the financial crisis of 2007-09, which is captured by the sample used in this analysis. When the German economy was hit by the slowdown of global economic activity, many firms faced a sharp drop in demand for their products and uncertainty about firms' creditworthiness increased (International Monetary Fund, 2009). In the context of the model by Giannetti and Yu (2014) this can be interpreted as a deterioration of borrower quality and increasing information acquisition costs. The model predicts that depositors in Germany preferred to lend to "connected entrepreneurs" rather than acquiring information on potential borrowers. This is in line with data in this study providing evidence of a widespread supply of F&F finance in the German economy.

The remainder of the chapter is structured as follows. In Section 2.2, two hypotheses are derived from existing literature. Section 2.3 provides a description of the data set as well as descriptive statistics on the use of F&F finance and unsuccessful bank credit negotiations. Section 2.4 provides estimation results for OLS and IV approaches. Section 2.6 addresses sample selection issues and the role of discouraged borrowers. Finally, Section 2.7 summarises the findings.

2.2 Hypotheses

Existing empirical literature shows that firms' use of informal finance is correlated with credit constraints (e.g., Allen, Chakrabarti, De, Qian, and Qian (2012), Beck, Demirgüç-Kunt, and Maksimovic (2008)). This is in line with the pecking order theory predicting that firms first borrow from closely connected sources before moving towards arm's length finance. It also raises the question whether credit constraints actually drive firms into F&F finance or if the correlation can be explained by the signalling effect of F&F finance on a firm's probability of receiving bank credit. In particular, it is worth testing the significance of such a causal effect for established firms in a highly developed economy.

Hypothesis 1: Unsuccessful bank credit negotiations cause firms to use of F&F finance, even established firms in a highly developed country.

Resolving endogeneity issues raises the follow-up question whether the use of informal finance facilitates access to credit or if it makes it more difficult. A strand of literature provides evidence of the signalling effect of trade credit on access to bank credit because of trade partners' ability to assess a firm's creditworthiness (e.g., Giannetti, Burkart, and Ellingsen (2011), Engemann, Eck, and Schnitzer (2011)). In a similar manner, such a signalling effect could exist for F&F finance due to the information that connected firms and individuals gather about a firm. Banks could take the fact that firms receive funding from closely related sources as a positive signal of creditworthiness. This may be particularly important when asymmetric information is high, for example, because firms are very young, face high entrepreneurial risk, or do not have a track record of repaid debt.

Hypothesis 2: Using F&F finance is a positive signal of a firm's creditworthiness.

Alternatively, a firm's use of F&F finance could prompt a bank to abstain from granting credit for two reasons. First, the use of F&F finance could be a result of previous unsuccessful credit negotiations, possibly with other financial institutions. Second, from a bank's point of view, the use of F&F finance might carry a whiff of opaqueness, which could threaten the success of bank credit negotiations.

2.3 Data

2.3.1 The Data Set

The following analysis is based on data concerning firms' use of F&F finance, their recent bank credit negotiations, and numerous firm characteristics from the Ifo "Financing of the German Economy" survey, which is described in Chapter 1. The age structure of the firms in the sample is skewed towards older firms (see descriptive statistics in Section 2.3.4), which makes the data set particularly suited to analyse the use of F&F finance in a broad set of firms without a focus on start-up businesses.

Because the survey data contains only a small amount of information on firms' financial situations, it is complemented with 2011 firm balance sheet data from the Bureau van Dyk (BvD) Amadeus database and 2011 credit ratings from Creditreform, a German rating agency. A description of all variables is provided in Table 2.1.

2.3.2 The Use of F&F Finance in Germany

In the following, firms are divided into those that use F&F finance (F & F firms) and those that do not (*non-F \& F firms*). Considering that Germany is a highly developed economy and that the sample does not focus on start-up businesses, the fraction of F&F firms in the sample is surprisingly high at 15.46 percent. In comparison, capital market finance is used by 3.6 percent and factoring by 10 percent of the firms.

The data distinguishes two kinds of F&F finance. "F&F Business" is defined as a firm receiving capital from a firm or person close to the business (e.g., customers, suppliers). "F&F Private" indicates that capital is received from a person privately connected to the entrepreneur. "F&F Business" is used by 5.79 percent of the firms in the sample. "F&F Private" is about twice as important with 11.81 percent of the firms using it. Analysing both kinds of F&F finance separately provides additional insights as they potentially differ in their degree of informality. While "F&F Business" can be expected to be widely based on formal contracts, informal procedures are more likely for "F&F Private".

Variable	Description
	n
"Family and Friends" f	
FOF FOR :	Firm receives capital from "Family and Friends"
FOF Business	Firm receives capital from F&F close to the business (e.g.,
FRE Driverto	Eine received conital from EleE privately connected to the
FOF Private	entropropour
	entrepreneur
Bank Credit Negotiatio	ons
Rejected (line)	Last negotiated line of credit (since 2008) was rejected or
	only partially granted
Rejected (loan)	Last negotiated loan (since 2008) was rejected or only par-
	tially granted
Discretion	Loan officer had a larger impact on the credit granting de-
	cision than the bank-internal credit rating
Firm size and age	Number of our loss of
Empl	The sector secto
Assets	Total assets
Age	Firm age in years, based on year of foundation
Transparency	
Incorporated	Firm is a corporation by its legal status
Ext. rating	Firm has external rating besides banks' internal ones
Customer	Share of business that is conducted with the three most
	important customers
Export	Firm is exporting
Ownership	_
Group	Firm is part of a group company
Family	Largest shareholder is a single person or family
Control	Percentage share held by largest shareholder
Operating	Largest shareholder active in operative management
Financial condition	
Rating	Score between 100 (sound) to 600 (risky)
Equity	Equity/Assets
Long-term debt	Long-term debt/Assets
Cash	Cash/Assets
Return	ÉBIT/Turnover
Earlypay	Cash discounts drawn/Cash discounts offered to firm

Table 2.1: Variable descriptions

Sources: Ifo "Financing of the German Economy" survey, BvD Amadeus database, and Creditreform.

2.3.3 F&F Finance and Unsuccessful Bank Credit Negotiations

Data on the use of F&F finance can be linked to a firm's most recent bank credit negotiations. In total, 496 firms negotiated a line of credit between 2008 and 2011 and 510 firms negotiated a bank loan.⁷ For this analysis, negotiations are defined as unsuccessful if credit is not granted or granted only at a smaller volume than demanded by the firm. It is important to note that this encompasses rejections of a credit application by the bank, as well as situations in which a firm decides to withdraw the application because of unfavourable terms of credit offered by the bank (e.g., when interest rates are too high). Either scenario can be interpreted as a situation in which a firm faces credit constraints.

	Rejected (line)			Rejected (loan)		
	Yes	No	p > t	Yes	No	p > t
Ν	71	425		64	446	
Perc.	14.31%	85.69%		12.55%	87.45%	
$F \mathscr{C} F$	30.00%	18.75%	0.03^{**}	35.48%	15.67%	0.000***
			0.00*		10.000	0 000***
FOF Business	23.53%	14.56%	0.06^{*}	30.65%	13.02%	0.000^{***}
F&F Private	13.64%	6.47%	0.04^{**}	13.56%	4.71%	0.006^{***}

Table 2.2: F&F finance and credit negotiations

Notes: The table shows the fraction of F&F firms separately among firms that have successfully negotiated bank credit and among those that have negotiated unsuccessfully; p-values are reported for two-group mean comparison t-tests on whether the two groups of firms differ significantly with respect to the probability of using F&F finance; * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 2.2 shows that F&F finance is significantly more widespread among firms that have experienced unsuccessful bank credit negotiations. This holds for both "F&F Business" and "F&F Private". The difference is more pronounced for unsuccessful negotiations of bank loans than for those of lines of credit, which suggests that F&F finance is more important in financing particular investments than for working capital financing. Table B.1 in the Appendix shows that 72.3 percent of the negotiated lines of credit in the sample were used for working capital finance, but only 57.1 percent served the financing of investments. In contrast, loans were primarily used for investments with enhancements

 $^{^{7}}$ As firms reported negotiations of lines of credit and loans separately in the survey, each firm could report up to two bank credit negotiations between 2008 and 2011.

being the most frequent purpose. Only 15.1 percent of the loans were used for "other" purposes, which includes working capital finance. The analysis of firms' text-based specification of "other" purposes of loans show that only 1.5 percent of the loans were used for working capital finance. Debt restructuring and liquidity protection were the most prominent "other" purposes of loans.

2.3.4 F&F Finance and Firm Characteristics

A key challenge in the following estimations is to hold constant the firm characteristics which drive both the outcome of a firm's credit negotiations and its use of F&F finance. The data provides a broad set of control variables for a firm's creditworthiness, characteristics that are found in the empirical literature to determine a firm's capital structure in general, and predictors of the use of bank debt and informal finance in particular. Table 2.3 shows how F&F firms differ from non-F&F firms in these variables.⁸ Altogether, they are significantly smaller and in a worse financial condition. They also differ in ownership variables. Evidence of differences in firm age and transparency, however, is less clear.

Firm size and age

Firm size and age are widely accepted as determinants of a firm's capital structure. For example, small firms rarely use public market finance because they are less capable of dealing with high transaction costs (Titman and Wessels, 1988). In line with existing empirical literature on firm size and informal finance⁹, Table 2.3 shows that F&F firms are smaller in terms of the number of employees and total assets.

Firm age is another determinant of a firm's capital structure as older firms are more likely to have a track record of successful business activity and repaid debt. According to Table 2.3, F&F firms are younger than non-F&F firms, which confirms previous studies.¹⁰

 $^{^8{\}rm Table~B.2}$ in the Appendix further provides descriptive statistics separately for firms that use "F&F Business" and for those that use "F&F Private".

⁹See, for example, Allen, Chakrabarti, De, Qian, and Qian (2012), Chavis, Klapper, and Love (2011), Sanyal and Mann (2010), Denis and Mihov (2003), Berger and Udell (2002), Bitler, Robb, and Wolken (2001), Romano, Tanewski, and Smyrnios (2001), and Berger and Udell (1998).

¹⁰See, for example, Allen, Chakrabarti, De, Qian, and Qian (2012), Chavis, Klapper, and Love (2011), Berger and Udell (2002), Romano, Tanewski, and Smyrnios (2001), Fluck, Holtz-Eakin, and Rosen (1998), and Berger and Udell (1998).

		$F & F = \overline{1}$			$F & F = \overline{0}$		
	(N = 169)		(N=924)		
	\bar{X}_{FF}	X_{FF}^{med}	σ_{FF}	\bar{X}_{noFF}	X_{noFF}^{med}	σ_{noFF}	p > t
log(Empl)	4.61	4.46	1.75	5.02	4.83	1.85	0.008^{***}
log(Assets)	8.80	8.69	2.18	9.23	9.08	2.27	0.04^{**}
log(Age)	3.95	4.17	1.04	4.05	4.36	0.96	0.22
T , 1	CD CO ⁰⁷	1	10.0007	CO 0107	1	10 5007	0.60
Incorporated	63.69%	1	48.23%	62.01%	1	48.56%	0.68
Ext. rating	25.00%	0	43.43%	21.25%	0	40.93%	0.28
Customer (< 10%)	18.67%	0	39.09%	20.04%	0	40.06%	0.68
Customer $(10\% \text{ to } < 30\%)$	40.36%	0	49.21%	45.26%	0	49.80%	0.24
Customer $(30\% to < 50\%)$	25.30%	0	43.61%	19.16%	0	39.38%	0.07^{*}
Customer $(50\% +)$	15.66%	0	36.45%	15.53%	0	36.24%	0.97
Export	82.63%	1	37.99%	88.46%	1	31.97%	0.04^{**}
a	0C C907	0	44 9907	20 6507	0	10.0407	0 001***
Group	26.63%	0	44.33%	39.65%	0	48.94%	0.001***
Family	82.25%		38.32%	74.59%	1	43.56%	0.03**
Control	70.36%	75%	26.46%	73.83%	88%	28.95%	0.15
Operating owner	68.64%	1	46.53%	59.50%	1	49.12%	0.03**
Ratina	216.79	208	90.38	197.49	186	90.46	0.01**
Equity	22.78%	27.47%	47.84%	37.55%	38.08%	32.45%	0.000***
Debt	33.92%	22.73%	48.83%	22.82%	13.95%	29.15%	0.000***
Cash	9.21%	3 99%	12.81%	11 37%	5 32%	14.37%	0.11
Return (< 3%)	58.33%	1	49.46%	43.13%	0	49.55%	0.000***
Return $(3 \text{ to } < 7\%)$	28.21%	0	45.14%	33.18%	0	47.11%	0.22
Return (7 to $< 10\%$)	7.69%	0 0	26.73%	14.69%	0	35.42%	0.02**
Return $(10\% +)$	5.77%	0 0	23.39%	9.00%	0	28.64%	0.18
Earlypay (0%)	4.29%	0	20.34%	3.01%	0	17.11%	0.39
Earlypay (< 2.5%)	19.63%	Ő	39.84%	13.73%	0	34.43%	0.05*
Earlypay (25 to $< 50\%$)	6.75%	0	25.16%	9.38%	0	29.16%	0.28
Earlypay (50 to $<75\%$)	16.56%	Ő	37.29%	9.38%	0	29.16%	0.006***
Earlupay (75% +)	52.76%	1	50.08%	64.51%	1	47.88%	0.004***

Table 2.3: F&F finance and firm characteristics

 $\frac{Earlypay(15\% + f)}{\text{Notes: The table shows descriptive statistics for firm characteristics separately for F&F firms and non-F&F firms; p-values are reported for t-tests with <math>H_0$: $\bar{X}_{FF} = \bar{X}_{noFF}$; * p < 0.1, ** p < 0.05, *** p < 0.01.

Transparency

Furthermore, transparency affects a firm's capital structure and the use of (informal) funding from personal sources (Berger and Udell, 1998). First, incorporation measures transparency because it determines a firm's publication obligations, which affect access to bank debt and public debt (Cassar, 2004; Ang, 1992) as well as informal finance (Chavis, Klapper, and Love, 2011; Sanyal and Mann, 2010; Bitler, Robb, and Wolken, 2001). Second, having an external rating indicates transparency and affects a firm's capital structure (Croci, Doukas, and Gonenc, 2011; Faulkender and Petersen, 2006; Denis and Mihov, 2003). Additionally, customer concentration is used as a proxy for transparency as firms with widespread customers have a higher interest in transparency. Finally, export business can increase uncertainty and thereby requires a firm to be transparent. The data in Table 2.3 shows that F&F firms do not differ significantly from non-F&F firms in these variables, except that they are significantly less likely to export.

Ownership

Empirical literature provides evidence that a firm's ownership status also affects its capital structure. First, belonging to a group company reduces the need to use external finance as more internal funds are available (Masulis, Pham, and Zein, 2011). Second, family businesses and non-family businesses differ in their capital structures.¹¹ Furthermore, capital structure decisions can be affected by the owner's control rights (Croci, Doukas, and Gonenc, 2011; Stulz, 1988) and by whether the largest shareholder of the firm is a member of the operating board (Chavis, Klapper, and Love, 2011).

As Table 2.3 shows, F&F firms are significantly less likely to belong to a group company. They are more likely to be family-owned and the largest shareholder is more likely to be part of a firm's operating board. The share of the largest shareholder is somewhat smaller among F&F firms, but the difference is not statistically significant.

¹¹See, for example, Chua, Chrisman, Kellermanns, and Wu (2011), Croci, Doukas, and Gonenc (2011), Ellul (2010), Romano, Tanewski, and Smyrnios (2001), and Coleman and Carsky (1999).

Financial condition

Finally, a set of variables measured in 2011 is used to assess the impact of a firm's financial situation on its capital structure (Denis and Mihov, 2003; Lemmon and Zender, 2010). According to Table 2.3, F&F firms are more risky in terms of the Creditreform rating, have a lower equity ratio, more long-term debt, and less cash. Variables for a firm's return on sales and the fraction of early payment discounts that the firm was able to draw underline that F&F firms are in a worse financial condition than non-F&F firms.

2.4 Methodology and Results

2.4.1 Methodology

Ordinary least squares (OLS) estimations

In a first step, OLS estimations are applied to rule out that the correlation of unsuccessful bank credit negotiations and F&F finance is driven by firm heterogeneity. Although the dependent variables in the following estimations are binary, a linear probability model (LPM) is applied instead of a non-linear estimation to make the results comparable to the linear IV estimations below. Estimations are therefore based on the following linear model:

$$F\&F_i = 1 = \beta_0 + \beta_1 Rejected_i + \beta_2 X_i + \epsilon_i \tag{2.1}$$

where X_i contains control variables for firm characteristics listed in Table 2.3 and industry dummy variables based on the two-digit WZ 2008 industry classification. The latter rule out potential industry effects on the firms' capital structure and the use of F&F finance (Bitler, Robb, and Wolken, 2001; Romano, Tanewski, and Smyrnios, 2001). Rejected_i stands either for Rejected (line) or Rejected (loan).

The instrumental variable: Discretionary lending

OLS estimators of the effect of unsuccessful bank credit negotiations on the use of F&F finance do not allow a causal interpretation. As discussed in Section 2.2, receiving capital from a connected source could serve as a positive (or negative) signal of a firm's credit-worthiness and increase (or decrease) its chances of receiving bank credit. In both cases, OLS estimations would be affected by endogeneity.

Identification of a causal effect can be achieved by applying an IV approach, which requires an instrumental variable that satisfies three conditions. First, it must be a relevant predictor of the outcome of bank credit negotiations. Second, it must not affect a firm's decision to use F&F finance, except through the outcome of bank credit negotiations. Finally, the instrumental variable has to be independent of the use of F&F finance conditional on the control variables.

In the following, a variable indicating that a bank applies discretionary lending in the decision about granting credit provides an instrumental variable that satisfies all three conditions. In practice, banks can decide about credit applications based on a rules-based and pre-codified lending process or leave the credit granting decision to the discretion of the loan officer. In the Ifo "Financing of the German Economy" survey, firms that have negotiated bank credit report whether the loan officer had a larger impact on the credit granting decision than a firm's credit rating. The variable $Discretion_i$ equals one if this was the case and therefore indicates that the bank applied discretionary lending.

There are two arguments why discretionary lending is a relevant predictor of bank credit negotiation outcomes in line with the first condition of an instrumental variable. First, in rules-based lending, banks decide whether to lend based on credit scores that are derived only from hard information about the firm (e.g., balance sheet information). If discretionary lending is applied, soft information is also considered in the credit granting decision. Therefore, a broader range of information is collected and taken into account, which lowers uncertainty for the bank and improves its ability to price the credit contract. This increases the probability that the bank grants credit given a firm's creditworthiness. Second, discretionary lending provides more room for credit decisions to be based on the flow of information from the personal relationship between the loan officer and the

customer. This increases the probability of a firm receiving credit, in particular in times of constrained credit supply (e.g., Sharpe (1990); von Thadden (2004)). How discretionary lending predicts a bank's credit granting decisions is further discussed by Puri, Rocholl, and Steffen (2011b). Their empirical analysis based on a large data set of loans to retail customers shows that discretion in the loan approval decision increases the number of customers that receive credit, which further supports the relevance of discretionary lending as an instrumental variable in the following estimations.

The second condition is also satisfied by discretionary lending as the decision process is deeply rooted in the organisational structure and operational culture of the bank. Therefore, the presence of discretionary lending in a bank affects a firm's financing decisions (e.g., about F&F finance) only through the result of bank credit negotiations.

Whether discretionary lending is independent of F&F finance conditional on the control variables remains the key assumption in the following IV estimations. Gropp, Gruendl, and Guettler (2013) show that certain firms select themselves into borrowing from discretionary lenders, for example, depending on their creditworthiness. Such selection behaviour can be ruled out as affecting the IV estimations in this study because firms' credit risk and their financial situation is controlled for by a large set of control variables. Proving the validity of the instrumental variable beyond factors held constant by control variables, the discussion in Section 2.5 provides further evidence of F&F firms not selecting themselves into negotiations with banks that are likely to apply discretionary lending.

IV estimations: Avoiding the "forbidden" regression

Endogeneity enters the basic model in Equation 2.1 through $Rejected_i$, which is a binary variable. This raises concerns about the "forbidden regression" in which predicted values from a non-linear first-stage estimation are used in an IV approach. Angrist and Pischke (2008), and Wooldridge (2010) suggest the following estimation approach to make use of potential non-linearity in the conditional expectations function.

The first step contains a binary probit estimation of the model

$$Pr(Rejected_i = 1) = \Phi(\alpha_0 + \alpha_1 X_i + \alpha_2 Discretion_i)$$
(2.2)

where $\Phi(\cdot)$ is the cumulative distribution function of a standard normal distribution. X_i contains all control variables and industry dummy variables, and *Discretion_i* is the instrumental variable. Two separate estimations are run for *Rejected (line)* and *Rejected* (loan).¹² Predicted values for the probability of unsuccessful bank credit negotiations are calculated from both estimations. Following Angrist and Pischke (2008), and Wooldridge (2010), these predicted values are used as instrumental variables for *Rejected (line)* and *Rejected (loan)*, respectively, in two-stage least squares estimations.

2.4.2 Results: Ordinary Least Squares Estimations

According to OLS estimations in Table 2.4, the use of F&F finance is not significantly affected by unsuccessful negotiations of a line of credit. The effect of unsuccessful negotiations of a loan, however, is substantially larger and statistically significant. Therefore, F&F finance is more important to firms that need to finance particular investments than to those that need working capital finance.

Table 2.4 further suggests that the effect is driven by "F&F Private", which is a more important substitute for unsuccessfully negotiated bank credit than "F&F Business". In Estimations (3) and (4), unsuccessful bank credit negotiations do not have a significant effect on "F&F Business". Furthermore, unsuccessful negotiations of lines of credit do not affect the probability that a firm uses "F&F Private" in Estimation (5). Estimation (6), however, confirms that unsuccessful negotiations of a bank loan are associated with a significantly higher probability of using "F&F Private".

¹²The results for the two estimations are presented in Table B.3 in the Appendix. The coefficients of the variable $Discretion_i$ are large and highly statistically significant in both estimations, which underlines the relevance of the instrumental variable in predicting the outcome of bank credit negotiations.

	(1)	(2)	(3)	(4)	(5)	(6)
	$F \mathscr{C} F$					
			Business	Business	Private	Private
Rejected (line)	0.053		0.009		0.072	
	(0.09)		(0.07)		(0.08)	
Rejected (loan)		0.178^{*}		0.097		0.160^{*}
		(0.09)		(0.06)		(0.08)
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.1480	0.1116	0.1360	0.1306	0.1497	0.1354
Ν	289	308	279	303	284	304

Table 2.4: OLS estimations

Notes: The table shows results for six separate OLS estimations; firm characteristics comprise all control variables listed in Table 2.3; industry dummy variables are included based on the two-digit WZ 2008 industry classification; robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

2.4.3 Results: Instrumental Variables Estimations

Table 2.5 provides results for IV estimations ruling out endogeneity. Estimation (1) shows that unsuccessful negotiations of a line of credit do not cause firms to use F&F finance. According to Estimation (2), however, firms use F&F finance as a substitute for unsuccessfully negotiated bank loans. Estimations (4) and (6) underline that this substitution effect is driven by both kinds of F&F finance. Therefore, Hypothesis 1 is confirmed for bank loans, but not for lines of credit. This underlines that F&F finance is used by firms as an alternative source of funding for particular investments, but not for working capital finance.

The summary statistics of the first-stage regression in Table 2.5 show that the F-statistic is above ten for all estimations.¹³ The test criteria suggested by Stock and Yogo (2005) are satisfied at the ten percent level in all estimations. Therefore, discretionary lending provides a sufficiently strong instrument for unsuccessful bank credit negotiations. Comparing the different estimations, it seems to be a stronger instrument for unsuccessful negotiations of a line of credit than for unsuccessful negotiations of a loan.

 $^{^{13}}$ A critical value of ten is suggested as a rule of thumb by Staiger and Stock (1997).

	(1)	(2)	(3)	(4)	(5)	(6)
	$F \mathscr{C} F$					
			Business	Business	Private	Private
Rejected (line)	0.219		0.137		0.041	
	(0.21)		(0.16)		(0.17)	
Rejected (loan)		0.634^{**}		0.430**		0.480^{*}
		(0.31)		(0.21)		(0.27)
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
F	33.21***	15.16^{***}	34.55^{***}	12.89^{***}	31.49^{***}	15.91***
MES	35.81^{***}	18.08***	35.94***	17.14^{***}	35.34***	18.49^{***}
Ν	234	225	225	220	230	222

Table 2.5: IV estimations

Note: The table shows results of separate IV estimations; firm characteristics comprise all control variables listed in Table 2.3; industry dummy variables are included based on the two-digit WZ 2008 industry classification; robust standard errors are reported in parentheses; F is the F-statistic for the first-stage estimation testing for weak instruments; * p < 0.1, ** p < 0.05, *** p < 0.01; MES provides the minimum eigenvalue statistic for the Stock and Yogo (2005) test for weak instruments with *** indicating significance at the 10 percent level.

2.4.4 Signalling Effect of F&F Finance

The comparison of the estimated effects from IV estimations to those from OLS estimations sheds light on whether F&F finance is taken by banks as a positive or a negative signal of a firm's creditworthiness (see Hypothesis 2). If it was a positive signal, its use would be negatively correlated with the probability of unsuccessful credit negotiations, which would lead OLS estimators to underestimate the effect of unsuccessful credit negotiations of F&F finance. IV estimations would therefore lead to higher estimated effects than OLS estimations. If F&F finance was a negative signal, the opposite would be the case and IV estimations would reduce estimated effects relative to OLS estimations.

Compared to OLS estimation, the impact of unsuccessful bank credit negotiations on the use of "F&F Business" increases in both IV estimations, which suggests that "F&F Business" is taken as a positive signal of a firm's creditworthiness. This result supports Hypothesis 2. For "F&F Private", however, the picture is mixed with estimated effects decreasing in IV estimations of the effect of unsuccessfully negotiated lines of credit, but increasing in estimations of the effect of unsuccessfully negotiated loans.

This shows that receiving capital from business-related sources (e.g., suppliers or customers) is a stronger positive signal of creditworthiness than receiving capital from privately connected sources. The latter could even be a negative signal. This is reasonable as business-related lenders might be more motivated and better able to assess a firm's business activity, and therefore its creditworthiness than persons who are only privately connected to the entrepreneur. In contrast, the latter may have the incentive to provide capital for altruistic reasons even if creditworthiness is low. Furthermore, the degree of informality could be higher for "F&F Private", which could induce banks to abstain from lending to the firm.

2.5 Discussion of the Instrumental Variable

Using discretionary lending as an instrumental variable for unsuccessful bank credit negotiations is valid if discretionary lending and the use of F&F finance are independent conditional on the control variables included in the estimations. In the following, the data is used to test the satisfaction of this condition by analysing under which circumstances discretionary lending is applied by a bank and whether these are more likely to be prevalent for F&F firms.

First, the data allows to test whether F&F firms select themselves into negotiations with banks that are particularly likely to apply discretionary lending. Whether banks do so depends primarily on their business model. As described by Hackethal (2004), the German banking system comprises several classifications of banks, which differ substantially in their business models. Private commercial banks are typically universal banks offering a wide range of financial products. Savings banks are publicly owned and focus on taking in deposits and providing credit to the economy within their region. Landesbanken serve as central banks to savings banks and provide large scale funding that cannot be offered by small savings banks. Cooperative banks have a business model comparable to the one of savings banks, but they are mutually owned by their members, whom they serve. Finally, the data contains a category of "Other banks" that comprises banks that do not fall into either of these categories (e.g., foreign banks).

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Commercial	0.0022						-0.0559
Savings bank	(0.04) 0.0727						(-0.70) 0.0177
Landesbank	(1.15) 0.1643^*						(0.21) 0.1460
Cooperative	(1.84) 0.0684						(1.43) 0.0074
log(Longth)	(0.96)	0 0204					(0.08) 0.0300
		(-1.06)	0.0150				(-1.42)
log(Distance)			-0.0158 (-0.80)				-0.0132 (-0.55)
Credit (%)				0.0014^{**} (2.62)	**		0.0010^{*} (1.72)
Meeting				· · · ·	0.0864 (1.35)		0.0811 (1.23)
Vol. $(0.1 \text{ to } < 0.2)$	25)				(1.00)	0.0955	(1.20) 0.0863 (1.18)
Vol. $(1 \text{ to } < 5)$						(1.44) 0.0084	(1.18) 0.0256
Vol. $(5 \text{ to } < 50)$						(0.14) -0.0176	$(0.37) \\ 0.0836$
Vol. (50+)						(-0.25) -0.0472	(1.04) 0.0155
						(-0.57)	(0.15)
Ν	1,125	1,099	1,110	1,073	1,114	1,340	982

Table 2.6: Determinants of Discretionary Lending

Notes: The table shows results for OLS estimations in a sample of all credit negotiations of all firms since 2008; the dependent variable in all estimations is the dummy variable indicating that the bank decided about granting credit based on discretionary lending; the baseline category for the bank classification dummy variables is "Other banks"; relationship length is measured in years; distance is measured in travel minutes from the headquarters of the firm to the bank; the variable credit provides the share of credit a firm has with the bank; the dummy variable meeting indicates that the firm meets with the bank in person on a monthly basis rather than less frequently; "Vol." denotes the volume of credit measured in million EUR with "Vol. (< 100)" being the baseline category; standard errors are clustered at the firm-level and reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

To figure out whether different classifications of banks are differently inclined to apply discretionary lending, the firm-level data from the Ifo "Financing of the German Economy" survey can be transformed into a data set of the last bank credit negotiations of all firms. Each firm reports the last negotiations of a line of credit and the last negotiations of a bank loan so that up to two observations per firm are available. For every negotiation, firms could report with which classification of banks these negotiations were held. This information can be linked to whether the credit granting decision was made based on discretionary lending.

The application of discretionary lending may also be driven by certain firm-bank relationship characteristics, which are provided in the data. First, the relationship length and the geographical distance (measured by the travel distance in minutes) could drive discretionary lending because a long relationship and proximity might lead to close personal ties. Other factors could be the fraction of credit that the firm proceeds through the bank and whether a firm's banker is met in person on a monthly basis or less frequently. In addition to these firm-bank relationship characteristics, the impact of negotiated credit volume on the likelihood of discretionary lending can be assessed.

Based on all credit negotiations of all firms, Estimation (1) in Table 2.6 shows that discretionary lending is significantly more likely for Landesbanken with "Other banks" being the baseline category. When including all variables in Estimation (7), the effect turns insignificant. Furthermore, there is no evidence that savings banks and cooperative banks are particularly likely to apply discretionary lending. For private commercial banks, the probability is somewhat lower, but the difference is not statistically significant. Among the firm-bank relationship characteristics, only the fraction of credit proceeded through the bank seems to drive discretionary lending. The effect of monthly meetings is insignificant, but very stable between Estimations (5) and (7).

Therefore, if F&F firms selected themselves into negotiations with discretionary lenders, they should be more likely to negotiate with Landesbanken, banks through which they proceed a large share of credit, and banks with which they meet on a monthly basis. Table 2.7 shows regressions of variables indicating the classification of the bank with which a firm negotiated on whether a firm uses F&F finance. The results show that F&F

	(1)	(2)	(3)	(4)	(5)
	Commercial	Savings bank	Landesbank	Cooperative	Other bank
F & F	-0.056	0.019	0.028	0.071*	-0.062***
	(0.05)	(0.04)	(0.03)	(0.04)	(0.01)
Ν	1,113	1,113	$1,\!113$	1,113	1,113

Table 2.7: F&F firms, credit negotiations, and bank classifications

Notes: The table shows results for OLS estimations in a sample of all credit negotiations of all firms since 2008; standard errors are clustered at the firm-level and reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

firms are not more likely to negotiate with Landesbanken. They tend to select themselves away from negotiations with "Other Banks" and towards cooperative banks, but these do not differ in their attitude towards discretionary lending.

Furthermore, Table 2.8 shows that the firm-bank relationships in which F&F firms negotiate bank credit are not characterised by a particularly high share of credit held with the bank or a high likelihood of monthly meetings. Therefore, the circumstances under which discretionary lending is applied are not significantly more or less likely to be in place for F&F firms. Hence, neither bank classifications nor firm-bank relationship characteristics show any signs of F&F firms selecting themselves into negotiations with discretionary lenders, which strengthens the view that discretionary lending provides a valid instrumental variable for unsuccessful bank credit negotiations.

	(1) Credit (%)	(2) Meeting	
F & F	-0.281 (3.22)	-0.004 (0.03)	
N	1,061	1,097	

Table 2.8: F&F firms, credit negotiations, and bank-firm relationships

Notes: The table shows results for OLS estimations in a sample of all credit negotiations of all firms since 2008; standard errors are clustered at the firm-level and reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

2.6 Robustness Checks

2.6.1 Controlling for Sample Selection Bias

So far, all estimations have been conducted in the sample of firms that have negotiated a line of credit or a loan, respectively, which raises concerns about sample selection bias. This is addressed by applying a two-step estimation procedure suggested by Heckman (1979). From estimations of a probit model for firms' decisions to enter bank credit negotiations, the inverse Mills ratio can be calculated for every observation. The ratio is then included as a control variable in the OLS and IV estimations analogue to Section 2.4. In a comparable setup, Brown, Ongena, Popov, and Yeşin (2011) apply this procedure to control for selection effects when analysing the determinants of credit demand and credit access of firms in Eastern Europe using data from the Business Environment and Enterprise Survey (BEEPS).

In this empirical analysis, however, the second step of such a selection model contains an IV estimation. To the best of my knowledge, there is no standard procedure to estimate the correct standard errors in such a setup. Unfortunately, bootstrapped standard errors are not available due to the small number of observations in the data set. Therefore, the following results can only be considered as a "back-of-the-envelope" analysis to control for sample selection bias. The selection process is modelled as

$$Pr(Negotiation_i) = \Phi(\beta_0 + \beta_1 X_i + \beta_2 Competition_i)$$
(2.3)

where $\Phi(\cdot)$ is the cumulative distribution function of a normal distribution, X_i is the set of control variables and industry dummy variables, and *Competition_i* is a set of dummy variables measuring the level of competition a firm faces.¹⁴ The latter serve as exclusion restrictions in order to ensure identification of the two-step estimator. Hainz and Nabokin (2013) also use competition as an exclusion restriction when estimating how firms select themselves into having credit demand. They argue that "firms may invest more often in order to improve their position relative to other competitors" when facing high levels

¹⁴The level of competition on a scale from 1 to 11 is reported by firms in the Ifo "Financing of the German Economy" survey.

of competition and that this increases their probability of having credit demand. They further claim that the exogeneity with respect to the success of bank credit negotiations is ensured because banks assess the level of competition at the industry-level only. As industry dummy variables are included in all estimations in this study, any effects of competition on the outcome of credit negotiations should be ruled out.

The argument of Hainz and Nabokin (2013) is directly applicable to the estimation of a model for firms' selection into negotiations of bank loans, which are used to finance investments. It is also applicable to negotiations of lines of credit because some firms use them to finance investments. When facing high levels of competition, firms may also increase their working capital to improve their competitiveness.

From the probit estimations, the inverse Mills ratio is calculated and used as a control variable in the OLS estimations as described in Section 2.4.1. The results in Table B.4 in the Appendix confirm previous OLS estimation results from Section 2.4.2: Unsuccessful negotiations of a bank loan are associated with a higher probability of using F&F finance. This is primarily driven by the significant effect of unsuccessful loan negotiations on the use of "F&F Private". Even after controlling for selection bias, estimations do not show any signs of F&F finance being used as a substitute for unsuccessfully negotiated lines of credit.

As another robustness check, the inverse Mills ratio from the selection model is included as a control variable into IV estimations in comparison to estimations in Section 2.4.3. The results in Table B.5 in the Appendix confirm that firms use F&F finance in response to unsuccessfully negotiated loans, but not as a substitute for lines of credit. Unsuccessfully negotiated loans increase the probability of both "F&F Business" and "F&F Private" being used by a firm.

In both the OLS and IV estimations, the results suggest that the estimation of the effect of unsuccessful bank credit negotiations on the use of F&F finance are not affected by sample selection. The estimated coefficients hardly change compared to the estimations in which selection bias is not accounted for. Furthermore, the estimated coefficients of the inverse Mills ratio are statistically insignificant in all estimations.

2.6.2 The Role of Discouraged Borrowers

The extent to which F&F finance is used as a substitute for unsuccessfully negotiated bank credit could underestimate the true substitution effect in response to credit constraints if firms turn to F&F finance before even entering bank credit negotiations. They could do so, for example, because they expect a rejection of their credit application. Such firms are generally referred to as *discouraged* borrowers.

Whether discouraged borrowers provide further evidence in support of the result that credit constraints drive firms into informal finance can be tested with data from the Ifo "Financing of the German Economy" survey. If firms did not negotiate about bank credit, they could report in the survey whether they did not do so because they expected negotiations with banks to be unsuccessful.

OLS estimations are used to test the significance of the effect of a firm being discouraged on the use of F&F finance. The results in Table B.6 in the Appendix suggest that no further substitution behaviour into informal finance is prevalent because of expected unsuccessful bank credit negotiations. IV estimations that would allow a causal interpretation are not available here because the data does not contain any suitable instrumental variables for a firm being discouraged.

2.7 Conclusion

Based on a novel data set from the Ifo "Financing of the German Economy" survey, this chapter shows that F&F finance is surprisingly widespread among German firms. Considering that Germany is a highly developed country and that the sample does not focus on start-up businesses, this deserves special attention. In particular, it is important to analyse whether bank credit constraints are a cause of the widespread use of this particular kind of informal finance.

Descriptive statistics show that firms that unsuccessfully negotiate about bank credit are significantly more likely to use F&F finance. OLS estimations are used to control for a broad set of firm characteristics that previous literature found to determine a firm's

capital structure, its decision to use informal finance, and its risk of facing bank credit constraints. The results suggest that F&F finance is significantly more likely to be used if a firm has unsuccessfully negotiated a bank loan. Such an effect is not found for unsuccessful negotiations of lines of credit.

Since OLS estimations are affected by endogeneity stemming from the fact that F&F finance may have a signalling effect on banks' decisions to grant or reject credit, an IV approach is applied to estimate the causal effect of unsuccessful bank credit negotiations. Thereby, the dummy variable indicating that a bank follows discretionary lending provides an instrumental variable for unsuccessful bank credit negotiations.

IV estimations show that unsuccessfully negotiated bank loans drive the use of F&F finance, which highlights its importance in financing specific investment projects. For working capital finance, however, F&F finance seems less important as there is no significant effect of unsuccessfully negotiated lines of credit on the use of F&F finance. The results show that the substitution effect exists for both "F&F Business" and "F&F Private". The comparison of OLS and IV estimators further indicate that "F&F Business" serves as a positive signal of a firm's creditworthiness in bank credit negotiations while evidence of the signalling effect of "F&F Private" is mixed. Robustness checks show that these findings are unaffected by accounting for sample selection bias. Furthermore, there is no evidence that discouraged borrowers use F&F finance in response to expectations of unsuccessful bank credit negotiations.

Since turning to informal sources of finance is associated with higher borrowing costs (Djankov, Lieberman, Mukherjee, and Nenova, 2003) and welfare gains from financial intermediation no longer materialise, attention should be paid to the extent to which firms use informal finance in response to credit constraints. This study provides evidence that this phenomenon is not only occurring in developing countries or among start-up businesses, but that it is highly relevant even among established firms in a highly developed economy.

Chapter 3

Are Real Effects of Credit Supply Overestimated? Bias from Firms' Current Situation and Future Expectations^{*}

3.1 Introduction

Designing policy measures in response to credit constraints at the firm-level requires an understanding of whether they are caused by credit supply-side factors (e.g., a bank liquidity shock) or firm-side factors (e.g., a shock to the firms' creditworthiness). During the financial crisis of 2007-09, for example, banks in many countries faced a severe liquidity shock and reduced their lending to non-financial firms. Lacking access to credit, firms postponed investment and reduced their business activity. Therefore, credit supply-side factors caused a slowdown in real economic activity.¹ In other countries, however, banks weathered the financial crisis quite well, yet non-financial firms nevertheless experienced credit constraints as a reflection of their deteriorating creditworthiness when the world-

^{*}This chapter is based on joint work with Michael Kleemann.

¹This view has been supported by Brunnermeier (2009) and Shleifer and Vishny (2010), whereas Kahle and Stulz (2013) challenge this "bank credit channel" story.

wide economic slowdown hit demand for their products.² According to this narrative, credit constraints and the economic slowdown were not caused only by credit supply-side, but also by firm-side factors.

To achieve an unbiased estimation of real effects of credit supply-side factors, these must be disentangled from firm-side factors. For this purpose, existing empirical studies primarily use control variables from firms' balance sheets. These mirror past business, but fail to capture a firm's current business situation and future expectations, which induce estimation bias by affecting both credit supply and real economic activity. Market-based variables can be used to address this issue, but these are available for listed firms only and may not perfectly capture all contemporaneous and forward-looking firm-side factors.

We address the question of whether not controlling for firms' current business situation and future expectations, in addition to balance sheet data, leads to biased estimations of the real effects of credit supply. The analysis is based on data from the "EBDC Business Expectations Panel" for Germany between 2003 and 2011, which combines firms' balance sheets with survey-based assessments of their current business situation and future expectations.³ The data set provides a time-specific treatment variable indicating the experience of constrained credit-supply at the firm-level based on monthly panel data on firms' perceptions of bank lending supply. Applying ordinary least squares (OLS) and propensity score matching (PSM) estimation approaches, we estimate the treatment effect of constrained credit supply on monthly changes in a firm's production.

We find that the sole reliance on balance sheet data to rule out firm-side factors leads to an overestimation of credit supply-side effects on real economic activity. When controlling for balance sheet variables only, OLS and PSM estimations suggest that credit supply-side factors cause a significant increase in the probability of slowed down firm-level production during the post-treatment year. However, when including control variables for firms' current business situation and future expectations, the OLS estimator is significantly reduced. In PSM estimations, the estimated treatment effect even turns insignificant,

 $^{^2 {\}rm The}$ International Monetary Fund (2009) describes the economic crisis in Germany following this pattern.

³German data is particularly suited to analyse the real effects of credit constraints because Germany is a bank-based economy where many firms do not have access to public market finance as a substitute for bank credit and may therefore be particularly affected by credit constraints.

which raises doubts about the importance of credit supply-side shocks in Germany during the financial crisis. These findings are further supported by estimations using annual firmlevel employment growth as an alternative outcome variable. From our results, we draw the conclusion that the empirical literature analysing the real effects of credit supply should develop sufficient approaches to rule out firm heterogeneity that is not captured by balance sheets.

This analysis contributes to a strand of research on the estimation of the real effects of credit supply⁴ at the firm-level.⁵ In particular, we focus on the question of how to rule out firm-side factors in such estimations.⁶ Therefore, the control variables used in previous empirical studies are of particular interest here.

First, several studies use a combination of balance sheet and market data to rule out firm-side factors when estimating the real effects of credit supply-side shocks (e.g., on firms' investments).⁷ They all control for different sets of balance sheet variables such as firm size, cash flow, cash holdings, leverage, and profitability. Since balance sheets are backward-looking only, all these studies additionally control for at least one market-based variable (e.g., market-to-book value), assuming that market prices contain all relevant contemporaneous and forward-looking information about a firm.⁸

Market data to control for contemporaneous and forward-looking firm-side factors, however, is only available when firms are publicly traded. Using a sample of both public and non-public firms, Chodorow-Reich (2014) therefore does not use market data, but

⁴The link between credit supply and real economic activity was first shown by Bernanke (1983). Theoretical models including such a bank lending channel were developed by Bernanke and Blinder (1988), Bernanke and Gertler (1989), Kashyap and Stein (1994), Bernanke and Gertler (1995), Bernanke, Gertler, and Gilchrist (1996), Holmstrom and Tirole (1997), Kiyotaki and Moore (1997), and Gertler and Kiyotaki (2010).

⁵In addition to firm-level studies, Kroszner, Laeven, and Klingebiel (2007), Dell'Ariccia, Detragiache, and Rajan (2008), and Duygan-Bump, Levkov, and Montoriol-Garriga (2011) analyse the real effects of credit supply at the sector-level following Rajan and Zingales (1998) by using external finance dependence of a sector as a measure for financial constraints. Peek and Rosengren (2000) estimate the impact of the Japanese Banking crisis on the U.S. economy at the macroeconomic level.

 $^{^{6}}$ Therefore, the paper is not comparable to the analysis by Farre-Mensa and Ljungqvist (2013), who deal with the question of how to measure credit constraints.

⁷Gan (2007), Duchin, Ozbas, and Sensoy (2010), Almeida, Campello, Laranjeira, and Weisbenner (2011), Chava and Purnanandam (2011), and Lin and Paravisini (2012) do so by using financial crises as credit supply-side shocks, whereas Amiti and Weinstein (2013) identify bank shocks directly.

⁸Market data is also used to estimate the impact of bank health on firms' market values, for example by Yamori and Murakami (1999), Kang and Stulz (2000), Bae, Kang, and Lim (2002), and Ongena, Smith, and Michalsen (2003).

controls for firms' borrowing patterns, size, age, access to public bond markets, and a set of variables from the Dealscan database when estimating the effect of bank health on U.S. firm-level employment. In a robustness test, he also applies a within-firm approach as used by Khwaja and Mian (2008), which does not require controlling for a certain set of firm characteristics.

To the best of our knowledge, the study by Campello, Graham, and Harvey (2010) is the only one that includes control variables explicitly measuring firms' current business situation and future expectations in the estimation of real effects of credit supply. Similar to our analysis, they use survey data to show that self-reported financial constraints adversely affect several firm-level outcome variables (e.g., planned employment cuts). At first, they only control for firm size, ownership, industry, and rating category to rule out firm heterogeneity. Their estimated real effects of financial constraints turn out slightly smaller, but still significant, when they account for contemporaneous and forward-looking firm-side factors. They do so by including control variables for profitability, growth prospects, and dividend payer status. The last variable, however, is also a market-based variable that is only available for publicly traded firms.

We contribute to the literature on real effects of credit supply by controlling for a large set of explicit survey-based measures of firms' current business situation and future expectations without the need for market data. We test whether the sole reliance on balance sheet data leads to an overestimation of the real effects of credit supply. Thereby, we amend the approach of Campello, Graham, and Harvey (2010) by using panel data instead of cross-sectional data. Furthermore, the use of monthly survey data allows us to identify a direct and time-specific treatment variable for constrained credit supply and estimate its effect on monthly changes in firm-level production, in contrast to analyses based on annual data and more indirect measures of financial constraints. Finally, we estimate credit supply-side effects on annual employment growth, which is often not possible due to a lack of firm-level employment data.⁹

⁹A notable exception is the study by Chodorow-Reich (2014), who uses firm-level employment data. Campello, Graham, and Harvey (2010) estimate the effects of financial constraints on planned employment cuts without observing whether these plans are implemented. Duygan-Bump, Levkov, and Montoriol-Garriga (2011) estimate employment effects of financing constraints at the sector-level by using workers' employment status from the U.S. Current Population Survey (CPS).

Our results are also of interest for the estimation of credit supply-side effects on loan outcomes (without estimating real effects), in which firm-side factors must be held constant as well. For example, in their analysis of bank-side effects on loan outcomes using Spanish data, Jimènez, Ongena, Peydro, and Saurina (2012) control for firm size, leverage, and liquidity in addition to firm age and credit history. Analysing bank-side effects on credit rejections in Eastern Europe, Popov and Udell (2012) control for firm size, age, ownership, export status, and external auditing. Santos (2011) also includes market-based firm characteristics as control variables in his analysis of loan pricing during the recent subprime crisis. Explicit measures of firms' current situation and future expectations could amend analyses along this line of research.

The chapter is structured as follows. Section 3.2 derives the testable hypothesis. Section 3.3 describes the data set, the treatment definition, and provides descriptive statistics for all outcome and control variables. Section 3.4.1 explains the empirical strategy. Results for OLS and PSM estimation are presented in Sections 3.4.2 and 3.4.3. Section 3.5 contains several robustness checks. Section 3.6 discusses the external validity of the results and Section 3.7 provides concluding remarks.

3.2 Hypothesis

If the treatment of constrained credit supply was randomly assigned to firms, the observed difference in real economic activity between restricted and unrestricted firms could be interpreted as being caused by credit supply-side factors. Theoretical literature provides two explanations for why credit constraints could be randomly assigned to firms. According to Stiglitz and Weiss (1981), one symptom of credit rationing is the possibility that one firm is granted bank credit while the credit application of an identical other firm is rejected. Therefore, credit constraints would be considered to be randomly assigned. Furthermore, the model by Sharpe (1990) explains differences in credit supply between otherwise similar firms by heterogeneity in firms' bank relationships. Empirical evidence lends support to the argument that the health of relationship banks has an impact on firms (e.g., Almeida, Campello, Laranjeira, and Weisbenner (2011), Santos (2011), and

Chodorow-Reich (2014)). This is further supported by studies showing that banks with different characteristics, such as size and capitalisation, are differently inclined to transmit monetary policy changes to the real sector¹⁰ and that they differed in their lending behaviour during the financial crisis of 2007-09.¹¹

The empirical identification of the real effects of credit supply, however, is complicated by the fact that a bank's credit granting decision is typically not random, but depends on firm-side factors determining a firm's creditworthiness. If, for example, a firm experiences a product demand shock and its creditworthiness deteriorates, it is more likely to experience constrained credit supply. It is also more likely to reduce its production, but it does so in response to the product demand shock, not due to credit supply-side factors. The observed differences between restricted and unrestricted firms would overstate the real effects of credit supply-side factors. Alternatively, the observed difference could understate credit supply-side effects if firms with strong growth potentials share certain features that make them prone to face constrained credit supply. This could, for example, be the case for start-up businesses that are young, small, and risky.

Therefore, controlling for firm heterogeneity is crucial to avoid biased estimators of credit supply-side effects on real economic activity. For this purpose, control variables from balance sheets provide accurate measures of heterogeneity in firms' financial conditions. Balance sheets, however, mirror only past business and fail to capture firms' current business situation and future expectations. The latter predict a firm's future business activity and when deciding about granting credit, banks consider them (e.g., by looking at order books, interim financial statements, or business plans). Consequently, such contemporaneous and forward-looking firm characteristics, which is not captured by balance sheets, could induce estimation bias.

Hypothesis: Not explicitly controlling for firms' current business situation and future expectations, in addition to balance sheet variables, leads to biased estimations of the effects of credit supply-side factors on real economic activity.

¹⁰See, for example, Kashyap and Stein (2000), Kishan and Opiela (2000), Gambacorta (2005), Kishan and Opiela (2006), and Jimènez, Ongena, Peydro, and Saurina (2012).

¹¹See, for example, Albertazzi and Marchetti (2010), Ivashina and Scharfstein (2010), and Puri, Rocholl, and Steffen (2011a).
3.3 Data

3.3.1 The Data Set

For our analysis, we use data from the German "EBDC Business Expectations Panel". The data set links firms' balance sheets from the Bureau van Dyk (BvD) Amadeus database¹² and the Hoppenstedt database¹³ to panel data from the Ifo Business Survey.¹⁴ The latter is a monthly survey asking 3,600 plants from the German manufacturing sector for appraisals of their current business situation and expectations for their future business. It provides the basis for the Ifo Business Climate Index, an indicator of economic activity.

For the calculation of the index, the Ifo Institute continuously ensures that the panel of firms included in the survey is representative of the German manufacturing sector. A more detailed description of the data set is provided by Becker and Wohlrabe (2008).¹⁵ All variables used in this analysis are described in Table 3.1.

The Ifo Business Survey data is well-suited to test our hypothesis for three reasons. First, firms in the survey report their perception of bank lending supply, from which we can derive a month-specific treatment variable indicating constrained credit supply at the firm-level. Second, firms report recent changes in production on a monthly basis, which provides a precise measure of changes in post-treatment production. Third, and most importantly, the survey data contains a broad set of appraisals of firms' current business situation and future expectations. In combination with balance sheet variables, this provides a unique opportunity to test how controlling for contemporaneous and forwardlooking firm characteristics affects the estimation of the real effects of credit supply.

¹²The BvD Amadeus database contains balance sheet data and other firm-specific information about European firms, including about 1 million mainly non-listed German firms. Its primary source for Germany is the Creditreform, a German rating agency.

¹³Hoppenstedt is a leading provider of balance sheet data for German firms. The public press and commercial registries are among its main data sources. It has almost full coverage of publicly available financial statements in Germany.

¹⁴When linking annual balance sheet and monthly survey data, which is done based on the name and postal address of the firms, the following issue arises. The fiscal years of some firms in the sample do not coincide with calendar years. We use the monthly frequency of the data and put the most recently published balance sheet into every monthly observation. Thereby, we assume that a firm's balance sheet is made available in the credit application process immediately at the end of the fiscal year.

¹⁵Kipar (2011) uses data from the Ifo Business Survey to test the impact of credit constraints on firm-level innovation activity.

Variable	Description	Freq.
Restricted	Change in perception of bank lending from "accommo- dating" or "normal" to "restrictive"	Monthly
Variables of int	erest	M + l- l
Slowdown Slowdown_avg	Fraction of last 12 months in which production de-	Monthly
$\Delta Empl$	Vear-on-vear employment growth rate	Annual
Empl decrease	Negative $\Delta Empl$	Annual
		1111110001
Firm size		
Empl	Number of employees at company-level	Annual
Assets	Total assets	Annual
Balance sheet c	lata	A
	Equity / Assets	Annual
Cash	Cash holdings / Assets	Annual
Long-term debt	Long-term debt / Assets	Annual
Short-term debt	Short-term debt / Assets	Annual
Cash flow	Cash flow / Assets	Annual
ROA	Operating profit / Assets	Annual
Interest coverage	Operating profit / Interest expenses	Annual
Current busine	ss situation	
State	Appraisal: Current business situation	Monthly
Diale	(good / satisfactory / bad)	Wollding
Ordono	(good / satisfactory / bad)	Monthly
Oruers	(high / manual)	Monuny
	(mgn / enougn / too small)	M
Production	Production during last month	Monthly
01	(increased / unchanged / decreased)	
Short-time	Firm currently uses short-time work	Quarterly
Export	Firm is exporting	Quarterly
Future expecta	tions	
State exp	Expected business over next 6 months	Monthly
	(improvement / no change / worsening)	<i>j</i>
Empl ern	Expected employment over next 3 months	Monthly
Empi Cap	(increase / no change / decrease)	wionuny
Headcount	Employees for demand over novt 12 months	Quartorly
11600000000	(too few / enough / too many)	Quarterry
Short-time exp	Expecting to use short-time work during next 3 months	Quarterly
		- (a car o cor r y

Table 3.1: Variable descriptions

3.3.2 The Treatment of Constrained Credit Supply

Based on the panel structure of the data, we derive a variable indicating that a firm receives the treatment of constrained credit supply in a particular month. Since 2003, firms in the Ifo Business Survey are asked how they perceive "banks' willingness to supply credit". Possible appraisal categories are "restrictive", "normal", and "accommodating". This enables us to directly measure a change in credit supply to a firm. We define a firm as *restricted* or *treated* in month t (*Restricted*_{i,t} = 1) if it reports "restrictive" bank lending in month t after reporting "normal" or "accommodating" bank lending in the previous twelve months. A firm is defined as *unrestricted* or *untreated* in t (*Restricted*_{i,t} = 0) if it reports "normal" or "accommodating" bank lending in twelve subsequent months and does not switch to reporting "restrictive" bank lending in month t. This treatment definition allows the estimation of a treatment effect without bias from possible previous treatments. However, it comes at the cost of using only a fraction of the firms in the panel data set.

Estimating the treatment effect on a monthly basis requires an assumption on the exact treatment month for the time from 2003 to September 2008 in which the survey question on credit supply is asked only twice a year, namely in March and August. If a firm switches to reporting "restrictive" bank lending between two surveys in this time period, we assume that the treatment of constrained credit supply occurs in the first month after a firm has reported "normal" or "accommodating" the last time. This assumption is further illustrated and discussed in Appendix A.

After conditioning on the availability of all control variables in the pre-treatment month t-1 (see Sections 3.3.4 and 3.3.5), the sample consists of 333 firm-month observations in the treatment group and 5,061 untreated firm-month observations in the control group.¹⁶ Figure 1 in Appendix C shows the distribution of treated observations over time. As expected, the number of treated firms increases sharply in the wake of the financial crisis.

¹⁶The number of observations used in the following analysis is somewhat lower because of missing data in the post-treatment outcome variables.

Our approach to measuring credit supply is similar to the one used by Campello, Graham, and Harvey (2010), who also use survey data to identify financial constraints at the firmlevel. Such a perception-based approach has the caveat that the definition of what firms consider to be restrictive bank lending may differ over time or across industries. In the OLS estimations, this problem is dealt with by the inclusion of time and industry dummy variables. PSM estimation allows an even more stringent solution, namely matching firms exactly on quarter-industry cells.

3.3.3 Measuring Firm-Level Real Economic Activity

The Ifo Business Survey provides a precise measure of firm-level real economic activity. On a monthly basis, firms report whether their production has "increased", "not changed", or "decreased" during the last month compared to the month before. Answers to this question in every month from t+1 to t+12 measure changes in firm-level production after the treatment of constrained credit supply is assigned in month t.

The main dependent variable in the following analysis is the variable $Slowdown_{i,t+12}$, which indicates that a firm reports a decrease in production at least once during the twelve post-treatment months. We focus our analysis on this negative outcome because we are primarily concerned about restraining effects of credit supply. As a robustness check, we also run estimations using the outcome variable $Slowdown_avg_{i,t+12}$, which measures the fraction of the twelve post-treatment months in which a firm reports a decreasing production.

Furthermore, we estimate the treatment effect of constrained credit supply on a firm's annual employment growth rate $\Delta Empl_{i,t+12}$, which is calculated as the symmetric growth rate

$$\Delta Empl_{i,t+12} = \frac{Empl_{i,t+12} - Empl_{i,t}}{0.5(Empl_{i,t+12} + Empl_{i,t})}$$
(3.1)

as suggested by Haltiwanger, Jarmin, and Miranda (2013) and applied by Chodorow-Reich (2014). To avoid estimations being affected by extreme values due to survey re-

sponse behaviour, one percent of the observations are cut off from both sides of the distribution of $\Delta Empl_{i,t+12}$.

A caveat in the employment data stems from the fact that firms report their number of employees in the Ifo Business Survey every year in November. As the treatment variable is defined on a monthly basis, this raises difficulties in the distinction between post-treatment and pre-treatment changes in employment. Appendix B contains a discussion of this issue and presents the conservative approach to deal with it in this study. Nevertheless, the exact level of the employment effects of credit supply estimated should be interpreted with caution. Therefore, we use the dummy variable $Empl_decrease_{i,t+12}$ indicating that a firm has a negative post-treatment employment growth rate as an alternative dependent variable.

	R	$estricted_{i,t}$	= 1	$Restricted_{i,t} = 0$				
t+12	Ν	\bar{X}_R	σ_R	Ν	\bar{X}_U	σ_U	p > t	
Production								
Slowdown	316	69.9%	0.46	4877	52.5%	0.50	0.000^{***}	
$Slowdown_avg$	316	23.5%	0.24	4877	13.6%	0.19	0.000***	
Employment								
$\Delta Empl$	302	-1.5%	0.12	4739	1.0%	0.10	0.000^{***}	
Empl decrease	314	46.8%	0.50	4827	31.8%	0.47	0.000^{***}	

Table 3.2: Firms' post-treatment business activity

Notes: The table shows the descriptive statistics of post-treatment outcome variables separately for treated and untreated firms; treatment status $Restricted_{i,t}$ is defined as described in Section 3.3.2; p-values are reported for a two-group mean comparison t-test with H_0 : $\bar{X}_R = \bar{X}_U$; $\Delta Empl$ is cut by one percent from both sides of the distribution to deal with extreme values; these observations are kept in $Empl_decrease$; the samples contain only observations for which all pre-treatment control variables listed in Table 3.3 and Table 3.4 are available; * p < 0.1, ** p < 0.05, *** p < 0.01.

According to Table 3.2, firms that experience the treatment of constrained credit supply in t show significantly lower post-treatment real economic activity than untreated firms in all four outcome variables. They are significantly more likely to report a decreasing production at least once during the post-treatment year. On average, restricted firms face decreasing production in almost a quarter of the twelve post-treatment months while this figure is 10 percentage points lower for unrestricted firms. Employment growth rates show that restricted firms are shrinking by 1.5 percent on average in the post-treatment year while unrestricted firms grow by 1.0 percent. Finally, the probability of a decreasing employment is 15 percentage points higher for a restricted firm.

3.3.4 Heterogeneity in Firms' Balance Sheets

Table 3.3 illustrates how restricted and unrestricted firms differ in pre-treatment size and balance sheet variables $B_{i,t-1}$. These variables are often used in the literature to control for firm-side factors when estimating the real effects of credit supply. Firm size is widely used as a predictor of financial constraints because large firms tend to be older and more transparent, which may facilitate access to credit. In Table 3.3, however, restricted and unrestricted firms do not differ significantly in log(Empl) or log(Assets).

	$Restricted_{i,t} = 1$			$Restricted_{i,t} = 0$			
		(N=333))	(N=5,061	.)	
	_			_			
t-1	\overline{X}_R	X_R^{med}	σ_R	\overline{X}_U	X_U^{med}	σ_U	p > t
Firm size							
log(Empl)	5.4	5.3	1.3	5.5	5.4	1.1	0.25
log(Assets)	17.1	16.9	1.7	17.2	17.0	1.6	0.38
Balance sheet data							
Equity	33.2%	31.1%	25.4%	39.2%	38.3%	21.9%	0.000***
Cash	10.0%	4.3%	12.8%	11.3%	5.7%	13.8%	0.08^{*}
Long-term debt	15.4%	7.7%	19.7%	13.4%	6.4%	17.5%	0.05^{*}
Short-term debt	32.7%	30.2%	24.1%	30.7%	25.7%	40.6%	0.36
Cash flow	7.9%	7.8%	11.9%	10.5%	9.6%	12.2%	0.000***
ROA	-8.1%	3.0%	38.6%	-8.3%	6.1%	47.7%	0.93
Interest coverage	17.4	1.3	42.3	22.9	4.4	46.5	0.03**

Table 3.3: Firms' balance sheets in t-1

Notes: The table shows descriptive statistics of pre-treatment variables $B_{i,t-1}$ separately for treated and untreated firms; treatment status $Restricted_{i,t}$ is defined as described in Section 3.3.2; p-values are reported for a two-group mean comparison t-test with H_0 : $\bar{X}_R = \bar{X}_U$; Interest coverage is set to zero if values are negative and is winsorized at the 95 percentile; the samples contain only observations for which all pre-treatment control variables listed here and in Table 3.4 are available; * p < 0.1, ** p < 0.05, *** p < 0.01.

The other variables in Table 3.3 are in line with existing empirical literature showing that restricted firms are in a worse financial condition than unrestricted ones. They have significantly lower equity ratios, fewer cash holdings, more long-term debt, lower cash flows, and lower interest coverage ratios.¹⁷ Restricted firms also hold more short-term debt and are characterised by a lower operating profitability (ROA). Therefore, it is important to rule out heterogeneity in firms' balance sheets in the estimation of real effects of credit supply.

3.3.5 Heterogeneity in Firms' Current Business Situation and Future Expectations

Beyond the information contained in balance sheets, Table 3.4 shows that restricted and unrestricted firms also differ significantly in their pre-treatment current business situation $C_{i,t-1}$ and future expectations $F_{i,t-1}$. All variables stem from the firms' appraisals and assessments reported in the Ifo Business Survey.

First, firms are asked directly how they appraise their current business situation. Restricted firms are significantly less likely to report a "good" situation (*State* (+)), but significantly more likely to report a "bad" one (*State*(-)).¹⁸ Second, firms provide an assessment of their current stock of orders, which restricted firms are significantly more likely to report as being "too small" (*Orders* (-)). Third, restricted firms are significantly more likely to report a decreasing production in the pre-treatment month (*Production* (-)), which indicates a pre-treatment trend in the outcome variable for real economic activity. Finally, variables indicating that a firm is currently using short-time work and whether it is exporting do not show any significant differences.¹⁹

 $^{^{17}}$ Interest coverage is set to zero if it takes on negative values and winsorised at the 95 percentile.

¹⁸Throughout this paper, only good and bad appraisals or assessments are reported for variables from the Ifo Business Survey. Neutral categories constitute the baseline category in all estimations.

¹⁹Short-time work is a labour market instrument widely used by German firms to adjust their capacities to business cycle or seasonal demand fluctuations. The practice was particularly widespread during the financial crisis. Export status is a crucial control variable because foreign demand in Germany was severely affected by the financial crisis, which could affect both access to credit and real economic activity of a firm.

Forward-looking firm characteristics in the data show further heterogeneity between restricted and unrestricted firms that might not be captured by backward-looking balance sheet variables. As shown in Table 3.4, restricted firms are significantly less likely to expect business to "improve" over the next six months (*State exp* (+)), but significantly more likely to expect business to "worsen" (*State exp* (-)).

	$Restricted_{i,t} = 1$		$Restricted_{i,t} = 0$				
		(N=333)		(N=5,061)			
t-1	\bar{X}_R	X_R^{med}	σ_R	\bar{X}_U	X_U^{med}	σ_U	p > t
Current business situ	ation						
State (+)	21.6%	0	41.2%	25.9%	0	43.8%	0.08^{*}
State (-)	27.9%	0	44.9%	20.8%	0	40.6%	0.002***
Orders (+)	10.5%	0	30.7%	13.0%	0	33.6%	0.19
Orders (-)	39.3%	0	48.9%	32.9%	0	47.0%	0.02**
Production (+)	12.3%	0	32.9%	14.4%	0	35.1%	0.29
Production (-)	23.7%	0	42.6%	19.3%	0	39.5%	0.05^{*}
Short-time	13.2%	0	33.9%	14.5%	0	35.2%	0.53
Export	87.4%	1	33.2%	88.6%	1	31.8%	0.50
Future expectations							
State $exp(+)$	15.3%	0	36.1%	19.0%	0	39.2%	0.09^{*}
State exp (-)	25.5%	0	43.7%	18.8%	0	39.1%	0.002***
Empl exp (+)	5.7%	0	23.2%	6.6%	0	24.9%	0.51
Empl exp (-)	23.1%	0	42.2%	15.8%	0	36.5%	0.000***
Headcount (+)	5.7%	0	23.2%	6.3%	0	24.3%	0.67
Headcount (-)	27.0%	0	44.5%	20.1%	0	40.1%	0.002***
Short-time exp	19.2%	0	39.5%	19.7%	0	39.8%	0.82

Table 3.4: Firms' current business situation and future expectations in t-1

Notes: The table shows descriptive statistics of pre-treatment variables $C_{i,t-1}$ and $F_{i,t-1}$ separately for treated and untreated firms; treatment status $Restricted_{i,t}$ is defined as described in Section 3.3.2; p-values are reported for a two-group mean comparison t-test with H_0 : $\bar{X}_R = \bar{X}_U$; the samples contain only observations for which all pre-treatment control variables listed here and in Table 3.3 are available; * p < 0.1, ** p < 0.05, *** p < 0.01. This picture is confirmed for the variable indicating that a firm expects its employment figures to decrease over the next three months (Empl exp (-)) and the variable indicating that a firm considers its workforce too large for the product demand over the next twelve months (*Headcount (-)*). There is no such difference in the expectation of short-time work (*Short-time exp*).

In sum, variables measuring firms' current business situation and future expectations are highly relevant in predicting constrained credit supply. As a firm's production and employment decisions also depend on its current business situation and future expectations, heterogeneity in these variables must be controlled for to ensure an unbiased estimation of the real effects of credit supply. As balance sheet variables are backward-looking only, additional control variables from survey data could provide a key contribution in this concern.

3.4 Methodology and Results

3.4.1 Methodology

For every firm *i* in the panel data set, we observe whether it receives a treatment of constrained credit supply in month t (*Restricted*_{*i*,*t*} = 1) and estimate the effect of this treatment on the likelihood of a decrease in production during the subsequent twelve months (*Slowdown*_{*i*,*t*+12} = 1).²⁰ In a first step, we estimate

$$E[Slowdown_{i,t+12}|B_{i,t-1}] = \beta_0^B + \beta_1^B Restricted_{i,t} + \beta_2^B B_{i,t-1} + \epsilon_i$$
(3.2)

where $B_{i,t-1}$ is a set of variables from a firm's most recent balance sheet in the pretreatment month t-1.

 $^{^{20}{\}rm We}$ illustrate the empirical strategy based on the notation of an OLS estimation. In Section 3.4.3, we also test the hypothesis using PSM estimation.

We then extend the estimation to

$$E[Slowdown_{i,t+12}|B_{i,t-1}, C_{i,t-1}, F_{i,t-1}] = \beta_0^{BCF} + \beta_1^{BCF} Restricted_{i,t} + \beta_2^{BCF} B_{i,t-1} \quad (3.3)$$
$$+ \beta_3^{BCF} C_{i,t-1} + \beta_4^{BCF} F_{i,t-1} + \zeta_i$$

by additionally controlling for a set of survey-based variables measuring firms' pretreatment current business situation $C_{i,t-1}$ and expectations for the future business $F_{i,t-1}$. Finally, we calculate the impact of these additional control variables on the estimated treatment effect as

$$Diff^{CF} = \beta_1^B - \beta_1^{BCF} \tag{3.4}$$

which measures the degree to which the estimated real effects of credit supply are biased when controlling for balance sheet variables only, without accounting for heterogeneity in firms' current business situation and future expectations.

3.4.2 Ordinary Least Squares Estimations

Pooled OLS estimation provides a first test of whether controlling for firms' current business situation and future expectations affects the estimation of the real effects of credit supply. In line with Equation (1) in Section 3.4.1, the dummy variable $Slowdown_{i,t+12}$ is regressed on the treatment status $Restricted_{i,t}$, a set of balance sheet variables $B_{i,t-1}$, and quarter and industry dummy variables based on the two-digit WZ 2008 industry classification. The results in Estimation (1) in Table 3.5 suggest that constrained credit supply increases the probability of decreasing production between t and t+12 by 8.65 percentage points.

When additionally controlling for firms' current business situation $C_{i,t-1}$ and future expectations $F_{i,t-1}$ in Estimation (2), this effect is reduced to 7.01 percentage points. Column (3) shows that the difference between the two estimated coefficients is highly statistically significant.²¹ The adjusted R^2 statistics further show that the inclusion of the control variables $C_{i,t-1}$ and $F_{i,t-1}$ increases the explanatory power of the model substantially.

Therefore, OLS estimation provides first support for the hypothesis that not controlling for contemporaneous and forward-looking firm-side factors leads to a significant overestimation of the impact of credit supply-side factors on real economic activity at the firm-level. The inclusion of survey-based measures for firms' current business situation and future expectations reduces the estimated effect by almost 20 percent.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	0.0865^{***} (0.027)	0.0701^{***} (0.026)	0.0164**
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
Month	Yes	Yes	
Industry	Yes	Yes	
Adj. R^2	0.1666	0.2200	
Ν	$5,\!193$	$5,\!193$	

Table 3.5: OLS estimations using $Slowdown_{i,t+12}$

Notes: The table provides results for OLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are sets of control variables as listed in Table 3.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated coefficients; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

²¹To test this significance, standard test procedures are not applicable because the two coefficients come from different estimations. Therefore, the significance level for the test procedure that Clogg, Petkova, and Haritou (1995) have developed for such a case is reported here.

3.4.3 Propensity Score Matching Estimations

OLS estimation has two potential disadvantages in the context of this analysis. First, it contains the risk of misspecifying the functional form of $E[Slowdown_{i,t+12}]$. Second, OLS estimation leads to a comparison of observations outside the common support if distributions of control variables vary between treated and untreated firms. This is most likely to be a problem for the balance sheet variables $B_{i,t-1}$, which show large standard errors that differ between treated and untreated firms according to Table 3.3. To address these issues, the effect of constrained credit supply on firm-level production can be estimated using a matching estimation approach, as done by Campello, Graham, and Harvey (2010) and Almeida, Campello, Laranjeira, and Weisbenner (2011).

The following analysis is based on PSM estimation following Rosenbaum and Rubin (1983) because the large number of control variables, including continuous balance sheet variables in $B_{i,t-1}$, inhibits the identification of matching firms that are identical with respect to all control variables. Comparing each restricted firm to unrestricted ones with a similar propensity score provides an estimated treatment effect that is close to the one derived from an experimental setting (Dehejia and Wahba, 1999) in which restrictive bank lending is randomly assigned. The identifying assumption still hinges on the choice of matching variables. In this concern, our study provides a major contribution to the existing literature by using a broad set of matching variables

Following the empirical strategy in Section 3.4.1, firms are first matched on balance sheet variables $B_{i,t-1}$ only. This requires the estimation of a logistic regression model

$$Pr(Restricted_{i,t} = 1|B_{i,t-1}) = \Phi(\alpha_0 + \alpha_1 B_{i,t-1} + \alpha_2 Industry_i > u_i)$$
(3.5)

where $\Phi(\cdot)$ denotes the cumulative distribution function of the logistic distribution and $Industry_i$ is a set of industry dummy variables.²² To allow for time-heterogeneous effects, estimations are run separately within every quarter.²³

²²To avoid the impact of extreme values on the estimation, the delta deviance influence statistic is extracted for every observation. Following Agresti and Finlay (2008), an observation is dropped if $\sqrt{|ddeviance|} > 3$ and the maximum likelihood estimation is re-run until no observations are dropped.

²³Estimations within quarter-industry cells are not possible due to a lack of observations. We therefore estimate the model separately for every quarter and include industry dummy variables.

From the estimated models, the propensity score, which is the conditional probability of being treated, is predicted for every firm-month observation. Every restricted firm is then linked to unrestricted firms from the same quarter-industry cell that have a similar propensity score based on ten nearest neighbours matching. Using exact matching on quarter-industry cells rules out macroeconomic and industry-specific effects, and deals with both the time- and industry-dependence of the perception-based treatment variable. Finally, the treatment effect is estimated using a Weighted Least Squares (WLS) estimation of the outcome variable $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$ with weights being drawn from the PSM procedure. In a second step, we re-run this matching procedure adding matching variables for firms' current business situations $C_{i,t-1}$ and future expectations $F_{i,t-1}$ in addition to $B_{i,t-1}$.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$5.10\%^{**}$ (0.0236)	2.59% (0.0297)	2.51%
Industry	Yes	Yes	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
p>t	0.03	0.38	
Upper bound	8.98%	7.47%	
Lower bound	1.22%	-2.30%	
Treated obs.	223	141	
Matching obs.	$1,\!351$	806	

Table 3.6: PSM estimation using $Slowdown_{i,t+12}$

Notes: The table provides results for WLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$; in Estimation (1), weights are derived from PSM based on pre-treatment firm size and balance sheet data; in Estimation (2), weights are derived from PSM based on pre-treatment firm size, balance sheet data, current business situation, and future expectations; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of significance of the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval of each estimator; standard errors are reported in parentheses; * p < 0.1, *** p < 0.05, *** p < 0.01.

When matching only on balance sheet variables, the results for Estimation (1) in Table 3.6 suggest that constrained credit supply increases the probability of a decreasing production in the post-treatment year significantly by 5.10 percentage points. Estimation (2), however, shows that this treatment effect is only 2.59 percentage points and statistically insignificant when also matching on firms' current business situation and future expectations.

This suggests that the sole reliance on balance sheet data leads to an overestimation of real effects of credit supply-side factors by almost 50 percent.²⁴ Unfortunately, the statistical significance of the difference between the two estimators cannot be tested because the test by Clogg, Petkova, and Haritou (1995) requires each observation to receive the same weight in both estimations, which is contrary to the core idea of a matching estimator.

It is particularly noteworthy that Estimation (2) shows no evidence for credit supplyside effects on real economic activity in Germany between 2003 and 2011 as suggested in Estimation (1). Therefore, PSM estimation strengthens the empirical support for the hypothesis that a lack of control variables to measure firms' current business situations and future expectations leads to an overestimation of real effects of credit supply.

3.5 Robustness Checks

3.5.1 Potential Caveats of the Survey Data

The survey questions raise three problems in the context of our analysis. First, respondents could be in a bad mood because of their firm's general situation and future expectations, and react by reporting restrictive bank lending (e.g., to blame banks for their situation). Therefore, the lower estimated effect of constrained credit supply could not be attributed to survey-based variables providing information that is not captured by balance sheets. Empirical evidence by Abberger, Birnbrich, and Seiler (2009) provides strong evidence against this critique. They use data from a meta-study among respon-

²⁴Table C.1 in Appendix C shows that matching on balance sheet variables only does not eliminate bias in the survey-based variables entirely. Matching on all variables for firms' current business situation and future expectations improves the balancing properties.

dents of the Ifo Business Survey to show that general mood and economy-wide factors are not important determinants of firms' general appraisals in the survey.

Second, when providing appraisals of their current business situation, firms may just have their financial situation and therefore their balance sheet in mind. If that was the case, the smaller estimated effects when including survey-based control variables would only be mechanical due to the fact that more control variables are used. For firms' current business situation, their revenues and turnover are the most important determinants according to Abberger, Birnbrich, and Seiler (2009). There is no evidence that firms think about their financial situation when filling out the survey.

Finally, forward-looking appraisals could contain expectations of constrained credit supply. In this case, the explanatory power of these variables would mean that firms' anticipation of the treatment turn out correct, but it could not be taken as evidence for the importance of ruling out forward-looking firm-side factors. For firms' future expectations, Abberger, Birnbrich, and Seiler (2009) show that costs and liquidity, which could encompass the consequences of constrained credit supply, are only the fourth and fifth most important determinants.

We test the robustness of our results with respect to the anticipation of constrained credit supply by re-running the estimations controlling only for those survey-based variables that explicitly refer to factual aspects of business. These are assessments of the current stock of orders, as well as the current status concerning short-time work and export. We also keep the variables *Headcount* (+) and *Headcount* (-) as they refer explicitly to employment figures relative to future demand and are therefore unaffected by firms' expectations of access to credit supply. The general appraisals *State*, *Production*, *State exp*, *Short-time exp*, and *Empl exp* are excluded because they could be affected by expectations of constrained credit supply.

Table C.2 in Appendix C shows that controlling for such a reduced set of variables for firms' current business situation and future expectations in OLS estimations leads to a significant reduction of the estimated credit supply-side effects by 1.5 percentage points. The difference is only slightly smaller than in our baseline estimation in Section 3.4.2, and

is still highly statistically significant.²⁵ The adjusted R^2 statistic increases substantially suggesting a better model fit in Estimation (2) even when using a reduced set of variables.

This is confirmed by PSM estimations with a reduced set of control variables. Matching on balance sheet variables only leads to an estimated treatment effect of 5.74 percentage points in Estimation (1) in Table C.3 in Appendix C. Controlling for the factual assessments of firms' current situation and future expectations without using general appraisals lowers the estimated effect to 3.16 percentage points in Estimation (2) and the estimator turns statistically insignificant.

Therefore, even if the power of general assessments of contemporaneous and forwardlooking firm-side factors are not to be believed, the omission of more factual variables still induces an overestimation of real effects of credit supply. For both the OLS and PSM estimation the suggested overestimation is only slightly smaller than in estimations based on all survey-based variables in Section 3.4.2 and 3.4.3.

3.5.2 An Alternative Measure for Firm-Level Production

So far, results were shown for the estimated effects of constrained credit supply on the probability that a firm reports a decreasing production at least once in the twelve post-treatment months ($Slowdown_{i,t+12}$). As a robustness check, we re-run the previous estimations using the fraction of the twelve post-treatment months in which firms report decreasing production ($Slowdown_avg_{i,t+12}$) as the dependent variable.

The results for OLS estimations in Table C.4 in Appendix C show that the omission of control variables for firms' current business situation and future expectations lowers the estimated treatment effect of $Restricted_{i,t}$ on $Slowdown_avg_{i,t+12}$ by about 12 percent. The difference is statistically significant and the adjusted R^2 statistic indicates a much better model fit when including contemporaneous and forward-looking control variables.

 $^{^{25}}$ The effect when controlling only for balance sheets variables differs from the one shown in Table 3.5. Although control variables do not change compared to the estimation in Table 3.5, conditioning on the availability of the smaller set of control variables in Estimation (2) affects the sample composition, and therefore alters the estimated effect.

When applying PSM estimation with $Slowdown_avg_{i,t+12}$ as the dependent variable, this is confirmed. Matching on balance sheet variables only, Estimation (1) in Table C.5 shows a highly significant credit supply-side effect on firm-level production of 3.78 percentage points. Adding matching variables for firms' current business situation and future expectations in Estimation (2) substantially lowers the effect by 1.01 percentage points. In contrast to estimations using $Slowdown_{i,t+12}$, however, the effect remains significant at the ten percent level. Even when using an alternative outcome variable for firm-level production, these estimation results underline the importance of controlling for firms' current business situation and future expectations in the estimation of real effects of credit supply.

3.5.3 Employment Effects

The previous estimated effects of constrained credit supply on firm-level production raise the question whether the omission of variables capturing contemporaneous and forwardlooking firm-side factors also affects the estimation of employment effects of constrained credit supply. Tables C.6 and C.7 in Appendix C show that controlling for firms' current business situation and future expectations lowers the estimated effect of constrained credit supply on a firm's annual employment growth rate. The difference is statistically significant in the OLS estimation. For PSM estimation, the effect turns insignificant and the two effects lie outside each other's 95 percent confidence intervals. As mentioned in Section 3.3.3, the exact level of the estimated employment effects should be interpreted with caution due to the problems in linking annual employment data to the monthly treatment variable (see Appendix B for a discussion).

Table C.8 in Appendix C therefore provides OLS estimations of credit supply-side effects on the probability that a firm experiences a negative employment growth rate in the post-treatment year ($Empl_decrease_{i,t+12}$). The inclusion of control variables for firms' current business situation and future expectations lowers the estimated effect significantly and the adjusted R^2 statistic increases substantially. PSM estimations in Table C.9 in Appendix C further show that the estimated treatment effect of constrained credit supply turns insignificant when including all control variables. The two estimators lie outside each others' 95 percent confidence intervals. In sum, these results show that heterogeneity in firms' current business situation and future expectations that is not captured by balance sheet data also affects the estimation of employment effects of constrained credit supply.

3.5.4 Sample Selection in PSM Estimations

The differences between PSM estimations with and without matching on firms' current business situation and future expectations could be caused by the mere fact that different samples of treated firms are used in the two estimations, which is obvious from the numbers of observations. The samples differ for three reasons. First, when matching on additional variables $C_{i,t-1}$ and $F_{i,t-1}$, the logistic regression may no longer converge in some quarters so that the propensity score is not estimated for the respective treated observations. Second, the inclusion of these binary variables may induce perfect predictability so that some variables and the respective treated observations are dropped. Finally, increasing the number of matching variables makes it more likely that observations are dropped because the common support condition is violated.

To test whether the previous results are actually driven by better matching quality, posttreatment outcome variables for the treated firms in both samples are compared in Table C.10 in Appendix C. For $Slowdown_{i,t+12}$, firms differ only slightly. In fact, firms in the second sample show an even larger probability of decreasing production, which strengthens the interpretation that solely matching on balance sheet data induces an overestimation of the real effects of credit supply. Two-group mean comparison tests indicate that the difference is statistically insignificant. The same holds for $Slowdown_avg_{i,t+12}$. For $Empl_decrease_{i,t+12}$ and $\Delta Empl_{i,t+12}$, parts of the reduction in the estimated treatment effect of constrained credit supply can be explained by differences between the treated firms, but the difference is statistically insignificant and there is still a substantial reduction remaining due to improved matching quality.

3.5.5 The Role of the Financial Crisis of 2007-09

The sample period from 2003 to 2011 covers the financial crisis, which might influence the extent to which omitting firms' current business situation and future expectations affect the estimation of the real effects of credit supply. Firms were operating under extreme conditions during the financial crisis. Therefore, firm balance sheet data from previous years turned less informative for banks. Under such circumstances, banks may base their lending decisions even more on contemporaneous and forward-looking firm characteristics.

As Tables C.11 and C.12 in Appendix C show, controlling for firms' current business situation and future expectations lowers the estimated treatment effect of constrained credit supply on firm-level production in the financial crisis subsample as of July 2007. For the pre-crisis sample, however, estimates are unreliable because only a few firms are treated in this period and the sample turns out too small for our estimation procedure. Therefore, the subsample analysis underlines the importance of contemporaneous and forward-looking information on firms in times of financial crisis when balance sheet data from previous years might turn less informative. Whether there is a difference compared to normal times, however, remains open for future research when more "non-crisis" data is available.

3.6 Discussion: No Real Effects of Credit Supply?

Most PSM estimations in this paper show no significant effect of credit supply-side factors on real economic activity. This contrasts what appears to be conventional wisdom;, for example about the financial crisis of 2007-09. The insignificant effects may be specific to this study because it is based on data from Germany where the economic crisis was less driven by bank-side factors.

According to the International Monetary Fund (2009), the economic crisis in the German economy, which depends strongly on exports, was induced by a sharp drop in foreign demand. Problems in the banking sector arose because firms were unable to repay their

debt and increasing uncertainty prevented banks from lending. This, however, is in line with credit constraints being driven by firm-side, and not by credit supply-side factors. Supporting this view, Rottmann and Wollmershaeuser (2013) show that capital ratios of German banks rose from 4% early 2008 to 4.5% by the end of 2009. They also argue that the establishment of the Financial Markets Stabilization Fund in October 2008 helped avoid liquidity constraints in the banking sector. Therefore, it is not surprising that previous PSM estimations did not provide evidence for real effects of credit supply in Germany between 2003 and 2011.

Concerning the external validity of these results, we wish to emphasise that this picture may be different for other time periods or other countries. However, the finding that a lack of controlling for contemporaneous and forward-looking firm-side factors leads to overestimated credit supply-side effects, is generally applicable.

3.7 Conclusion

To design appropriate policy measures encountering a lack of access to credit for firms, it is crucial to understand whether this is driven by credit supply-side or firm-side factors. If economic growth is slowed down by limited lending due to credit supply-side factors (e.g., due to banks facing a liquidity crunch), government intervention in the banking sector may be justifiable. If, however, growth rates are hampered by firm-side factors (e.g., a drop in demand for the firms' products), and credit volumes decrease in response to higher default risk of firms, government intervention, if such is deemed necessary, should not necessarily be directed at banks.

To identify credit supply-side effects on real economic activity, firm-side effects must be ruled out. Existing firm-level analyses do so primarily using data from firms' balance sheets. Although these provide an accurate picture of a firm's financial condition, they are backward-looking and do not contain information on firms' current business situation and future expectations. Therefore, analyses are often complemented with market data, which limits empirical studies to publicly traded firms.

This study analyses the question of whether controlling for survey-based measures of firms' current business situation and future expectations makes a difference in the estimation of the real effects of credit supply. To that end, we estimate the effect of constrained credit supply on firm-level production in Germany between 2003 and 2011. When including survey-based appraisals of firms' current business situation and future expectations, in addition to balance sheet variables, OLS estimators remain significant, but turn out significantly lower than without these additional control variables. When applying a PSM approach, the estimated treatment effect of credit supply on firm-level production is also substantially reduced and turns insignificant. The importance of ruling out contemporaneous and forward-looking firm-side factors is further shown for estimations of the effects of constrained credit supply on firm-level employment growth.

Our results indicate that ignoring contemporaneous and forward-looking information on the credit demand-side may lead to an overestimation of credit supply-side effects on real economic activity. Studies analysing the real effects of credit supply should take this into account. For firm-level analyses, micro-data from surveys are an option for circumventing overestimation if other data (e.g., market data or precise information on order books) are not available. For macro-level analyses, indices such as the Purchasing Mangers Index (PMI) could be used. Further research using data for publicly traded firms could also shed light on the question whether the direct measures for firms' current situation and future expectations rule out bias that is not captured by market data.

Appendices

Appendix to Chapter 1

Appendix: Sample Representativeness

As the analysis in Chapter 1 is based on survey data, response behaviour could prompt sample selection issues. Firms with impaired credit finance could be overrepresented in the sample since the topic of the survey was the firms' financing situation. One could argue that firms that did not need credit finance and those that did not experience any impairments of it were less likely to participate in the survey because they lack interest in the topic or they did not consider themselves to have anything to contribute.

To test whether certain firms are overrepresented in the sample, we compare firms responding to the Ifo "Financing of the German Economy" survey ("response" firms) to those who did not respond ("non-response" firms) with respect to information that the Ifo Investment Survey provides about both groups. Since both surveys are based on the same address database, there is a sufficient overlap between the data sets. That firms with impaired credit finance are also overrepresented in the sample of the Ifo Investment Survey is unlikely because the main topic of this survey is the development and the structure of firms' investment activities. The motivation to respond should therefore not be affected by the firm's credit financing during the financial crisis.

The first two columns of Table A.1 show that response firms are slightly larger than non-response firms in terms of employment and turnover in 2010, but the differences are statistically insignificant. Even stronger arguments in favour of sample representativeness can be found in the lower part of Table A.1. Every year, firms participating in the Ifo Investment Survey are asked to assess how the financing situation affects their investment in the current year. The answers range from 1 ("strong animation") to 5 ("strong slowdown"). If firms that see credit finance impaired by the financial crisis were to be more likely to respond to the Ifo "Financing of the German Economy" survey, we would expect the according answers in the Ifo Investment Survey to differ significantly between response and non-response firms. Table A.1, however, shows no difference in the influence of finance on investment between the two groups.

	Non-response	Response	n > t
		Response	p > v
Turnover (2010, in m Euro)	265.68	398.94	0.2811
Ν	1,118	591	
Employment (2010)	808.49	1050.32	0.3881
Ν	1,118	591	
Influence Finance 2007	2.96	2.93	0.4737
Ν	809	444	
Influence Finance 2008	3.05	3.03	0.5120
Ν	859	468	
Influence Finance 2009	3.38	3.41	0.4632
Ν	922	500	

Table A.1: Analysis of response behaviour based on data from the Ifo InvestmentSurvey

Note: The table shows the comparison of firms that participated in the Ifo "Financing of the German Economy" survey ("Response") and firms that did not return the questionnaire ("Non-response"); all variables used to compare the two groups are drawn from the Ifo Investment Survey; p-values are reported for a standard mean comparison t-test; employment is measured in heads; *Influence Finance* provides a firm's appraisal of the influence of the financing situation on investment in the current year with the following answer categories: 1 Strong animation, 2 Little animation, 3 No influence, 4 Little slowdown, 5 Strong slowdown.

Empirical Appendix

	Ν	Freq.	Perc.
Long duration	$1,\!610$	$1,\!374$	85.34%
Personal support	$1,\!610$	1,067	66.27%
Short distance	1,610	845	52.48%
Company knowledge	1,610	694	43.11%
Difficult times	1,610	591	36.71%
Important creditor	1,610	514	31.93%
Others	1,610	39	2.42%

Table A.2: Features of main bank relationships

Note: For the two most important bank relationships, firms reported whether or not the different features characterise the bank relationship and whether the bank is a main bank. Here, a data set of main bank relationships is constructed. The dummy variables equal one if the respective feature is reported for the respective main bank relationship.

	(1)	(2)	(3)	(4)	(5)	(6)
	Information	Reduction	Availability	Interest	Maturities	Collateral
Main bank (1)	-0.1543***	0.0017	-0.0371	-0.0990**	-0.0547**	-0.0789**
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
Main bank (2)	-0 1044**	0.0125	-0.0298	-0 1122***	-0.0455**	-0.0472
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
log(Fmployees)	0.0066	0.0020	0.0203	0.0272*	0.0187	0.0101
log(Employees)	(0.0000)	(0.0023)	(0.01)	(0.0212)	(0.01)	(0.0131)
1(A+-)	(0.02)	(0.01)	0.0011	(0.02)	(0.01)	0.0026
log(Assels)	0.0080	(0.0104)	-0.0011	-0.0100	(0.0132)	-0.0050
F :	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Equity	0.0025	-0.0341	-0.0402	-0.0422	-0.0004	-0.0373
T , 11.	(0.05)	(0.04)	(0.05)	(0.05)	(0.03)	(0.04)
Long-term debt	0.0330	0.0299	0.0963	0.0645	-0.0204	0.0222
	(0.06)	(0.05)	(0.07)	(0.06)	(0.03)	(0.06)
Cash	-0.3082***	-0.0612	-0.1880**	-0.1475*	-0.0287	-0.2384^{***}
	(0.10)	(0.07)	(0.09)	(0.09)	(0.05)	(0.08)
Return (3% to $<7\%$)	-0.0759**	-0.0599***	-0.0394	-0.0782^{***}	-0.0143	-0.0651**
	(0.03)	(0.02)	(0.03)	(0.03)	(0.01)	(0.03)
Return (7% to <10%)	-0.0416	-0.0846***	0.0085	-0.0615*	-0.0008	-0.0234
	(0.04)	(0.02)	(0.04)	(0.04)	(0.02)	(0.04)
Return (10% +)	-0.0638	-0.0514*	0.0157	-0.0660	-0.0094	-0.0673*
	(0.05)	(0.03)	(0.05)	(0.04)	(0.02)	(0.04)
Earlypau (0% to $< 25\%$)	-0.0114	0.0048	0.0037	0.0642	-0.0959	-0.0201
	(0.11)	(0.07)	(0, 09)	(0.10)	(0.08)	(0.10)
Earlypay (25% to $< 50\%$)	-0.0294	0.0353	0.0143	-0.0309	-0.0751	-0.0029
Earigpag (2070-00 <0070)	(0.11)	(0.07)	(0.09)	(0.09)	(0.08)	(0.10)
$Early = (50\% t_0 < 75\%)$	0.0342	0.0331	0.0448	0.0542	0.1312*	0.0067
Earlypag $(50\% t0 < 15\%)$	(0.0342)	(0.0551)	(0.00)	(0.10)	-0.1312	(0.10)
Farlunger (75% 1)	0.1058	0.0462	0.0762	0.0657	0.1194	0.10)
Earrypay (75% +)	-0.1056	-0.0405	-0.0705	-0.0037	-0.1124	-0.0770
I(A)	(0.10)	(0.00)	(0.08)	(0.09)	(0.08)	(0.09)
log(Age)	0.0037	-0.0070	-0.0234	0.0178	0.0016	-0.0123
T	(0.02)	(0.01)	(0.01)	(0.02)	(0.01)	(0.02)
Incorporated	0.0317	0.0188	0.0008	0.0657**	0.0017	0.0401
	(0.03)	(0.02)	(0.03)	(0.03)	(0.02)	(0.03)
Ext. rating	-0.0114	-0.0127	0.0247	-0.0374	-0.0127	0.0204
	(0.04)	(0.02)	(0.03)	(0.03)	(0.01)	(0.03)
Group	-0.0149	0.0013	0.0473	0.0286	0.0203	-0.0159
	(0.04)	(0.03)	(0.03)	(0.04)	(0.02)	(0.04)
Family	0.0196	0.0490	0.0491	-0.0083	0.0272	0.0638
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
Export	0.0575	-0.0176	0.0267	0.0568^{*}	-0.0015	0.0121
	(0.04)	(0.03)	(0.03)	(0.03)	(0.02)	(0.04)
Concentration (10% to $<30\%$)	0.0242	-0.0101	0.0282	0.0459	0.0277^{*}	0.0387
х <i>У</i>	(0.04)	(0.02)	(0.03)	(0.03)	(0.02)	(0.03)
Concentration $(30\% \text{ to } < 50\%)$	0.0033	-0.0246	0.0307	-0.0118	0.0304	0.0534
()	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
Concentration $(50\% +)$	0.0286	-0.0130	-0.0095	0.0110	0.0357	0.0427
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
Savings bank	0.0016	0.0191	0.0382	0.0184	0.0131	-0.0021
Savings bank	(0.03)	(0.0131)	(0.0302)	(0.02)	(0.01)	(0.03)
Comparativa	0.0147	0.0022	0.02)	0.020	0.0140	0.0187
Cooperative	-0.0147	-0.0023	(0.0029)	(0.0208	-0.0149	(0.02)
T 1 1 1	(0.03)	(0.02)	(0.03)	(0.03)	(0.01)	(0.03)
Landesbank	-0.0072	-0.0346	-0.0371	0.0414	0.0073	-0.0634*
	(0.05)	(0.03)	(0.04)	(0.05)	(0.03)	(0.03)
Others	0.0097	-0.0194	-0.0216	0.0284	0.0017	0.0226
	(0.04)	(0.03)	(0.03)	(0.04)	(0.02)	(0.04)
Industry	Yes	Yes	Yes	Yes	Yes	Yes
Adj. R^2	0.0522	0.0164	0.0441	0.1059	0.0183	0.0275
Ν	652	652	652	652	652	652

Table A.3: Baseline estimations

Note: The table shows OLS estimation results for regressions of dummy variables for different kinds of impairments of credit finance due to the financial crisis on the number of main bank relationships, a set of control variables for firm characteristics, bank classification dummy variables, and industry dummy variables; the baseline category for the main bank relationship dummy variables is *Main bank* (0/3+); robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)
	(1) Information	(2) Reduction	Availabilitu
	111/01/11/01/1	neauction	1100000000g
Main bank (1)	-0.1176***	0.0089	-0.0191
	(0.03)	(0.02)	(0.02)
	(0.00)	(0002)	(0.02)
Main bank (2)	-0.0654**	0.0302	-0.0090
	(0.03)	(0.03)	(0.02)
	× /		
Firm char.	Yes	Yes	Yes
Industry	Yes	Yes	Yes
Bank class.	Yes	Yes	Yes
Pseudo \mathbb{R}^2	0.1638	0.1901	0.2087
Ν	643	497	576
	(4)	(5)	(6)
	Interest	Maturities	Collateral
			0.0000**
Main bank (1)	-0.0538**	-0.0295**	-0.0626**
	(0.02)	(0.01)	(0.03)
Main hank (0)	0.0505***	0.0120**	0.0214
<i>Main ounk (2)</i>	-0.0090	-0.0139	(0.02)
	(0.02)	(0.01)	(0.02)
Firm char	Ves	Ves	Ves
Industry	Ves	Ves	Ves
Bank class	Ves	Ves	Ves
Dalla Class.	100	100	100
Pseudo R^2	0.2589	0.3100	0.1658
Ν	617	347	586

Table A.4: Estimations of binary probit models

Note: The table shows the estimated binary probit marginal effects of the number of main bank relationships on the probability of facing different kinds of impairment of credit finance due to the financial crisis; estimations include a set of control variables for firm characteristics, bank classification dummy variables, and industry dummy variables; the baseline category for the main bank relationship dummy variables is *Main bank* (0/3+); robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	
	Information	Reduction	Availability	
	v		· · · · · · · · · · · · · · · · · · ·	
Main bank (1)	-0.1541***	0.0046	-0.0059	
	(0.05)	(0.03)	(0.04)	
Main bank (2)	-0.1001**	0.0097	-0.0028	
	(0.05)	(0.03)	(0.04)	
	()		()	
Firm char.	Yes	Yes	Yes	
Industry	Yes	Yes	Yes	
Bank class.	Yes	Yes	Yes	
Adj. R^2	0.0480	0.0066	0.0450	
Ň	564	564	564	
	(1)	(5)	(6)	
	(4) Interest	(J) Matumitica	(0)	
	Interest	Muturnites	Contacertai	
Main hank (1)	-0 1143**	-0.0599**	-0.0755*	
1110000 000000 (1)	(0.05)	(0.0333)	(0.04)	
	(0.00)	(0.00)	(0.04)	
Main bank (2)	-0.1270***	-0.0555**	-0.0392	
	(0.04)	(0.03)	(0.04)	
	(0.01)	(0.00)	(0101)	
Firm char.	Yes	Yes	Yes	
Industry	Yes	Yes	Yes	
Bank class.	Yes	Yes	Yes	
Adj. R^2	0.0907	0.0232	0.0116	
N	564	564	564	

Table A.5: Robustness check for reverse causality

Note: Firms for which at least one of the two most important bank relationships has been established in 2007 or later are dropped from the sample to rule out reverse causality concerns; the table shows OLS estimations results for regressions of dummy variables for different kinds of impairments of credit finance due to the financial crisis on the number of main bank relationships, a set of control variables for firm characteristics, bank classification dummy variables, and industry dummy variables; the baseline category for the main bank relationship dummy variables is *Main bank* (0/3+); robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)	(5)	(6)
	(1) Information	(2) Reduction	Availabilitu	Interest	Maturities	(0) Collateral
	111/01/11/01/01/	Reademon	110000000000000	10001000	11/1/0/ 00000	Contaccian
Main bank (1)	-0.1438***	0.0205	-0.0432	-0.1135***	-0.0495**	-0.0516
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
Main bank (2)	-0.0918**	0.0279	-0.0249	-0.1138***	-0.0420*	-0.0324
	(0.05)	(0.03)	(0.04)	(0.04)	(0.02)	(0.04)
$log(Bank assets)_{max}$	-0.0034	-0.0094	0.0074	-0.0132	-0.0078	-0.0047
5 (max	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
$log(Bank \ assets)_{min}$	-0.0034	0.0108	-0.0111	0.0017	0.0039	-0.0018
5(),,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	(0.01)	(0.01)	(0.01)	(0.01)	(0.00)	(0.01)
Bank equity _{max}	-0.0016	0.0013	0.0025	-0.0060**	-0.0029*	-0.0030
1 Junit	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Bank equity _{min}	-0.0105	0.0019	-0.0025	-0.0017	-0.0002	0.0001
1 0//000	(0.01)	(0.01)	(0.01)	(0.01)	(0.00)	(0.01)
Bank liquidity _{max}	0.0088	0.3078**	-0.1915	0.0745	0.0408	-0.1186
Jimaz	(0.16)	(0.13)	(0.13)	(0.14)	(0.06)	(0.13)
$Bank \ liquidity_{min}$	0.0226	-0.2984*	-0.0763	0.1077	0.0119	-0.0246
1 0.000	(0.22)	(0.18)	(0.17)	(0.21)	(0.08)	(0.19)
Bank depositsmar	-0.1229	0.0913	-0.1840	0.0361	0.0625	0.0647
I mar	(0.15)	(0.10)	(0.12)	(0.14)	(0.08)	(0.13)
Bank depositsmin	0.1042	-0.0855	0.1323	-0.0711	-0.0479	-0.1046
	(0.17)	(0.10)	(0.14)	(0.16)	(0.09)	(0.16)
	()	()	(-)	()	()	()
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
Adj. R^2	0.0606	0.0450	0.0505	0.0959	0.0186	0.0319
Ν	649	649	649	649	649	649
		1				

Table A.6: Robustness check using bank balance sheet variables

Note: The table shows OLS estimation results for regressions of dummy variables for different kinds of impairments of credit finance due to the financial crisis on the number of main bank relationships, a set of control variables for firm characteristics, bank balance sheet variables, and industry dummy variables; the baseline category for the main bank relationship dummy variables is *Main bank* (0/3+); robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix to Chapter 2

Purpose (line)		Purpose (loan	Purpose (loan)			
(N=531)		(N=535)				
Working capital	72.32%	Substitution	47.57%			
Investments	57.06%	Enhancement	64.74%			
Others 12.24%		Rationalisation	28.17%			
		R&D	10.09%			
		Acquisitions	8.79%			
		Foreign investment	6.92%			
		Others	15.14%			

 Table B.1: Purpose of negotiated credit

Notes: The table provides the purposes of negotiated lines of credit and loans as reported by firms in the Ifo "Financing of the German Economy" survey. For every negotiated credit contract, firms could report more than one purpose.

	F&F Bi	isiness	F&F Private			
	Yes	No	p > t	Yes	No	p > t
			1			1
Ν	128	956		62	1,009	
Perc.	11.81%	88.19%		5.79%	94.21%	
			a a ciluli			
log(Empl)	4.57	5.02	0.01**	4.62	4.97	0.14
log(Assets)	8.61	9.25	0.009^{***}	9.12	9.16	0.90
log(Age)	3.96	4.05	0.32	3.82	4.06	0.06^{*}
Incorporated	61.72%	62.41%	0.88	68.85%	61.74%	0.27
Ext. rating	25.00%	21.40%	0.36	25.00%	21.46%	0.52
Customer (< 10%)	20.00%	19.91%	0.98	11.48%	20.24%	0.10
Customer $(10\% \text{ to } < 30\%)$	41.60%	45.05%	0.47	44.26%	45.02%	0.91
Customer $(30\% \text{ to } < 50\%)$	26.40%	18.96%	0.05^{*}	24.59%	19.44%	0.33
Customer $(50\% +)$	12.00%	16.08%	0.24	19.67%	15.31%	0.36
Export	83.33%	88.22%	0.12	79.03%	88.04%	0.04**
Group	21.88%	39.79%	0.000***	32.26%	38.10%	0.36
Family	91.41%	74.18%	0.000***	62.90%	76.05%	0.02**
Control	70.22	73.69	0.21	70.26	73.85	0.34
Operating owner	72.66%	59.49%	0.004***	59.68%	61.14%	0.82
Rating	216.21	198.46	0.04**	224.79	197.87	0.02**
Eauity	21.89%	36.84%	0.000***	16.47%	36.45%	0.000***
Debt	33.64%	23.14%	0.003***	41.43%	23.54%	0.000***
Cash	9.61%	11.19%	0.30	8.16%	11.39%	0.12
Return $(<3\%)$	54.31%	44.10%	0.04**	66.10%	44.05%	0.001***
Return (3 to $<7\%$)	31.90%	32.65%	0.87	20.34%	32.79%	0.05^{*}
Return (7 to $<10\%$)	7.76%	14.32%	0.05^{*}	8.47%	14.29%	0.21
Return $(10\% +)$	6.03%	8.93%	0.30	5.08%	8.87%	0.32
Earlypay (0%)	4.10%	3.02%	0.52	6.56%	3.17%	0.16
Earlypay (< 25%)	16.39%	14.55%	0.59	29.51%	13.60%	0.001***
Earlypay (25 to $<50\%$)	7.38%	9.27%	0.49	6.56%	9.00%	0.52
Earlypay (50 to $<75\%$)	14.75%	9.70%	0.08^{*}	14.75%	10.02%	0.24
Earlypay $(75\% +)$	57.38%	63.47%	0.19	42.62%	64.21%	0.001***

Table B.2: F&F finance and firm characteristics

Notes: The table shows descriptive statistics for firm characteristics separately for F&F firms and non-F&F firms; p-values are reported for t-tests of the significance of the difference between respective groups of firms with respect to the firm characteristics; * p < 0.1, ** p < 0.05, *** p < 0.01.

Appendix to Chapter 2

	(1) Rejected (line)	(2) Rejected (loan)	
Discretion	-1.183***	-0.916***	
log(Empl)	(0.34) -0.305	(0.30) -0.148	
	(0.22)	(0.14)	
log(Assels)	(0.17)	(0.141)	
log(Age)	-0.183 (0.14)	-0.133 (0.13)	
Incorporated	(0.11) 1.048^{***} (0.25)	(0.13) 0.409 (0.27)	
Ext. rating	(0.33) -0.521 (0.24)	(0.27) -0.292 (0.21)	
Customer (10% to $<30\%$)	(0.54) 0.514	(0.31) -0.644*	
Customer (30% to $<50\%$)	(0.54) -0.288	(0.35) 0.236	
Customer $(50\% +)$	(0.62) 0.373	(0.40) -0.478	
Group	$(0.66) \\ 0.478$	(0.45) 0.125	
Family	$(0.36) \\ 0.037$	(0.34) -0.317	
Control	$(0.35) \\ 0.007$	(0.40) 0.002	
Export	(0.00) -0.067	(0.00) 0.347	
Operating	(0.44) 1.086^{***}	(0.36) 0.973^{**}	
Rating	(0.37) 0.001	(0.40) -0.000	
Equity	(0.00) -1.157*	(0.00) -1.493**	
Long-term debt	(0.61) -1.355**	(0.64) 0.056	
Cash	(0.61) 5.849***	(0.56) 2.829**	
Return (3% to <7%)	(1.82) -1.357***	(1.22) -0.690**	
Return (7% to <10%)	(0.37) 0.279	(0.28) 0.187	
Return (10% +)	(0.42) -1.090	(0.39)	
Earlypay (0% to $<25\%$)	(1.12) -0.014	-0.277	
Earlypay (25% to $<50\%$)	(0.51) 0.343	(0.51) -0.316	
Earlypay (50% to $<75\%$)	(0.58) -0.827	(0.58) -0.856	
Earlypay (75% $+$)	(0.63) -1.232** (0.56)	(0.57) -1.109** (0.49)	
Industry	Yes	Yes	
N	235	231	

Table B.3: Probit estimations for IV approach

Notes: The table shows results for two separate probit estimations; firm characteristics comprise all variables listed in Table 2.3; industry dummy variables are included based on two-digit WZ 2008 industry classifications; robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)	(5)	(6)
	$F \mathscr{C} F$	$F \mathscr{C} F$	F & F	$F \mathscr{C} F$	$F \mathscr{C} F$	$F \mathscr{C} F$
			Business	Business	Private	Private
	0.059		0.007		0.079	
Rejected (line)	0.053		0.007		0.073	
	(0.09)		(0.07)		(0.08)	
Rejected (loan)		0.176^{*}		0.093		0.161^{*}
		(0.09)		(0.06)		(0.08)
λ_1	-0.059		0.219	. ,	-0.095	. ,
	(0.19)		(0.14)		(0.17)	
λ_2	· · · ·	0.124	. ,	0.137	· · · ·	-0.044
		(0.34)		(0.22)		(0.28)
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.1472	0.1114	0.1439	0.1318	0.1511	0.1350
Ν	286	307	276	302	281	303

Table B.4: Sample selection in OLS estimations

Notes: The table shows results for six separate OLS estimations; firm characteristics comprise all control variables listed in Table 2.3; industry dummy variables are included based on the two-digit WZ 2008 industry classification; robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	(3)	(4)	(5)	(6)
	$F \mathscr{C} F$					
			Business	Business	Private	Private
Rejected (line)	0.236		0.097		0.077	
	(0.20)		(0.16)		(0.17)	
Rejected (loan)	× ,	0.682**	× ,	0.495^{**}	× /	0.472^{*}
0		(0.31)		(0.22)		(0.25)
λ_1	-0.149		0.187	()	-0.145	()
	(0.22)		(0.15)		(0.18)	
λ_{2}	(0.22)	-0.292	(0120)	-0.308	(0120)	-0.282
71 <u>2</u>		(0.39)		(0.27)		(0.33)
		(0.05)		(0.21)		(0.00)
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
F	35.16***	15.99***	34.85***	13.07***	32.47***	15.82***
MES	36.08***	19.43***	35.13***	17.03***	35.54^{***}	18.43***
Ν	234	225	225	220	230	222

Table B.5: Sample selection in IV estimations

Note: The table shows results for separate IV estimations; λ is the inverse Mills ratio from probit estimations for the firms' decision to enter bank credit negotiations; firm characteristics comprise all control variables listed in Table 2.3; industry dummy variables are included based on the two-digit WZ 2008 industry classification; robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01; F is the F-statistic for the first-stage estimation testing the significance of the instrumental variable; MES provides the minimum eigenvalue statistic for the Stock and Yogo (2005) test with *** indicating the 10 percent confidence level and ** indicating the 15 percent confidence level.

	(1)	(2)	(3)	(4)	(5)	(6)
	$F \mathscr{C} F$	$F \mathscr{C} F$	$F \mathscr{E} F$	$F \mathscr{C} F$	$F \mathscr{C} F$	$F \mathscr{C} F$
			Business	Business	Private	Private
Discouraged (line)	0.285		0.267		0.061	
	(0.22)		(0.19)		(0.12)	
Discouraged (loan)		0.126		0.104		0.049
<u> </u>		(0.12)		(0.11)		(0.10)
λ_1	-0.162	× /	-0.032	· · · ·	-0.147	· · ·
	(0.10)		(0.04)		(0.10)	
λ_2	. ,	0.151		0.119	. ,	-0.049
		(0.22)		(0.15)		(0.18)
Firm char.	Yes	Yes	Yes	Yes	Yes	Yes
Industry	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.1707	0.1893	0.2441	0.3084	0.1391	0.1347
Ν	314	306	314	302	310	301

Table B.6: OLS estimations for discouraged borrowers

Notes: The table shows OLS estimation results for separate regressions; λ is the inverse Mills ratio from probit estimations for the firms' decision **not** to enter bank credit negotiations; firm characteristics comprise all control variables listed in Table 2.3; industry dummy variables are included based on the two-digit WZ 2008 industry classification; robust standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.
Appendix to Chapter 3

Appendix A: The Timing of the Treatment

As of 2008, the bank lending supply question in the Ifo Business Survey is asked on a monthly basis. This allows the exact specification of the treatment month t as the month in which the firm first reports restrictive bank lending after reporting normal or accommodating bank lending in previous surveys. From 2003 to 2008, however, the question was asked only twice a year, in March and August. If, for example, a firm reports normal or accommodating bank lending in March and restrictive bank lending in August, it is unclear whether the shift has occurred in August or in a month between March and August.

For our analysis, we assume the treatment month t to be the month right after the firm reports "normal" or "accommodating" bank lending the last time (which would be April in this example). This ensures that our control variables, which are drawn from t-1, are measured in a month in which the firm is definitely untreated and not already affected by constrained credit supply.

Alternatively, we could assume that the treatment occurs in the month in which the firm reports "restrictive" bank lending the first time. We have re-run our whole analysis based on this alternative specification and found our results to be insensitive to a variation of the timing assumption.

Appendix B: Discussion of Annual Frequency in Employment Data

In the Ifo Business Survey firms report their employment figures in November of each year. Two extreme alternatives are available to link annual employment growth rates to monthly data, each leading to severe problems in the estimation of credit supply-side effects on firm-level employment growth.

Alternative 1: Put the number of employees in November into every **subsequent** month until a new figure becomes available in November of the following year and calculate year-on-year growth rates in every month.

Problem: If a firm receives the treatment of constrained credit supply (as defined in Section 3.3.2) in October 2008, for example, the employment growth rate in t+12 (i.e. October 2009) contains the growth between November 2007 and November 2008, which obviously captures a large extent of what is in fact pre-treatment growth, but only one month of post-treatment growth. Estimated credit-supply side effects on firm-level employment growth would therefore be potentially biased.

Alternative 2: Put the number of employees in November into every **previous** month until an old figure in November of the previous year appears and calculate year-on-year growth rates in every month.

Problem: If a firm receives the treatment of constrained credit supply (as defined in Section 3.3.2) in December 2008, for example, the employment growth rate in t+12 (i.e. December 2009) contains the growth between November 2009 and November 2010, which captures only a very small part of the post-treatment year. Estimated credit-supply side effects on firm-level employment growth would again be potentially biased.

Our solution: We put the number of employees in November into the five previous and six subsequent months and calculate year-on-year growth rates in every month. For some observations, this may still not capture post-treatment employment growth precisely, but this intermediate solution is the most conservative way to deal with the problem.





Figure 3.1: Number of treated firms over time

Notes: The graph shows the number of firms that are treated in every quarter; a firm is treated if it reports "restrictive" bank lending after having reported "normal" or "accommodating" bank lending in the previous 12 months; from 2003 to 2008, treatments can occur only in the second and third quarter because firms are surveyed on bank lending only in March and August and the treatment is assumed to occur in the month right after "normal" or "accommodating" bank lending was last reported.

		(1)			(2)	
	$(N_P =$	223: $N_{II} =$	1,351)	$(N_{P} =$	$= 141: N_{II} =$	= 806)
	(1.1		1,001)	(- ' n		
t-1	$\bar{X}_R - \bar{X}_U$	p > t	Bias	$\bar{X}_R - \bar{X}_U$	p > t	Bias
	11 0	1		11 0	1	
Firm size						
log(Empl)	-9.5	0.40	-7.82	-4.7	0.74	-3.94
log(Assets)	-9.4	0.54	-5.70	-5.6	0.77	-3.48
Balance sheet dat	a	0.47	0.04	0 =04	0.00	2 00
Equity	-1.5%	0.47	-6.24	-0.7%	0.80	-2.88
Cash	-0.7%	0.54	-5.24	-1.0%	0.54	-7.74
Long-term debt	1.7%	0.36	9.07	1.4%	0.54	7.73
Short-term debt	1.1%	0.59	3.31	0.5%	0.85	1.48
Cash flow	-0.5%	0.71	-3.72	0.0%	1.00	0.07
ROA	0.4%	0.91	0.99	0.2%	0.97	0.42
Interest Coverage	-0.49	0.89	-1.11	-4.07	0.36	-9.53
Current business	situation					
State (+)	-1.5%	0.71	-3.49	-0.2%	0.96	-0.58
State (-)	6.8%	0.10	15.87	2.3%	0.66	5.42
Orders(+)	-3.2%	0.29	-9.81	-1.9%	0.58	-5.96
Orders (-)	3.2%	0.48	6.77	2.8%	0.63	5.78
Production $(+)$	2.2%	0.45	6.53	-1.6%	0.60	-4.69
Production (-)	0.6%	0.88	1.48	0.2%	0.98	0.37
Short-time	3.1%	0.30	8.86	-1.8%	0.60	-5.13
Export	-1.8%	0.55	-5.52	0.2%	0.95	0.68
Future expectatio	ns					
Business expect $(+)$	-0.3%	0.94	-0.67	0.2%	0.96	0.56
Business expect (-)	2.8%	0.51	6.66	2.4%	0.66	5.60
$Empl \ expect \ (+)$	-1.3%	0.54	-5.53	-0.6%	0.76	-2.71
Empl expect (-)	3.4%	0.39	8.56	0.8%	0.86	2.03
Head count (+)	-0.9%	0.70	-3.68	0.2%	0.95	0.72
Headcount (-)	6.4%	0.13	15.04	4.2%	0.42	9.87
Short-time expect	2.3%	0.52	5.91	-2.5%	0.59	-6.17

Table C.1: Balancing properties for the two matching procedures

Notes: Sample (1) is derived from ten nearest neighbours matching on firm size and balance sheet variables only; Sample (2) is derived from ten nearest neighbours matching on firm size, balance sheet variables, current business situation, and future expectations; differences in means between restricted and unrestricted firms are reported for pre-treatment control variables $B_{i,t-1}$, $C_{i,t-1}$, and $F_{i,t-1}$; p-values are reported for t-tests with H_0 : $\bar{X}_R = \bar{X}_U$; bias statistics are calculated as $(\bar{X}_R - \bar{X}_U)/\sqrt{\frac{\sigma_R^2 + \sigma_U^2}{2}}$ following Rosenbaum and Rubin (1985).

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	0.0887^{***} (0.026)	0.0737^{***} (0.026)	0.015***
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Partly	
Future expectations	No	Partly	
Month	Yes	Yes	
Industry	Yes	Yes	
Adj. R^2	0.1683	0.1895	
Ν	5,503	5,503	

Table C.2: Alternative OLS estimations using $Slowdown_{i,t+12}$

Notes: The table provides results for OLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are reduced sets of control variables (excluding general appraisals) as explained in Section 3.5.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated effects; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$5.74\%^{**}$ (0.0225)	3.16% (0.0241)	2.58%
Industry	Yes	Yes	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Partly	
Future expectations	No	Partly	
p>t	0.01	0.19	
Upper bound	9.44%	7.12%	
Lower bound	2.03%	-0.80%	
Treated obs.	245	214	
Matching obs.	1,516	1,298	

Table C.3: Alternative PSM estimations using $Slowdown_{i,t+12}$

Notes: The table provides results for WLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$; in Estimation (1), weights are derived from PSM based on firm size and balance sheet data in t-1; in Estimation (2), weights are derived from PSM based on firm size, balance sheet data, and reduced sets of control variables (excluding general appraisals) for current business situation and future expectations as explained in Section 3.5.1; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of significance the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval; standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	0.0545^{***}	0.0477^{***}	0.068^{**}
	(0.010)	(0.010)	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
Month	Yes	Yes	
Industry	Yes	Yes	
Adj. R^2	0.2258	0.3041	
Ν	$5,\!193$	$5,\!193$	

Table C.4: OLS estimations using $Slowdown_avg_{i,t+12}$

Notes: The table provides results for OLS estimations of $Slowdown_avg_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are sets of control variables as listed in Table 3.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated effects; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$3.78\%^{***}$ (0.0116)	$2.78\%^{*}$ (0.0149)	1.01%
Industry Firm size Balance sheet data Current situation Future expectations	Yes Yes No No	Yes Yes Yes Yes Yes	
p>t Upper bound Lower bound	$\begin{array}{c} 0.001 \\ 5.69\% \\ 1.88\% \end{array}$	$\begin{array}{c} 0.06 \\ 5.23\% \\ 0.33\% \end{array}$	
Treated obs. Matching obs.	$223 \\ 1,351$	141 806	

Table C.5: PSM estimations using $Slowdown_avg_{i,t+12}$

Notes: The table provides results for WLS estimations of $Slowdown_avg_{i,t+12}$ on the treatment status $Restricted_{i,t}$; in Estimation (1), weights are derived from PSM based on firm size and balance sheet data in t-1; in Estimation (2), weights are derived from PSM based on firm size, balance sheet data, current business situation, and future expectations in t-1; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of significance of the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval; standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	-0.0180***	-0.0152**	0.0025^{**}
	(0.006)	(0.006)	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
Month	Yes	Yes	
Industry	Yes	Yes	
Adj. R^2	0.0598	0.0849	
Ν	5,041	5,041	

Table C.6: OLS estimations using $\Delta Empl_{i,t+12}$

Notes: The table provides results for OLS estimations of $\Delta Empl_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are sets of control variables as listed in Table 3.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated effects; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$-1.08\%^{*}$ (0.0059)	0.35% (0.0078)	-1.43%
Industry	Yes	Yes	
Balance sheet data	Yes Yes	Yes	
Current situation Future expectations	No No	Yes Yes	
p>t	0.07	0.65	
Upper bound Lower bound	-0.11% -2.04%	1.64% -0.94\%	
Treated obs. Matching obs	208 1-295	130 773	

Table C.7: PSM estimations using $\Delta Empl_{i,t+12}$

Notes: The table provides results for WLS estimations of $\Delta Empl_{i,t+12}$ on the treatment status $Restricted_{i,t}$; in Estimation (1), weights are derived from PSM based on firm size and balance sheet data in t-1; in Estimation (2), weights are derived from PSM based on firm size, balance sheet data, current business situation, and future expectations in t-1; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of significance of the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval; standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	0.0937^{***}	0.0749^{***}	0.0188^{***}
	(0.027)	(0.026)	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
Month	Yes	Yes	
Industry	Yes	Yes	
Adj. R^2	0.0881	0.1263	
Ν	$5,\!141$	5,141	

Table C.8: OLS estimations using $Empl_Decrease_{i,t+12}$

Notes: The table provides results for OLS estimations of $Empl_Decrease_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are sets of control variables as listed in Table 3.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated effects; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$6.23\%^{**}$ (0.0250)	1.24% (0.0324)	4.99%
Industry	Yes	Yes	
Firm size	Yes	Yes	
Balance sheet data	Yes	Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
	0.01	0.70	
p>t	0.01	0.70	
Upper bound	10.34%	6.56%	
Lower bound	2.13%	-4.08%	
Treated obs.	221	136	
Matching obs.	1,340	792	

Table C.9:	PSM	estimation	using	$Empl_{-}$	$_Decrease_{i,t+12}$
------------	-----	------------	-------	------------	----------------------

Notes: The table provides results for WLS estimations of $Empl_Decrease_{i,t+12}$ on the treatment status $Restricted_{i,t}$; in Estimation (1), weights are derived from PSM based on firm size and balance sheet data in t-1; in Estimation (2), weights are derived from PSM based on firm size, balance sheet data, current business situation, and future expectations in t-1; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of the significance of the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval; standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)		
	$\overline{\mathbf{v}}B$	$\bar{\mathbf{v}}^{BCF}$		
	XB	XBOI	p > t	
Production (t+12)				
Slowdown	70.0%	71.6%	0.73	
$Slowdown_avg$	23.4%	24.1%	0.80	
Employment (t+12)				
$\Delta Empl$	-1.0%	-0.3%	0.56	
Empl Decrease	45.2%	41.9%	0.54	

Table C.10: Treated firms in different matched samples

Notes: The table provides the means of all outcome variables for treated firms in the matched sample when matching on firm size and balance sheet variables in t-1 and the matched sample when matching on all variables in t-1; p-values are provided for a two-group mean comparison test with H_0 : $\bar{X}_B = \bar{X}_{BCF}$.

Restricted	0.0731^{**} (0.032)	0.0587^{*} (0.031)	0.0144*
Firm size Balance sheet data	Yes Yes	Yes Yes	
Current situation	No	Yes	
Future expectations	No	Yes	
Month Industry	Yes Yes	Yes Yes	
, Adj. R^2 N	$0.2086 \\ 3,315$	$0.2552 \\ 3,315$	

Table C.11: Financial crisis OLS estimations using $Slowdown_{i,t+12}$

Notes: The table provides results for OLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$ and different sets of pre-treatment control variables in the subsample as of July 2007; standard errors (reported in parentheses) are clustered at the firm-level; "Firm size", "Balance sheet data", "Current situation", and "Future expectations" are sets of control variables as listed in Table 3.1; industry dummy variables are included based on the two-digit WZ 2008 industry classification; the two samples contain only those observations for which all control variables of Estimation (2) are available; the third column provides the difference between the two estimated effects; its significance is tested using a t-test with H_0 : $\beta_1^B = \beta_1^{BCF}$ based on Clogg, Petkova, and Haritou (1995); * p < 0.1, ** p < 0.05, *** p < 0.01.

	(1)	(2)	$\beta_1^B - \beta_1^{BCF}$
Restricted	$\begin{array}{c} 4.97\%^{*} \\ (0.0260) \end{array}$	3.84% (0.0306)	1.14%
Industry	Yes	Yes	
Firm size Balance sheet data	Yes Yes	Yes Yes	
Current situation Future expectations	No No	Yes Yes	
	0.06	0.91	
p>t Upper bound	9.25%	8.87%	
Lower bound	0.70%	-1.20%	
Treated obs.	160	116	
Matching obs.	928	690	

Table C.12: Financial crisis PSM estimation using $Slowdown_{i,t+12}$

Notes: The table provides results for WLS estimations of $Slowdown_{i,t+12}$ on the treatment status $Restricted_{i,t}$ in the subsample as of July 2007; in Estimation (1), weights are derived from PSM based on firm size and balance sheet data in t-1; in Estimation (2), weights are derived from PSM based on firm size, balance sheet data, current business situation, and future expectations in t-1; industry dummy variables based on the two-digit WZ 2008 industry classification are also included in all PSM estimations; p-values are reported for a t-test of significance of the estimated treatment effect; the significance of the difference between the two estimated effects in the third column cannot be tested; upper and lower bounds are reported for the 95 percent confidence interval; standard errors are reported in parentheses; * p < 0.1, ** p < 0.05, *** p < 0.01.

List of Figures

3.1 Number of tre	reated firms over time	. 105
-------------------	------------------------	-------

List of Tables

1.1	Variable descriptions	17
1.2	Impairment of credit finance due to the financial crisis	18
1.3	Relationship banking and impaired credit finance	21
1.4	Firm characteristics and relationship banking	22
1.5	Bank classifications, impaired credit finance, and relationship banking	24
1.6	Baseline OLS estimations	28
1.7	Bank balance sheets and relationship banking	32
2.1	Variable descriptions	41
2.2	F&F finance and credit negotiations	42
2.3	F&F finance and firm characteristics	44
2.4	OLS estimations	50
2.5	IV estimations	51
2.6	Determinants of Discretionary Lending	53
2.7	F&F firms, credit negotiations, and bank classifications $\hdots \hdots \hdots$	55
2.8	F&F firms, credit negotiations, and bank-firm relationships	55
3.1	Variable descriptions	68
3.2	Firms' post-treatment business activity	71
3.3	Firms' balance sheets in $t-1$	72
3.4	Firms' current business situation and future expectations in $t-1$	74
3.5	OLS estimations using $Slowdown_{i,t+12}$	77
3.6	PSM estimation using $Slowdown_{i,t+12}$	79
A.1	Analysis of response behaviour based on data from the Ifo Investment Survey	91

LIST OF TABLES

A.2	Features of main bank relationships	92
A.3	Baseline estimations	93
A.4	Estimations of binary probit models	94
A.5	Robustness check for reverse causality	95
A.6	Robustness check using bank balance sheet variables	96
B.1	Purpose of negotiated credit	97
B.2	F&F finance and firm characteristics	98
B.3	Probit estimations for IV approach	99
B.4	Sample selection in OLS estimations	100
B.5	Sample selection in IV estimations	101
B.6	OLS estimations for discouraged borrowers	102
C.1	Balancing properties for the two matching procedures	106
C.2	Alternative OLS estimations using $Slowdown_{i,t+12}$	107
C.3	Alternative PSM estimations using $Slowdown_{i,t+12}$	108
C.4	OLS estimations using $Slowdown_avg_{i,t+12}$	109
C.5	PSM estimations using $Slowdown_avg_{i,t+12}$	110
C.6	OLS estimations using $\Delta Empl_{i,t+12}$	111
C.7	PSM estimations using $\Delta Empl_{i,t+12}$	112
C.8	OLS estimations using $Empl_Decrease_{i,t+12}$	113
C.9	PSM estimation using $Empl_Decrease_{i,t+12}$	114
C.10	Treated firms in different matched samples	114
C.11	Financial crisis OLS estimations using $Slowdown_{i,t+12}$	115
C.12	Financial crisis PSM estimation using $Slowdown_{i,t+12}$	116

Bibliography

- ABBERGER, K., M. BIRNBRICH, AND C. SEILER (2009): "Der "Test des Tests" im Handel - eine Metaumfrage zum ifo Konjunkturtest," *Ifo Schnelldienst*, 62(21), 34–41.
- ADRIAN, T., P. COLLA, AND H. S. SHIN (2012): "Which Financial Frictions? Parsing the Evidence from the Financial Crisis of 2007-9," NBER Macro Annual Conference, April 20-21, 2012.
- AGRESTI, A., AND B. FINLAY (2008): *Statistical Methods for the Social Sciences*. Pearson Prentice Hall.
- ALBERTAZZI, U., AND D. J. MARCHETTI (2010): "Credit Supply, Flight to Quality and Evergreening: An Analysis of Bank-Firm Relationships After Lehman," Banca d'Italia Working Paper No. 756.
- ALLEN, F., E. CARLETTI, J. QIAN, AND P. VALENZUELA (2013): "Financial Intermediation, Markets, and Alternative Financial Sectors," in *Handbook of the Economics of Finance*, ed. by G. M. Constantinides, M. Harris, and R. M. Stulz, vol. 2, Part B, pp. 799–1612. North Holland.
- ALLEN, F., R. CHAKRABARTI, S. DE, J. QIAN, AND M. QIAN (2012): "Financing Firms in India," *Journal of Financial Intermediation*, 21(3), 409–445.
- ALLEN, F., AND J. QIAN (2010): "Comparing Legal and Alternative Institutions in Finance and Commerce," in *Global Perspectives on the Rule of Law*, ed. by J. J. Heckman, R. L. Nelson, and L. Cabatingan, chap. 6, pp. 108–144. Routledge.
- ALLEN, F., J. QIAN, AND M. QIAN (2005): "Law, Finance, and Economic Growth in China," *Journal of Financial Economics*, 77(1), 57–116.

- ALLEN, F., J. QIAN, AND C. ZANG (2011): "An Alternative View on Law, Institutions, Finance and Growth," Unpublished Working Paper.
- ALMEIDA, H., M. CAMPELLO, B. LARANJEIRA, AND S. WEISBENNER (2011): "Corporate Debt Maturity and the Real Effects of the 2007 Credit Crisis," *Critical Finance Review*, 1(1), 3–58.
- AMITI, M., AND D. E. WEINSTEIN (2013): "How Much do Bank Shocks Affect Investment? Evidence from Matched Bank-Firm Loan Data," NBER Working Paper 18890.
- ANG, J. S. (1992): "On the Theory of Finance for Privately Held Firms," Journal of Small Business Finance, 1(3), 185–203.
- ANGRIST, J. D., AND J.-S. PISCHKE (2008): Mostly Harmless Econometrics: An Empiricist's Companion. Princeton University Press.
- ASTEBRO, T., AND I. BERNHARDT (2003): "Start-Up Financing, Owner Characteristics, and Survival," *Journal of Economics and Business*, 55(4), 303–319.
- AYYAGARI, M., A. DEMIRGÜC-KUNT, AND V. MAKSIMOVIC (2010): "Formal Versus Informal Finance: Evidence from China," *The Review of Financial Studies*, 23(8), 3048–3097.
- BAE, K.-H., J.-K. KANG, AND C.-W. LIM (2002): "The Value of Durable Bank Relationships: Evidence from Korean Banking Shocks," *Journal of Financial Economics*, 64(2), 181–214.
- BARRAZA, J. S. E., W. Y. LEE, AND T. J. YEAGER (2014): "Financial Crisis and the Supply of Corporate Credit," Electronic copy available at http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2422678.
- BASU, A., AND S. C. PARKER (2001): "Family Finance and New Business Start-Ups," Oxford Bulletin of Economics and Statistics, 63(3), 333–358.
- BECK, T., A. DEMIRGÜÇ-KUNT, AND V. MAKSIMOVIC (2005): "Financial and Legal Constraints to Growth: Does Firm Size Matter?," *Journal of Finance*, 60(1), 137–177.

- BECK, T., A. DEMIRGÜÇ-KUNT, AND V. MAKSIMOVIC (2008): "Financing Patterns Around the World: Are Small Firms Different?," *Journal of Financial Economics*, 89(3), 467–487.
- BECK, T., R. LEVINE, AND N. LOAYZA (2000): "Finance and the Sources of Growth," Journal of Financial Economics, 58(1-2), 261–300.
- BECKER, B., AND V. IVASHINA (2014): "Cyclicality of Credit Supply: Firm Level Evidence," *Journal of Monetary Economics*, 62, 76–93.
- BECKER, S., AND K. WOHLRABE (2008): "Micro Data at the Ifo Institute for Economic Research - The "Ifo Business Survey", Usage and Access," *Journal of Applied Social Science Studies*, 128(2), 307–319.
- BEHN, M., R. HASELMANN, AND V. VIG (2014): "The Limits of Model-Based Regulation," Unpublished Working Paper.
- BERGER, A. N., AND G. F. UDELL (1995): "Relationship Lending and Lines of Credit in Small Firm Finance," *Journal of Business*, 68(3), 351–381.
- (1998): "The Economics of Small Business Finance: The Roles of Private Equity and Debt Markets in the Financial Growth Cycle," *Journal of Banking and Finance*, 22(6-8), 613–673.
- (2002): "Small Business Credit Availability and Relationship Lending: The Importance of Bank Organisational Structure," *Economic Journal*, 112(477), F32–F53.
- BERNANKE, B. S. (1983): "Non-Monetary Effects of the Financial Crisis in the Propagation of the Great Depression," *American Economic Review*, 73(3), 257–276.
- BERNANKE, B. S., AND A. S. BLINDER (1988): "Credit, Money, and Aggregate Demand," *American Economic Review*, 78(2), 435–439.
- BERNANKE, B. S., AND M. GERTLER (1989): "Agency Costs, Net Worth, and Business Fluctuations," *American Economic Review*, 79(1), 14–31.

(1995): "Inside the Black Box: The Credit Channel of Monetary Policy Transmission," *Journal of Economic Perspectives*, 9(4), 27–48.

- BERNANKE, B. S., M. GERTLER, AND S. GILCHRIST (1996): "The Financial Accelerator and the Flight to Quality," *Review of Economics and Statistics*, 78(1), 1–15.
- BHARATH, S. T., S. DAHIYA, A. SAUNDERS, AND A. SRINIVASAN (2011): "Lending Relationships and Loan Contract Terms," *Review of Financial Studies*, 24(4), 1141– 1203.
- BITLER, M. P., A. M. ROBB, AND J. D. WOLKEN (2001): "Financial Services Used by Small Businesses: Evidence from the 1998 Survey of Small Business Finances," *Federal Reserve Bulletin*, Issue April, 183–205.
- BOLTON, P., X. FREIXAS, L. GAMBACORTA, AND P. E. MISTRULLI (2013): "Relationship and Transaction Lending in a Crisis," NBER Working Paper No. 19467.
- BOOT, A. W. (2000): "Relationship Banking: What Do We Know?," Journal of Financial Intermediation, 9(1), 7–25.
- BOOT, A. W., AND A. V. THAKOR (1994): "Moral Hazard and Secured Lending in an Infinitely Repeated Credit Market Game," *International Economic Review*, 35(4), 899–920.
- BREWER, E., H. GENAY, W. C. HUNTER, AND G. G. KAUFMAN (2003): "The Value of Banking Relationships During a Financial Crisis: Evidence from Failures of Japanese Banks," *Journal of the Japanese and International Economies*, 17(3), 233–262.
- BROWN, M., S. ONGENA, A. A. POPOV, AND P. YEŞIN (2011): "Who Needs Credit and Who Gets Credit in Eastern Europe?," *Economic Policy*, 26(65), 93–130.
- BRUNNERMEIER, M. K. (2009): "Deciphering the Liquidity and Credit Crunch 2007-2008," Journal of Economic Perspectives, 23(1), 77–100.
- CAMPELLO, M., J. R. GRAHAM, AND C. R. HARVEY (2010): "The Real Effects of Financial Constraints: Evidence from a Financial Crisis," *Journal of Financial Economics*, 97(3), 470–487.
- CARBÓ-VALVERDE, S., F. RODRÍGUEZ-FERNÁNDEZ, AND G. F. UDELL (2013): "Trade Credit, the Financial Crisis, and SME Access to Finance," Available at http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2307246.

- CASSAR, G. (2004): "The Financing of Business Start-Ups," Journal of Business Venturing, 19(2), 261–283.
- CHAVA, S., AND A. PURNANANDAM (2011): "The Effect of Banking Crisis on Bank-Dependent Borrowers," *Journal of Financial Economics*, 99(1), 116–135.
- CHAVIS, L. W., L. F. KLAPPER, AND I. LOVE (2011): "The Impact of the Business Environment on Young Firm Financing," *World Bank Economic Review*, 25(3), 486– 507.
- CHODOROW-REICH, G. (2014): "The Employment Effects of Credit Market Disruptions: Firm-level Evidence from the 2008-09 Financial Crisis," *Quarterly Journal of Economics*, 129(1), 1–59.
- CHUA, J. H., J. J. CHRISMAN, F. KELLERMANNS, AND Z. WU (2011): "Family Involvement and New Venture Debt Financing," *Journal of Business Venturing*, 26(4), 472–488.
- CLOGG, C. C., E. PETKOVA, AND A. HARITOU (1995): "Statistical Methods for Comparing Regression Coefficients Between Models.," *American Journal of Sociology*, 100(5), 1261–1293.
- COLE, R. A. (1998): "The Importance of Relationships to the Availability of Credit," Journal of Banking and Finance, 22(6-8), 959–977.
- COLE, R. A., L. G. GOLDBERG, AND L. J. WHITE (2004): "Cookie Cutter vs. Character: The Micro Structure of Small Business Lending by Large and Small Banks," *Journal of Financial and Quantitative Analysis*, 39(2), 227–251.
- COLEMAN, S., AND M. CARSKY (1999): "Sources of Capital for Small Family-Owned Businesses Evidence from the National Survey of Small Business Finances," *Family Business Review*, 12(1), 73–85.
- COULIBALY, B., H. SAPRIZA, AND A. ZLATE (2013): "Financial Frictions, Trade Credit, and the 2008–09 Global Financial Crisis," *International Review of Economics & Finance*, 26(C), 25–38.

- CROCI, E., J. A. DOUKAS, AND H. GONENC (2011): "Family Control and Financing Decisions," *European Financial Management*, 17(5), 860–897.
- DEGRYSE, H., L. LU, AND S. ONGENA (2013): "Informal or Formal Financing? Or Both? First Evidence on the Co-Funding of Chinese Firms," European Banking Center Discussion Paper No. 2013-034.
- DEGRYSE, H., AND P. VAN CAYSEELE (2000): "Relationship Lending within a Bank-Based System: Evidence from European Small Business Data," *Journal of Financial Intermediation*, 9(1), 90–109.
- DEHEJIA, R. H., AND S. WAHBA (1999): "Causal Effects in Nonexperimental Studies: Reevaluating the Evaluation of Training Programs," *Journal of the American Statistical Association*, 94(448), 1053–1062.
- DELL'ARICCIA, G., E. DETRAGIACHE, AND R. G. RAJAN (2008): "The Real Effect of Banking Crises," *Journal of Financial Intermediation*, 17(1), 89–112.
- DELOITTE & TOUCHE GMBH (2012): "Finanzierung im Mittelstand," Electronic copy available at http://www2.deloitte.com/de/de/pages/mittelstand/contents/Finanzierungim-Mittelstand.html.
- DEMIRGÜÇ-KUNT, A., AND V. MAKSIMOVIC (1998): "Law, Finance, and Firm Growth," *Journal of Finance*, 53(6), 2107–2137.
- DENIS, D. J., AND V. T. MIHOV (2003): "The Choice Among Bank Debt, Non-Bank Private Debt, and Public Debt: Evidence From New Corporate Borrowings," *Journal* of Financial Economics, 70(1), 3–28.
- DETRAGIACHE, E., P. GARELLA, AND L. GUISO (2000): "Multiple versus Single Banking Relationships: Theory and Evidence," *Journal of Finance*, 55(3), 1133–1161.
- DEYOUNG, R., A. GRON, G. TORNA, AND A. WINTON (2014): "Risk Overhang and Loan Portfolio Decisions: Small Business Loan Supply Before and During the Financial Crisis," Available at http://papers.ssrn.com/sol3/papers.cfm?abstract_id=2140952.
- DJANKOV, S., I. LIEBERMAN, J. MUKHERJEE, AND T. NENOVA (2003): "Going Informal: Benefits and Costs," in *The Informal Economy in the EU Access Countries: Size*,

Scope, Trends and Challenges to the Process of EU Enlargement, chap. 3, pp. 63–79. Center for Study of Democracy.

- DUCHIN, R., O. OZBAS, AND B. A. SENSOY (2010): "Costly External Finance, Corporate Investment, and the Subprime Mortgage Credit Crisis," *Journal of Financial Economics*, 97(3), 418–435.
- DUYGAN-BUMP, B., A. LEVKOV, AND J. MONTORIOL-GARRIGA (2011): "Financial Constraints and Unemployment: Evidence from the Great Recession," Federal Reserve Bank of Boston Working Paper No. QAU10-6.
- ELLUL, A. (2010): "Control Motivations and Capital Structure Decisions," Available at http://www.eief.it/files/2010/11/andrew-ellul.pdf.
- ELSAS, R. (2005): "Empirical Determinants of Relationships Lending," Journal of Financial Intermediation, 14(1), 32–57.
- ENGEMANN, M., K. ECK, AND M. SCHNITZER (2011): "Trade Credits and Bank Credits in International Trade: Substitutes or Complements?," Bavarian Graduate Program in Economics Working Paper No. 108.
- FAMA, E. F. (1985): "What's Different About Banks?," Journal of Monetary Economics, 15(1), 29–39.
- FARRE-MENSA, J., AND A. LJUNGQVIST (2013): "Do Measures of Financial Constraints Measure Financial Constraints?," NBER Working Paper No. 19551.
- FAULKENDER, M., AND M. A. PETERSEN (2006): "Does the Source of Capital Affect Capital Structure?," *Review of Financial Studies*, 19(1), 45–79.
- FLUCK, Z., D. HOLTZ-EAKIN, AND H. S. ROSEN (1998): "Where Does the Money Come From? The Financing of Small Entrepreneurial Enterprises," Metropolitan Studies Program Series Occasional Paper No. 191.
- GAMBACORTA, L. (2005): "Inside the Bank Lending Channel," *European Economic Review*, 49(7), 1737–1759.

- GAN, J. (2007): "The Real Effects of Asset Market Bubbles: Loan- and Firm-Level Evidence of a Lending Channel," *Review of Financial Studies*, 20(6), 1941–1973.
- GERTLER, M., AND N. KIYOTAKI (2010): "Financial Intermediation and Credit Policy in Business Cycle Analysis," in *Handbook of Monetary Economics*, ed. by B. M. Friedman, and M. Woodford, vol. 3, chap. 11, pp. 547–599. North Holland.
- GIANNETTI, M., M. BURKART, AND T. ELLINGSEN (2011): "What You Sell Is What You Lend? Explaining Trade Credit Contracts," *Review of Financial Studies*, 24(4), 1261–1298.
- GIANNETTI, M., AND X. YU (2014): "Development and Relationship-Based Financing," Available at http://papers.ssrn.com/sol3/papers.cfm?abstract_id=959188.
- GROPP, R., C. GRUENDL, AND A. GUETTLER (2013): "Hidden Gems and Borrowers with Dirty Little Secrets: Investment in Soft Information, Borrower Self-Selection and Competition," European Banking Center Discussion Paper No. 2013-005.
- HACKETHAL, A. (2004): "German Banks and Banking Structure," in *The German Fi*nancial System, ed. by J. P. Krahnen, and R. H. Schmidt, chap. 3, pp. 71–105. Oxford University Press.
- HAINZ, C., AND T. NABOKIN (2013): "Measurement and Determinants of Access to Loans," CESifo Working Paper Series No. 4190.
- HAINZ, C., AND M. WIEGAND (2013): "Financing of the German Economy During the Financial Crisis," *CESifo DICE Report*, 11(1), 48–54.
- HALTIWANGER, J., R. S. JARMIN, AND J. MIRANDA (2013): "Who Creates Jobs? Small versus Large versus Young," *Review of Economics and Statistics*, 95(2), 347–361.
- HARHOFF, D., AND T. KÖRTING (1998): "Lending Relationships in Germany Empirical Evidence from Survey Data," *Journal of Banking and Finance*, 22(10-11), 1317–1353.
- HECKMAN, J. J. (1979): "Sample Selection as a Specification Error," *Econometrica*, 47(1), 153–161.

- HOLMSTROM, B., AND J. TIROLE (1997): "Financial Intermediation, Loanable Funds, and the Real Sector," *The Quarterly Journal of Economics*, 112(3), 663–691.
- INDERST, R., AND H. M. MÜLLER (2007): "A Lender-Based Theory of Collateral," Journal of Financial Economics, 84(3), 826–859.
- INTERNATIONAL MONETARY FUND (2008): "Global Financial Stability Report: Containing Systemic Risks and Restoring Financial Soundness," Electronic copy available at http://www.imf.org/external/Pubs/FT/GFSR/2008/01/pdf/text.pdf.

— (2009): "Germany: 2008 Article IV Consultation," IMF Country Report No. 09/ 15.

- IVASHINA, V., AND D. SCHARFSTEIN (2010): "Bank Lending During the Financial Crisis of 2008," *Journal of Financial Economics*, 97(3), 319–338.
- JAYARATNE, J., AND P. E. STRAHAN (1996): "The Finance-Growth Nexus: Evidence from Bank Branch Deregulation," *Quarterly Journal of Economics*, 111(3), 639–670.
- JIMÈNEZ, G., S. ONGENA, J.-L. PEYDRO, AND J. SAURINA (2012): "Credit Supply Versus Demand: Bank and Firm Balance-Sheet Channels in Good and Crisis Times," European Banking Center Discussion Paper No. 2012-003.
- KAHLE, K. M., AND R. M. STULZ (2013): "Access to Capital, Investment, and the Financial Crisis," *Journal of Financial Economics*, 110(2), 280–299.
- KANG, J.-K., AND R. M. STULZ (2000): "Do Banking Shocks Affect Borrowing Firm Performance? An Analysis of the Japanese Experience," *Journal of Business*, 73(1), 1–23.
- KASHYAP, A. K., AND J. C. STEIN (1994): "Monetary Policy and Bank Lending," in Monetary Policy, ed. by N. G. Mankiw, chap. 7, pp. 221–261. University of Chicago Press.
 - (2000): "What Do a Million Observations on Banks Say about the Transmission of Monetary Policy?," *American Economic Review*, 90(3), 407–428.

- KHWAJA, A. I., AND A. MIAN (2008): "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market," *American Economic Review*, 98(4), 1413–1442.
- KING, R. G., AND R. LEVINE (1993): "Finance and Growth: Schumpeter Might be Right," *Quarterly Journal of Economics*, 108(3), 717–737.
- KIPAR, S. (2011): "The Effect of Restrictive Bank Lending on Innovation: Evidence from a Financial Crisis," Ifo Working Paper No. 109.
- KISHAN, R. P., AND T. P. OPIELA (2000): "Bank Size, Bank Capital, and the Bank Lending Channel," *Journal of Money, Credit and Banking*, 32(1), 121–141.
- (2006): "Bank Capital and Loan Asymmetry in the Transmission of Monetary Policy," *Journal of Banking and Finance*, 30(1), 259–285.
- KIYOTAKI, N., AND J. MOORE (1997): "Credit Cycles," Journal of Political Economy, 105(2), 211–248.
- KROSZNER, R. S., L. LAEVEN, AND D. KLINGEBIEL (2007): "Banking Crises, Financial Dependence, and Growth," *Journal of Financial Economics*, 84(1), 187–228.
- LEHMANN, E. E., AND D. NEUBERGER (2001): "Do Lending Relationships Matter? Evidence from Bank Survey Data in Germany," Journal of Economic Behaviour & Organization, 45(4), 339–359.
- LEMMON, M. L., AND J. F. ZENDER (2010): "Debt Capacity and Tests of Capital Structure Theories," *Journal of Financial and Quantitative Analysis*, 45(5), 1161–1187.
- LIN, H., AND D. PARAVISINI (2012): "The Effect of Financing Constraints on Risk," *Review of Finance*, 17(1), 229–259.
- MASULIS, R. W., P. K. PHAM, AND J. ZEIN (2011): "Family Business Groups Around the World: Financing Advantages, Control Motivations, and Organizational Choices," *Review of Financial Studies*, 24(11), 3556–3600.
- MYERS, S., AND N. S. MAJLUF (1984): "Corporate Financing and Investment Decisions When Firms Have Information That Investors Do Not Have," *Journal of Financial Economics*, 13(2), 187–221.

- ONGENA, S., D. C. SMITH, AND D. MICHALSEN (2003): "Firms and Their Distressed Banks: Lessons from the Norwegian Banking Crisis," *Journal of Financial Economics*, 67(1), 81–112.
- PEEK, J., AND E. S. ROSENGREN (2000): "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States," *American Economic Review*, 90(1), 30–45.
- PETERSEN, M. A., AND R. G. RAJAN (1994): "The Benefits of Lending Relationships: Evidence from Small Business Data," *Journal of Finance*, 49(1), 3–37.
- (1995): "The Effect of Credit Market Competition on Lending Relationships," *Quarterly Journal of Economics*, 110(2), 407–443.
- POPOV, A. A., AND G. F. UDELL (2012): "Cross-Border Banking, Credit Access, and the Financial Crisis," *Journal of International Economics*, 87(1), 147–161.
- PURI, M., J. ROCHOLL, AND S. STEFFEN (2011a): "Global Retail Lending in the Aftermath of the US Financial Crisis: Distinguishing Between Supply and Demand Effects," *Journal of Financial Economics*, 100(3), 556–578.
- PURI. J. S. М., STEFFEN (2011b): "Rules Versus ROCHOLL, AND Decisions," Discretion in Bank Lending Electronic available copy athttp://papers.ssrn.com/sol3/papers.cfm?abstract_id=1786831.
- RAJAN, R. G., AND A. WINTON (1995): "Covenants and Collateral as Incentives to Monitor," *Journal of Finance*, 50(4), 1113–1146.
- RAJAN, R. G., AND L. ZINGALES (1998): "Financial Dependence and Growth," American Economic Review, 88(3), 559–586.
- ROMANO, C. A., G. A. TANEWSKI, AND K. X. SMYRNIOS (2001): "Capital Structure Decision Making: A Model for Family Business," *Journal of Businss Venturing*, 16(3), 285–310.
- ROSENBAUM, P. R., AND D. B. RUBIN (1983): "The Central Role of the Propensity Score in Obervational Studies for Causal Effects," *Biometrika*, 70(1), 41–55.

- ROSENBAUM, P. R., AND D. B. RUBIN (1985): "Constructing a Control Group Using Multivariate Matched Sampling Methods That Incorporate the Propensity Score," *The American Statistician*, 39(1), 33–38.
- ROTTMANN, H., AND T. WOLLMERSHAEUSER (2013): "A Micro Data Approach to the Identification of Credit Crunches," *Applied Economics*, 45(17), 2423–2441.
- SANTIKIAN, L. (2014): "The Ties That Bind: Bank Relationships and Small Business Lending," *Journal of Financial Intermediation*, 23(2), 177–213.
- SANTOS, J. A. (2011): "Bank Loan Pricing Following the Subprime Crisis," Review of Financial Studies, 24(6), 1916–1943.
- SANYAL, P., AND C. L. MANN (2010): "The Financial Structure of Startup Firms: The Role of Assets, Information, and Entrepreneur Characteristics," Federal Reserve Bank of Boston Working Paper No. 10-17.
- SHARPE, S. A. (1990): "Asymmetric Information, Bank Lending, and Implicit Contracts: A Stylized Model of Customer Relationships," *Journal of Finance*, 45(4), 1069– 1087.
- SHLEIFER, A., AND R. W. VISHNY (2010): "Unstable Banking," Journal of Financial Economics, 97(3), 306–318.
- STAIGER, D., AND J. H. STOCK (1997): "Instrumental Variables Regression With Weak Instruments," *Econometrica*, 65(3), 557–586.
- STIGLITZ, J. E., AND A. WEISS (1981): "Credit Rationing in Markets with Imperfect Information," *American Economic Review*, 71(3), 393–410.
- STOCK, J. H., AND M. YOGO (2005): "Testing For Weak Instruments in Linear IV Regression," in *Identification and Inference for Econometric Models: Essays in Honor* of Thomas Rothenberg, ed. by D. W. K. Andrews, and J. H. Stock, chap. 5, pp. 80–108. Cambridge University Press.
- STULZ, R. (1988): "Managerial Control of Voting Rights: Financing Policies and the Market for Corporate Control," *Journal of Financial Economics*, 20(1-2), 25–54.

- THAKOR, A. V. (1996): "Capital Requirements, Monetary Policy, and Aggregate Bank Lending: Theory and Empirical Evidence," *Journal of Finance*, 51(1), 279–324.
- TITMAN, S., AND R. WESSELS (1988): "The Determinants of Capital Structure Choice," Journal of Finance, 43(1), 1–19.
- VON THADDEN, E.-L. (2004): "Asymmetric Information, Bank Lending and Implicit Contracts: The Winner's Curse," *Finance Research Letters*, 1(1), 11–23.
- WOOLDRIDGE, J. M. (2010): Econometric Analysis of Cross Section and Panel Data. MIT Press.
- YAMORI, N., AND A. MURAKAMI (1999): "Does Bank Relationship Have an Economic Value?: The Main Effect of Main Bank Failure on Client Firms," *Economics Letters*, 65(1), 115–120.

Eidesstattliche Versicherung

Ich versichere hiermit eidesstattlich, dass ich die vorliegende Arbeit selbstständig und ohne fremde Hilfe verfasst habe. Die aus fremden Quellen direkt oder indirekt übernommenen Gedanken sowie mir gegebene Anregungen sind als solche kenntlich gemacht. Die Arbeit wurde bisher keiner anderen Prüfungsbehörde vorgelegt und auch noch nicht veröffentlicht. Sofern ein Teil der Arbeit aus bereits veröffentlichten Papers besteht, habe ich dies ausdrücklich angegeben.

24. Juni 2014 Manuel Wiegand